574 LECTURE NOTES IN ECONOMICS AND MATHEMATICAL SYSTEMS

Rainer Kleber

Dynamic Inventory Management in Reverse Logistics



Lecture Notes in Economics and Mathematical Systems

Founding Editors:

M. Beckmann H.P. Künzi

Managing Editors:

Prof. Dr. G. Fandel Fachbereich Wirtschaftswissenschaften Fernuniversität Hagen Feithstr. 140/AVZ II, 58084 Hagen, Germany

Prof. Dr. W. Trockel Institut für Mathematische Wirtschaftsforschung (IMW) Universität Bielefeld Universitätsstr. 25, 33615 Bielefeld, Germany

Editorial Board:

A. Basile, A. Drexl, H. Dawid, K. Inderfurth, W. Kürsten, U. Schittko

Rainer Kleber

Dynamic Inventory Management in Reverse Logistics

With 55 Figures and 20 Tables



Dr. Rainer Kleber

Otto-von-Guericke-University Magdeburg Faculty of Economics and Management Universitätsplatz 2 39106 Magdeburg, Germany E-mail: rainer.kleber@ww.uni-magdeburg.de

ISBN-10 3-540-33229-4 Springer Berlin Heidelberg New York ISBN-13 978-3-540-33229-9 Springer Berlin Heidelberg New York

This work is subject to copyright. All rights are reserved, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilm or in any other way, and storage in data banks. Duplication of this publication or parts thereof is permitted only under the provisions of the German Copyright Law of September 9, 1965, in its current version, and permission for use must always be obtained from Springer-Verlag. Violations are liable for prosecution under the German Copyright Law.

Springer is a part of Springer Science+Business Media springer.com

© Springer-Verlag Berlin Heidelberg 2006 Printed in Germany

The use of general descriptive names, registered names, trademarks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

Typesetting: Camera ready by author Cover: Erich Kirchner, Heidelberg Production: LE-T_EX, Jelonek, Schmidt & Vöckler GbR, Leipzig SPIN 11731641 Printed on acid-free paper – 88/3100/YL – 5 4 3 2 1 0

Lecture Notes in Economics and Mathematical Systems

For information about Vols. 1–483 please contact your bookseller or Springer-Verlag

Vol. 484: D. D. Gatti, M. Gallegati, A. Kirman, Interaction and Market Structure. VI, 298 pages. 2000.

Vol. 485: A. Garnaev, Search Games and Other Applications of Game Theory. VIII, 145 pages. 2000.

Vol. 486: M. Neugart, Nonlinear Labor Market Dynamics. X, 175 pages. 2000.

Vol. 487: Y. Y. Haimes, R. E. Steuer (Eds.), Research and Practice in Multiple Criteria Decision Making. XVII, 553 pages. 2000.

Vol. 488: B. Schmolck, Ommitted Variable Tests and Dynamic Specification. X, 144 pages. 2000.

Vol. 489: T. Steger, Transitional Dynamics and Economic Growth in Developing Countries. VIII, 151 pages. 2000.

Vol. 490: S. Minner, Strategic Safety Stocks in Supply Chains. XI, 214 pages. 2000.

Vol. 491: M. Ehrgott, Multicriteria Optimization. VIII, 242 pages. 2000.

Vol. 492: T. Phan Huy, Constraint Propagation in Flexible Manufacturing. IX, 258 pages. 2000.

Vol. 493: J. Zhu, Modular Pricing of Options. X, 170 pages. 2000.

Vol. 494: D. Franzen, Design of Master Agreements for OTC Derivatives. VIII, 175 pages. 2001.

Vol. 495: I. Konnov, Combined Relaxation Methods for Variational Inequalities. XI, 181 pages. 2001.

Vol. 496: P. Weiß, Unemployment in Open Economies. XII, 226 pages. 2001.

Vol. 497: J. Inkmann, Conditional Moment Estimation of Nonlinear Equation Systems. VIII, 214 pages. 2001.

Vol. 498: M. Reutter, A Macroeconomic Model of West German Unemployment. X, 125 pages. 2001.

Vol. 499: A. Casajus, Focal Points in Framed Games. XI, 131 pages. 2001.

Vol. 500: F. Nardini, Technical Progress and Economic Growth. XVII, 191 pages. 2001.

Vol. 501: M. Fleischmann, Quantitative Models for Reverse Logistics. XI, 181 pages. 2001.

Vol. 502: N. Hadjisavvas, J. E. Martínez-Legaz, J.-P. Penot (Eds.), Generalized Convexity and Generalized Monotonicity. IX, 410 pages. 2001.

Vol. 503: A. Kirman, J.-B. Zimmermann (Eds.), Economics with Heterogenous Interacting Agents. VII, 343 pages. 2001.

Vol. 504: P.-Y. Moix (Ed.), The Measurement of Market Risk. XI, 272 pages. 2001.

Vol. 505: S. Voß, J. R. Daduna (Eds.), Computer-Aided Scheduling of Public Transport. XI, 466 pages. 2001.

Vol. 506: B. P. Kellerhals, Financial Pricing Models in Con-tinuous Time and Kalman Filtering. XIV, 247 pages. 2001.

Vol. 507: M. Koksalan, S. Zionts, Multiple Criteria Decision Making in the New Millenium. XII, 481 pages. 2001.

Vol. 508: K. Neumann, C. Schwindt, J. Zimmermann, Project Scheduling with Time Windows and Scarce Resources. XI, 335 pages. 2002.

Vol. 509: D. Hornung, Investment, R&D, and Long-Run Growth. XVI, 194 pages. 2002.

Vol. 510: A. S. Tangian, Constructing and Applying Objective Functions. XII, 582 pages. 2002.

Vol. 511: M. Külpmann, Stock Market Overreaction and Fundamental Valuation. IX, 198 pages. 2002.

Vol. 512: W.-B. Zhang, An Economic Theory of Cities.XI, 220 pages. 2002.

Vol. 513: K. Marti, Stochastic Optimization Techniques. VIII, 364 pages. 2002.

Vol. 514: S. Wang, Y. Xia, Portfolio and Asset Pricing. XII, 200 pages. 2002.

Vol. 515: G. Heisig, Planning Stability in Material Requirements Planning System. XII, 264 pages. 2002.

Vol. 516: B. Schmid, Pricing Credit Linked Financial Instruments. X, 246 pages. 2002.

Vol. 517: H. I. Meinhardt, Cooperative Decision Making in Common Pool Situations. VIII, 205 pages. 2002.

Vol. 518: S. Napel, Bilateral Bargaining. VIII, 188 pages. 2002.

Vol. 519: A. Klose, G. Speranza, L. N. Van Wassenhove (Eds.), Quantitative Approaches to Distribution Logistics and Supply Chain Management. XIII, 421 pages. 2002.

Vol. 520: B. Glaser, Efficiency versus Sustainability in Dynamic Decision Making. IX, 252 pages. 2002.

Vol. 521: R. Cowan, N. Jonard (Eds.), Heterogenous Agents, Interactions and Economic Performance. XIV, 339 pages. 2003.

Vol. 523: W.-B. Zhang, A Theory of Interregional Dynamics. XI, 231 pages. 2003.

Vol. 524: M. Frölich, Programme Evaluation and Treatment Choise. VIII, 191 pages. 2003.

Vol. 525: S. Spinler, Capacity Reservation for Capital-Intensive Technologies. XVI, 139 pages. 2003.

Vol. 526: C. F. Daganzo, A Theory of Supply Chains. VIII, 123 pages. 2003.

Vol. 527: C. E. Metz, Information Dissemination in Currency Crises. XI, 231 pages. 2003.

Vol. 528: R. Stolletz, Performance Analysis and Optimization of Inbound Call Centers. X, 219 pages. 2003.

Vol. 529: W. Krabs, S. W. Pickl, Analysis, Controllability and Optimization of Time-Discrete Systems and Dynamical Games. XII, 187 pages. 2003.

Vol. 530: R. Wapler, Unemployment, Market Structure and Growth. XXVII, 207 pages. 2003.

Vol. 531: M. Gallegati, A. Kirman, M. Marsili (Eds.), The Complex Dynamics of Economic Interaction. XV, 402 pages, 2004.

Vol. 532: K. Marti, Y. Ermoliev, G. Pflug (Eds.), Dynamic Stochastic Optimization. VIII, 336 pages. 2004.

Vol. 533: G. Dudek, Collaborative Planning in Supply Chains. X, 234 pages. 2004.

Vol. 534: M. Runkel, Environmental and Resource Policy for Consumer Durables. X, 197 pages. 2004.

Vol. 535: X. Gandibleux, M. Sevaux, K. Sörensen, V. T'kindt (Eds.), Metaheuristics for Multiobjective Optimisation. IX, 249 pages. 2004.

Vol. 536: R. Brüggemann, Model Reduction Methods for Vector Autoregressive Processes. X, 218 pages. 2004.

Vol. 537: A. Esser, Pricing in (In)Complete Markets. XI, 122 pages, 2004.

Vol. 538: S. Kokot, The Econometrics of Sequential Trade Models. XI, 193 pages. 2004.

Vol. 539: N. Hautsch, Modelling Irregularly Spaced Financial Data. XII, 291 pages. 2004.

Vol. 540: H. Kraft, Optimal Portfolios with Stochastic Interest Rates and Defaultable Assets. X, 173 pages. 2004.

Vol. 541: G.-y. Chen, X. Huang, X. Yang, Vector Optimization. X, 306 pages. 2005.

Vol. 542: J. Lingens, Union Wage Bargaining and Economic Growth. XIII, 199 pages. 2004.

Vol. 543: C. Benkert, Default Risk in Bond and Credit Derivatives Markets. IX, 135 pages. 2004.

Vol. 544: B. Fleischmann, A. Klose, Distribution Logistics. X, 284 pages. 2004.

Vol. 545: R. Hafner, Stochastic Implied Volatility. XI, 229 pages. 2004.

Vol. 546: D. Quadt, Lot-Sizing and Scheduling for Flexible Flow Lines. XVIII, 227 pages. 2004.

Vol. 547: M. Wildi, Signal Extraction. XI, 279 pages. 2005.

Vol. 548: D. Kuhn, Generalized Bounds for Convex Multistage Stochastic Programs. XI, 190 pages. 2005.

Vol. 549: G. N. Krieg, Kanban-Controlled Manufacturing Systems. IX, 236 pages. 2005.

Vol. 550: T. Lux, S. Reitz, E. Samanidou, Nonlinear Dynamics and Heterogeneous Interacting Agents. XIII, 327 pages. 2005.

Vol. 551: J. Leskow, M. Puchet Anyul, L. F. Punzo, New Tools of Economic Dynamics. XIX, 392 pages. 2005.

Vol. 552: C. Suerie, Time Continuity in Discrete Time Models. XVIII, 229 pages. 2005.

Vol. 553: B. Mönch, Strategic Trading in Illiquid Markets. XIII, 116 pages. 2005.

Vol. 554: R. Foellmi, Consumption Structure and Macroeconomics. IX, 152 pages. 2005.

Vol. 555: J. Wenzelburger, Learning in Economic Systems with Expectations Feedback (planned) 2005.

Vol. 556: R. Branzei, D. Dimitrov, S. Tijs, Models in Cooperative Game Theory. VIII, 135 pages. 2005.

Vol. 557: S. Barbaro, Equity and Efficiency Considerations of Public Higer Education. XII, 128 pages. 2005.

Vol. 558: M. Faliva, M. G. Zoia, Topics in Dynamic Model Analysis. X, 144 pages. 2005.

Vol. 559: M. Schulmerich, Real Options Valuation. XVI, 357 pages. 2005.

Vol. 560: A. von Schemde, Index and Stability in Bimatrix Games. X, 151 pages. 2005.

Vol. 561: H. Bobzin, Principles of Network Economics. XX, 390 pages. 2006.

Vol. 562: T. Langenberg, Standardization and Expectations. IX, 132 pages. 2006.

Vol. 563: A. Seeger (Ed.), Recent Advances in Optimization. XI, 455 pages. 2006.

Vol. 564: P. Mathieu, B. Beaufils, O. Brandouy (Eds.), Artificial Economics. XIII, 237 pages. 2005.

Vol. 565: W. Lemke, Term Structure Modeling and Estimation in a State Space Framework. IX, 224 pages. 2006.

Vol. 566: M. Genser, A Structural Framework for the Pricing of Corporate Securities. XIX, 176 pages. 2006.

Vol. 567: A. Namatame, T. Kaizouji, Y. Aruga (Eds.), The Complex Networks of Economic Interactions. XI, 343 pages. 2006.

Vol. 568: M. Caliendo, Microeconometric Evaluation of Labour Market Policies. XVII, 258 pages. 2006.

Vol. 569: L. Neubecker, Strategic Competition in Oligopolies with Fluctuating Demand. IX, 233 pages. 2006.

Vol. 570: J. Woo, The Political Economy of Fiscal Policy. X, 169 pages. 2006.

Vol. 571: T. Herwig, Market-Conform Valuation of Options. VIII, 104 pages. 2006.

Vol. 572: M. F. Jäkel, Pensionomics. XII, 316 pages. 2006.

Vol. 573: J. Emami Namini, International Trade and Multinational Activity, X, 159 pages, 2006.

Vol. 574: R. Kleber, Dynamic Inventory Management in Reverse Logistics, XII, 181 pages, 2006.

Vol. 575: R. Hellermann, Capacity Options for Revenue Management, XV, 199 pages, 2006 (planned).

Vol. 576: J. Zajac, Economics Dynamics, Information and Equilibrium, X, 284 pages, 2006 (planned).

Vol. 577: K. Rudolph, Bargaining Power Effects in Financial Contracting, XVIII, 330 pages, 2006 (planned).

to Kerstin, Lenny, and Lisa

Acknowledgements

I would like to acknowledge many people at the Faculty of Economics and Management at the Otto-von-Guericke University Magdeburg who have helped me during the course of this dissertation. First, I would like to express my sincere appreciation to my advisor, Karl Inderfurth, for his generous time and commitment. He offered me the chance to work with other researchers in an international network 'Reverse Logistics and its Effects on Industry', which was thankfully supported by the European Union. A substantial part of this work is a direct result of this opportunity.

I am also indebted to the readers of this thesis, Alfred Luhmer and Franz Wirl. During my graduate studies, Franz Wirl opened my eyes to the depths of Operations Research, more specifically that of dynamic optimization problems. I would like to thank Alfred Luhmer for his encouragement and interest in this subject as well as his valuable comments.

I have enjoyed the company of two co-authors in several research papers, namely Stefan Minner and Gudrun P. Kiesmüller. Stefan in particular was always an invaluable source of ideas and knowledge. I extend many thanks to my current and former colleagues for both their research-oriented focus and a nice working atmosphere. In particular, I am thankful to Gerald Heisig, Ian M. Langella, Ivo Neidlein, and the late Gerd Lindner for numerous scientific discussions and their friendship. Furthermore, Ian M. Langella and Dirk Matzner considerably improved readability of this work by enhancing my English.

Finally I would like to express my gratitude to my family and friends, for without their constant support this work would not have been possible. This is especially true for my Mother, who gave me the best start into a scientific career that could be imagined. To my wife and best friend, Kerstin, I wish to express my heartfelt gratitude for her understanding and patience, but also for our wonderful children, Leonard and Elisabeth, that keep our lives in balance.

Preface

Reverse Logistics is an area that has attracted growing attention over the last years both from the industrial as well as from the scientific side. The proper management of reverse flows of products and materials is of considerable importance in many industries because of its influence on economic performance and environmental impact. The respective management tasks, however, are connected with new challenging planning and control problems. This especially holds for product recovery management concerning remanufacturing operations where used products, after being returned to the manufacturer, are reprocessed such that they are as good as new and can be re-integrated into the forward logistics stream.

A major issue in remanufacturing is how to optimally coordinate the potential activities directed at meeting customer demands for serviceable products and to deal with returns of products after end-of-use. The respective decisions refer to finding a proper mix of manufacturing original and remanufacturing used products as well as of stock-keeping and disposing of returned items. Hereby, relevant cost impacts and time patterns of demand and returns have to be taken into consideration.

Up to now, research contributions to this field of Reverse Logistics have addressed only two main aspects that result in high complexity of decision making in product recovery management. One aspect is that of capacity restrictions and fixed costs in manufacturing and remanufacturing systems that makes coordination of lot-sizing a challenging problem. The second aspect refers to uncertainty of demands and returns that leads to complicated stochastic interactions which have to be coped with by appropriate decision rules and safety stock policies. While these issues are highly relevant for operational and tactical decision making, a third aspect with mainly strategic importance has largely been ignored. This is the aspect of time-variability and dynamic change of major input parameters for product recovery decisions. On the one hand, this refers to the variability of product demand and return schemes that can be observed both due to seasonality and the classical life cycle pattern for many product categories. On the other hand, over larger time spans we also face specific cost dynamics caused by experience effects in manufacturing and remanufacturing processes.

It is the commendable contribution of this book that it sheds some light into this complicated field of how to respond most effectively to the dynamically changing environment in product recovery strategy. This response refers to choice of time-varying coordination strategies of manufacturing, remanufacturing and disposal activities as well as to the timing of investment decisions in product recovery technologies. Embedded in these considerations an analysis is developed of how and why to use different kinds of strategic inventories to enable best reactions to dynamic cost, demand and return processes. Based on advanced quantitative modeling and optimization techniques a deep analysis of the addressed complex dynamic decision problems is given.

Summarizing, this book presents major progress in scientifically investigating the field of complex problems of product recovery management induced by several types of dynamics in the planning environment. The underlying dynamic problem aspects are of enormous practical importance, but have not been addressed appropriately in research contributions before. By studying this book the reader will learn novel and interesting findings on how to respond strategically to ongoing changes of a product recovery environment by responsive recovery policies and dynamic inventory management.

Magdeburg, April 2006

Karl Inderfurth

Contents

1	Inti	roduct	ion	1						
	1.1	Objective and Motivation 1								
	1.2	Invent	tory Management in Reverse Logistics	3						
	1.3	Metho	odology	6						
	1.4	Outlin	ne of the Thesis	7						
2	A Basic Quantitative Model for Medium and Long Range									
	\mathbf{Pro}	Product Recovery Planning								
	2.1		'iew	9						
	2.2	A Bas	sic Quantitative Model of Dynamic Product Recovery	9						
	2.3	Solution of the Model Without Initial Inventories								
		2.3.1	Solution Methodology	13						
		2.3.2	Necessary Conditions	14						
		2.3.3	The Structure of an Optimal Solution	17						
		2.3.4	Optimal Transitions Between Cases and Subcases	20						
		2.3.5	Properties of Optimal Return Collection Intervals	22						
		2.3.6	Solution Algorithm	25						
	2.4	Dealir	ng with Initial Inventories	27						
	2.5	Nume	nerical Examples							
	2.6	Comp	omparison of Holding Cost and Discounted Cash Flow							
		2.6.1	Holding Cost Results	34						
		2.6.2	-	35						
	2.7	Mana	anagerial Insights							
	2.8	Proof	s and Derivations	40						
3	On	the Ef	fects of Capacity Constraints in Product Recovery	47						
	3.1	Motiv	ation	47						
	3.2	2 Limited Production Capacity								
		3.2.1	Changes to the Basic Model	49						
		3.2.2	Properties of an Optimal Solution	49						
		3.2.3	Pure Effects of a Manufacturing Constraint	57						

		3.2.4Mixed Effects of a Manufacturing Constraint
		3.2.6 Managerial Insights
	3.3	Limited Remanufacturing Capacity
		3.3.1 Changes to the Basic Model
		3.3.2 Properties of an Optimal Solution
		3.3.3 Pure Effects of a Remanufacturing Constraint
		3.3.4 Managerial Insights
	3.4	Proofs
4	Kno	weldge Acquisition and Product Recovery
	4.1	Motivation
	4.2	A Model with Remanufacturing Knowledge Acquisition 98
	4.3	Optimality Conditions and General Results
	4.4	Optimal Policies in Specific Situations108
		4.4.1 Optimal Policy with a Zero Interest Rate
		4.4.2 Optimal Policy with a Positive Interest Rate
	4.5	Numerical Examples
	4.6	Conclusions and Managerial Insights121
	4.7	Proofs
5	Tec	hnology Selection in the Context of Reverse Logistics 127
	5.1	Motivation
	5.2	A Dynamic Modeling Environment for Strategic Decision
		Making
	5.3	Three Investment Projects
		5.3.1 Valuation of Investment Project (a) - Design for Single
		Use
		5.3.2 Investment Project (b) - Design for Reuse
		5.3.3 Investment project (c) - Design for Reuse with
	- 1	Strategic Inventory
	5.4	A Numerical Investigation
	5.5	Effects of a Limited Remanufacturing Capacity
	5.6	Conclusions and Possible Extensions
	5.7	Proofs and Statistical Tests
6	Con	clusions
List	t of S	Symbols
\mathbf{Lis}^{\dagger}	t of l	Figures
\mathbf{Lis}^{\dagger}	t of 7	Fables
Ref		

Introduction

1.1 Objective and Motivation

The integration of product recovery into regular production processes has developed into a challenge for the manufacturing industry (Guide and Van Wassenhove (2002)). While in the past a firm's main concern was to sell its products leaving the burden of final disposal to society, it is now increasingly assigned responsibility for what happens with the product after use. Consequently, product recovery leads to additional restrictions firms must take into account, but it also enables new opportunities (Stock et al. (2002)).

There are many reasons for this development. An increasing environmental consciousness of the public and limited availability of natural resources to manufacture new products on one side and the necessity to find alternatives to landfilling and incineration of waste led to new regulations that aim at reducing the quantity and environmental impact of waste. Environmental legislation incorporates the prohibition of substances that aggravate material recovery, the enforcement of collection networks, and industry specific take back and recovery obligations. Some of the many examples are the German Recycling and Waste Control Act (Kreislaufwirtschafts- und Abfallgesetz, KrW-/AbfG) enacted in 1996 that extended product responsibility of manufacturers to the end of life phase and the EU Directive on Waste Electric and Electronic Equipment (WEEE) from 2003 which calls for the installment of collection networks. A recent overview on end-of-life legislation issues with examples from the US, Europe, and Japan can be found in Toffel (2003).

In addition, economic motives lead to a voluntary product take back of original equipment manufacturers (OEMs), as detailed and classified in Toffel (2004). First, recovering products allows us to reduce production cost by using recovered material and components in lieu of virgin material and newly produced components. Second, the fact that there is demand for leased products in the marketplace forces us to confront these products again at the end of the lease period. In this example, dealing with returns is part of the 'price we pay' to service the demand for these products. Third, customer behavior seemed to be influenced by the environmental image of firm, and therefore using recovered material in products or merely engaging in product recovery itself increases the demand from this market segment. Fourth, aftermarkets are often lucrative revenue generators, and therefore must be protected against third parties servicing demand for spare parts, etc. Finally, it is also viewed that taking environmentally friendly steps, for instance implementing a product recovery system before take-back laws exist, at times successfully preempts environmental legislation. On the other hand, Reinhardt (1999) also points out that encouraging environmental legislation can lead to an improved position of the firm, forcing competitors into compliance.

An increasing return of used products encourages OEMs to produce more environmental friendly yet recoverable products. A large number of examples for product recovery due to varying incentives are assembled by de Brito et al. (2005) and range from the reprocessing of chemicals in the pharmaceutical industry (see Teunter et al. (2005)) to the remanufacturing of Kodak single-use cameras or of complex products like engines in the car manufacturing industry (the latter two examples will be further detailed below as case studies). Thierry et al. (1995) put forth an overview on strategic product recovery issues and differentiate between product recovery options recycling, repair, cannibalization, refurbishing, and remanufacturing. Out of these options, remanufacturing seems especially appealing to OEMs since large parts of the added value can be recovered (Klausner and Hendrickson (2000)).

Product recovery management is charged with the coordinated planning and control of both production and recovery processes that serve the same demand for materials, parts, or final products. In the context of remanufacturing, both sources are assumed to be perfect substitutes, and recovered products are usually said to be as good as new. It should be noted that although we might choose other recovery options (e.g. repair or refurbishing) to be performed on the returns, we restrict our attention to remanufacturing as we presume substitutability.

When dealing with product returns, logistic processes are more complicated to control since both forward and backward flows must be coordinated. Production planning is more complex since there now exist two possible sources to serve the demand which need to be coordinated, therefore raising new operational questions. These problems receive growing interest from researchers and practitioners alike and are summarized in the field of *Reverse Logistics*. The European Working Group on Reverse Logistics (REVLOG) uses the following definition:

The research area of Reverse Logistics covers "the process of planning, implementing and controlling backward flows of raw materials, in process inventory, packaging and finished goods, from a manufacturing, distribution or use point, to a point of recovery or point of proper disposal." (de Brito and Dekker (2004)) Quantitative approaches in reverse logistics have been surveyed by Fleischmann et al. (1997) and more recently by Dekker et al. (2004). According to the example of the latter work one can distinguish between three important domains of research within this field, namely: extended supply chain management dealing with relations between different partners inside a reverse logistics system, reverse distribution which includes collection and transportation aspects, and production and inventory management. Here, we focus on the last aspect and assume that an appropriate collection network exists which provides an OEM access to its own used products.

1.2 Inventory Management in Reverse Logistics

There are several reasons to keep stock in traditional production settings, as discussed in Silver et al. (1998), Chapter 3, and inventories can be classified based on their economic motivation. Specifically, safety stock is used to buffer from short term uncertainty of demand and supply, cycle stock is used to account for trade-offs between e.g. fixed setup and holding costs, and anticipation stock is often used to smooth capacity utilization in a dynamic (e.g. seasonal) environment.

Managing inventory in the presence of returns leads to additional complexity. In the case of safety stock, we must now account (in addition to the traditional demand uncertainty) for the uncertainty surrounding the supply of returns, whereas in lot sizing we must coordinate lot sizes and setup times for both production and remanufacturing. Stocks have to be distributed among inventories for serviceables and recoverables (returns). These issues have received attention in research (for reviews on inventory management in reverse logistics see Dekker and van der Laan (2003), de Brito and Dekker (2003) and Fleischmann and Minner (2004)), but they hardly explain the large amount of returned used products held in stock at remanufacturing facilities, as is confirmed by Seitz and Peattie (2004). When adapting our treatment of anticipation stock, we find that the addition of the return stream yields entirely new situations in which we hold stock, which directly result from the dynamic environment firms operate in.

A closer look at the product recovery environment reveals many factors which fluctuate over time. Starting with an obvious one, the demand for the product will vary over time. However, this is no surprise. Frequently, life cycle patterns as well as seasonality will influence demand. In medium-range aggregate production planning (Silver et al. (1998), Chapter 14), we seek to smooth capacity utilization by using seasonal inventory. The resulting solution lies between two extremes of nearly constant production (level) and production which is synchronized with demand (chase). The amounts of returned products may likewise vary over time, as is documented in the following two cases:

Case 1.1. DaimlerChrysler engines (see Kiesmüller et al. (2004))

DaimlerChrysler operates several facilities for recovering parts from used cars, one of which remanufactures used engines for Mercedes Benz cars at the plant Berlin-Marienfelde (MTC). Annually, about 12,000 engines from 28 classes and 800 different model variants are remanufactured. An ABC-classification revealed that 60% of the returns are contributed by 3 classes.

Dynamic issues, i.e. time dependent demands and returns, have to be considered for two reasons. First, demands for an engine class follow the shape of a product life cycle, starting with a phase of increasing sales, followed by the maturing phase and finally declining sales towards the end of a product's life cycle. Returns follow demands in a similar pattern, delayed by the usual life time of an engine and reduced by the number of not returned engines. In the growth phase, demand for remanufactured engines is significantly higher than available returns and all returned cores are remanufactured. Later in the maturing phase, demand decreases and returns can exceed remanufacturing orders. This divides the product life cycle into two main phases, the first with insufficient cores and another one with excessive cores. Similar effects, i.e. dynamic fluctuations of both, demand for remanufactured items and supply of returns, leading to shortage and overage situations have also been reported for car part remanufacturing at Volkswagen (van der Laan et al. (2004)).

Case 1.2. Kodak single-use cameras (see Goldstein (1994) and Guide et al. (2003b))

Introduced by Fuji Photo Film Co. as 'film with lens' and originally designed to be thrown away after use, the single-use camera now is another example of successfully closing the loop on a higher level of product recovery. According to Kodak (2003), about 775 Million cameras have been processed since the start of the product recovery program in 1990 and currently a worldwide return rate of 75% has been achieved. The amount of reusable materials ranges between 77-90% (by weight) of the product. Most recovered parts are plastic bodies, which are reused up to six times and the circuit boards required in flash cameras, which are used up to 10 times.

An important issue that Kodak faces is to deal with dynamic demand and return streams. Goldstein (1994) stated that there is "a lot of seasonality and cyclical behavior" in the market for single-use cameras. This stems from peak selling periods that differ among the various models: Underwater cameras mostly are sold in summer and winter vacation season, while flash cameras sell best around winter holiday season, and peak season for single-use cameras is between March and early September. On the reverse flow side, batch shipments from smaller photofinisher labs to collection facilities can lead to a delay of a couple of months between three and five months for a camera to be returned after being (re)manufactured. Although these numbers have been reduced in recent years, a large number of used products return at the end of a peak season or during off-season. As the two cases have shown, there will be periods where returns exceed demand (excess returns) and other moments where demand exceeds returns (excess demand). Since available returns can also be seen as a capacity to recover products, an inventory can be used to enlarge the capacity when needed. Fleischmann and Minner (2004) call such stocks 'opportunity stocks', because they enable additional recovery opportunities. From a more strategic point of view, cost parameters themselves can change over time caused not only by external influences (such as ever-increasing disposal fees) but also by internal impacts like learning (or experience) curve effects. Lastly, in the long run, available capacity for product recovery is also not fixed over time but can be changed through capacity expansion or reduction.

Most of product recovery and inventory management models are either restricted to stationary conditions or treat dynamic aspects only numerically, for recent overviews on stochastic inventory control see van der Laan et al. (2004) and for lot sizing issues Minner and Lindner (2004). As a consequence, Dekker et al. (2000) suggest more examination of the effects of non-stationary demand/return conditions on inventory control for joint manufacturing remanufacturing systems.

This thesis concerns itself with the incorporation of dynamic issues in medium and long-term product recovery management. In doing so, we expressly ignore more operative disassembly issues, which would complicate matters. We also restrict ourselves to time-varying deterministic environments, ignoring short term stochastic fluctuations as is common practice in other medium to long term models. We avoid more unnecessary complication by examining the simplest case of a single product or a single part/module. Decisions faced in this realm include (1) when to invest in remanufacturing capabilities (if at all), (2) when to start collecting, hold stock of, and dispose of returns. It specifically deals with use of anticipation inventories for smoothing both capacity supply (e.g. return availability) as well as capacity demand. Strategic implications are expressly considered, especially the decision of whether to engage in a higher level of recovery or not, including aspects such as knowledge acquisition (e.g. experience curve effects) and the additional operational and investment expenditures required to implement product recovery processes (Toffel (2004)). The consideration of more strategic issues in research has been recently demanded by Guide et al. (2003a) since it is seen to be of particular value for practitioners. Long term decisions involve significant sums of money and are often difficult (if not impossible) to change. Examining our problem specifically, we can see that the decision on when to invest if made erroneously could result in opportunity costs rising from either lost recovery cost advantage (if made late) or capital costs (if made early). Investment on product recovery capability is a decision made very carefully by managers, and one that they certainly do not want to get wrong. Likewise, deciding on the correct time to start keeping excess returns is also important and has far reaching effects. If we start too early, we sacrifice capital costs filling our inventory with unusable scrap. This error would be particularly

painful if we could have reaped a salvage value by 'disposing' of the returns. Waiting too long, on the other hand, results in lost recovery cost advantage.

1.3 Methodology

An appropriate way to examine long term issues is to use a continuous time model, which avoids the discretization of time and the influence of the choice of time units on the model and results. Another advantage of this modeling is that parameters can be given by continuous time functions, eliminating the need to specify them for each time period. Dynamic modeling properties motivates the use of the theory of *Optimal Control* as a solution method. Starting with the pioneering work by Pontryagin et al. (1962) it has reached a wide range of applications in economics and management, see e.g. Seierstad and Sydsæter (1987), Kamien and Schwartz (1991), or Sethi and Thompson (2000). It can be compared with dynamic programming methods developed by Bellman (1957) at about the same time but, according to Feichtinger and Hartl (1986), a main advantage of optimal control is the possibility of gaining insights into the general structure of solutions for an entire problem class.

Although there are extensions to solve discrete time problems, optimal control literature mainly deals with continuous time systems. A system in this sense is characterized by one or more state variables (e.g. an inventory stock level) which are changed by external influences (demand) or by choosing control variables (production). The development of the states is characterized by a differential equation named *state (transition) equation*. Both, state as well as control variables, can be subject to constraints which have to be considered. An optimal solution is given by optimal trajectories (functions of time) of the state and control variables which maximize (or minimize) a given objective function.

Pontryagin's Maximum Principle provides a set of necessary conditions for an optimal solution, which is also sufficient under certain conditions. Using the Maximum Principle the dynamic problem is decomposed into an infinite sequence of interrelated static problems, one for each time instant. These are connected by introducing co-state (also called adjoint) variables, which can interpreted as the shadow price of changing the system state. A static objective function, called *Hamiltonian*, is constructed in a way that it measures the total effects of the decisions at a certain time point on the objective. These can be split into a direct and an indirect effect. The direct effect is given for instance by the costs of producing an item. The indirect effect arises since decisions also have an influence on future opportunities by changing the system's state, e.g. by decreasing or increasing the inventory level. It is measured by the rate of change of the state times the corresponding shadow price (given by the co-state). This yields another advantage of optimal control since the co-states can also be interpreted as the value of e.g. another returned item, a produced, or remanufactured product.

As in dynamic programming, where an optimal decision is taken at each stage (instant of time) assuming that up to that point all decisions have been taken optimally and the same will hold for future decisions, the Hamiltonian has to be maximized at each instant of time by appropriately choosing the controls for given optimal state and co-state values, subject to relevant constraints on control and state variables. This is applied by using standard methods of non-linear programming. Further necessary conditions contain the rate of change of the co-state variables. Thus, a system of differential equations including the state transitions, co-state transitions, and optimal control policies has to be solved.

For an overview on traditional continuous time production and inventory models see e.g. Feichtinger and Hartl (1986), Chapter 9 or Sethi and Thompson (2000), Chapter 6. Well known examples are variants of the HMMS-model (Holt et al. (1960), Thompson and Sethi (1980)) that uses a quadratic objective in order to retain goal levels for both inventory and production. Linear inventory and convex production costs are used in Arrow-Karlin type models (Arrow and Karlin (1958)). More recently, these models have been extended to cope with environmental issues. Wirl (1991) and Hartl (1995) analyzed effects of environmental constraints in the Arrow-Karlin model and Dobos (1998) used the HMMS approach. Product recovery systems including remanufacturing and disposal of returned products under non-linear cost regimes are covered e.g. by Kistner and Dobos (2000). In most practical situations, however, a linear cost regime is present and will be used throughout this work. Recent applications of optimal control in dynamic product recovery are reviewed by Kiesmüller et al. (2004).

1.4 Outline of the Thesis

The road-map followed in the succeeding chapters is given as follows.

In *Chapter 2*, a basic model for product recovery is presented. It aims to explain under which conditions returns should be kept in an anticipation stock. It extends the investigation of a single product/single stage product recovery system by Minner and Kleber (2001) by allowing for discounting. Some attention is paid to the valuation of inventories which in this case can be quite easily accomplished through exploiting the advantages of the solution methodology.

In traditional medium-term aggregate production planning an anticipation stock is used when capacity of the cheaper regular mode becomes binding in order to avoid high costs of overtime. However, the second supply source might also be limited. Therefore, *Chapter 3* discusses the implications of capacity constraints on both the cheaper source (remanufacturing) and the 'overtime' mode (production).

Chapter 4 relaxes the assumption of general preferability of remanufacturing over manufacturing. Knowledge acquisition during repeated remanufacturing operations can itself lead to profitable remanufacturing, even if there is no immediate cost advantage. The influence of learning in the remanufacturing process on stock-keeping decisions is analyzed, revealing that (under certain circumstances) another motive for holding recoverables is to postpone the start of remanufacturing.

Chapter 5 deals with the use of anticipation inventory in controlling remanufacturing capacity over the product life cycle in the most simple case, the choice of the investment time of a remanufacturing facility with unlimited capacity. More specifically, when introducing a new product, two related decisions have to be considered, namely product design and the choice of the recovery mode and a corresponding technology. This is accomplished by considering the influences of such decisions on direct production costs and initial investment expenditures. Taking into account the limited availability of used products in the beginning of a products life cycle and a decreasing time value of the required investment expenditures connected with the set-up of the remanufacturing process, the issues addressed here are when to initiate this process and how the use of a strategic recoverables inventory does affect this decision.

Some concluding remarks are given in *Chapter 6*, along with a recapitulation of the main results of the thesis, as well as a short discussion of related work.

A Basic Quantitative Model for Medium and Long Range Product Recovery Planning

2.1 Overview

This chapter deals with a basic dynamic model of product recovery. A singlestage and single item version of a product recovery system is presented and analyzed under a linear cost regime. The remainder of this chapter is organized as follows. The dynamic optimization problem is formulated (Section 2.2) and solved (Section 2.3). Thereby, some general properties of an optimal recovery strategy and a construction method are outlined. In the first instance, this is accomplished by disregarding initial inventories. However, initial recoverable stocks play a role in rolling planning frameworks, and an explicit consideration of positive serviceables will facilitate our discussion in the case of limited capacities (see Chapter 3). We therefore present a method on how to deal with initial inventories in Section 2.4. The findings are illustrated using several numerical examples in Section 2.5. A comparison with the outcome of using an undiscounted (average) cost approach is provided in Section 2.6. The main results are summarized in Section 2.7.

2.2 A Basic Quantitative Model of Dynamic Product Recovery

In the following we analyze a generalized version of the single-product, singlestage product recovery system presented by Minner and Kleber (2001) that allows for discounting. The system under consideration is depicted in Figure 2.1. The system faces an external customer demand rate d(t). Customer return used products with rate u(t). Both rates are deterministic, non-negative, and continuous functions within a finite planning horizon [0, T], and can not be influenced by the decision maker.

As an example consider the following functions for cyclical demand and returns over a planning horizon of $T = 4\pi$:



Fig. 2.1. Product recovery system

 $d(t) = 1 + 0.5 \sin(t)$ and $u(t) = 0.7d(t - \pi)$

The demand function consists of a time independent base level of 1 which is superimposed by some seasonal influence introduced by a cyclical part $0.5 \sin(t)$. A fraction of 70% of previous demand becomes available after staying with the customer for a period of π . Here, it is additionally assumed that there existed demand before the begin of the planning period, i.e. for t < 0. This scenario is depicted in Figure 2.2.

Product requirements are satisfied from serviceables inventory which can either be replenished by producing new items with rate p(t) or by remanufacturing of returned products with rate r(t). We assume that remanufactured products have the same quality as produced items and therefore serve as perfect substitutes. Returned products that are not instantaneously remanufactured can either be kept in a recoverables inventory for future remanu-



Fig. 2.2. Demands and returns in the example

facturing or disposed of with rate w(t). There are no binding constraints on either process, i.e. it is possible to produce and to remanufacture at any rate. Additionally, the disposal level is assumed not to be limited.

The state of the recovery system is described by a serviceables $y_s(t)$ and a recoverables inventory $y_u(t)$. Initial values for the state variables are given by $y_s(0) = y_s^0$ and $y_u(0) = y_u^0$. The planning horizon T characterizes the end of all obligations for the product. Therefore, both inventories have to be zero at time T, i.e. $y_s(T) = 0$, $y_u(T) = 0$.

Since we assume zero lead times, decisions are implemented instantaneously. Thus, the corresponding state transition equations are represented by differential equations where the rate of change in the serviceables inventory $(\dot{y}_s(t) = dy_s(t)/dt)$ equals the sum of production and remanufacturing rate minus demand rate at time t

$$\dot{y}_s(t) = p(t) + r(t) - d(t).$$
 (2.1)

Analogously, the marginal increase of recoverables inventory is given by the return rate minus the sum of remanufacturing and disposal rates at time t

$$\dot{y}_u(t) = u(t) - r(t) - w(t).$$
 (2.2)

Since we do not allow for backlogging, the serviceables inventory has to be non-negative

$$y_s(t) \ge 0. \tag{2.3}$$

The availability of recoverable items limits both, remanufacturing and disposal decisions, being expressed by non-negativity of the recoverables inventory

$$y_u(t) \ge 0. \tag{2.4}$$

Further, non-negativity constraints hold for production, remanufacturing, and disposal control variables

$$p(t) \ge 0, \quad r(t) \ge 0, \quad w(t) \ge 0.$$
 (2.5)

The objective of the model is to satisfy customer demands over the planning horizon with a minimal total discounted cash (out-)flow for production, remanufacturing, disposal and holding serviceables and recoverables inventory. It is assumed that all payments depend linearly on the respective decision, i.e. we assume constant and time independent per unit payments for production $c_p > 0$, remanufacturing $c_r > 0$, and disposing items c_w , as well as constant out-of-pocket inventory holding costs per unit and time unit $h_s > 0$ and $h_u > 0$, respectively. The discount rate $\alpha > 0$ reflects real costs of capital (net of inflation), possibly calculated as weighted average of the companies equity and debt capital (see, e.g., Kaplan and Atkinson (1998)). In contrast to all other parameters, unit payments for disposing of a returned item can also be negative, as would be the case if a salvage revenue can be obtained. In order to ensure that remanufacturing is an alternative to production, direct costs of remanufacturing must be less than the sum of direct production and disposal costs per unit

$$c_p + c_w > c_r. \tag{2.6}$$

Otherwise, disposal of a returned product and production of a new one is always superior to remanufacturing it, and the model would reduce to a pure production/inventory model with linear direct costs, for which it is optimal to produce the demand at every time point (see Feichtinger and Hartl (1986)) after consuming initial inventories. The difference $(c_p + c_w) - c_r$ is therefore referred to as (direct) recovery cost advantage.

Further we assume, that out-of-pocket holding cost per item and time unit for serviceables is higher than for recoverables

$$h_s > h_u, \tag{2.7}$$

which is reasonable, because serviceables usually are treated more carefully than recoverables. Finally, it is not advantageous to hold unneeded returned products as opposed to disposing of them

$$h_u > \alpha c_w. \tag{2.8}$$

If this would not be the case, disposal would never take place because interests saved when delaying the disposal of an item would always outweigh out-ofpocket holding costs.

Both equations (2.1) and (2.2) imply that production, remanufacturing and disposal are executed with finite rates. But since we assume all controls (processes) to be unrestricted, state variables are allowed to jump, e.g. if there are more recoverables in inventory than needed, one can dispose of a real quantity by an impulse control, which leads to a negative jump in y_u . A (positive) jump in y_s by a production or remanufacturing impulse is more expensive than synchronizing production and remanufacturing with demand because of the additional holding costs incurred. A jump in y_u can only be advantageous at time zero by disposing a quantity w_0 with an impulse control. For later time points, disposal in advance reduces recoverables holding costs. The following proposition summarizes the sketched results on impulse controls.

Proposition 2.1. A (positive) jump in the serviceables inventory by production or remanufacturing impulse controls will never occur in an optimal policy. A (negative) jump in the recoverables inventory may be advantageous only at time zero.

A note on **Proofs.** For all proofs and derivations see the appendices of the respective chapter.

In the following, $x(t^-)$, $x(t^+)$ denote left- and right-side limits of x, respectively. With the introduced assumptions, the following control optimization model with two states $(y_s \text{ and } y_u)$, three control variables (p, r, w) and a single

impulse control variable (w_0) has to be solved subject to the state equations and one jump equation, two pure state constraints, initial and terminal conditions for the state variables as well as non-negativity constraints for all control variables

$$\min NPV = c_w w_0 + \int_0^T e^{-\alpha t} \left(c_p p(t) + c_r r(t) + c_w w(t) + h_s y_s(t) + h_u y_u(t) \right) dt$$

s.t. $\dot{y}_s(t) = p(t) + r(t) - d(t), \dot{y}_u(t) = u(t) - r(t) - w(t),$
 $y_s(t) \ge 0, y_s(0) = y_s^0, y_s(T) = 0,$
 $y_u(t) \ge 0, y_u(0^-) = y_u^0, y_u(T) = 0,$
 $y_u(0^+) - y_u(0^-) = -w_0,$
 $p(t) \ge 0, u(t) \ge 0, w(t) \ge 0, 0 \le w_0 \le y_u(0^-).$
(2.9)

2.3 Solution of the Model Without Initial Inventories

2.3.1 Solution Methodology

The dynamic optimal control problem (2.9) is solved by applying the direct adjoining approach for Pontryagin's Maximum Principle as shown in Feichtinger and Hartl (1986) for a deterministic, linear, non-autonomous model with pure constraints both in state and control variables and a possible jump in a state. Because of the linear objective function, the necessary conditions given below are also sufficient for an optimal policy.

The solution process as sketched in Section 1.3 requires to solve a system of differential equations which is complicated for several reasons. First, optimal trajectories of control variables are in general not given as closed form expressions and in case of unrestricted controls, there might even be 'jumps' in the slope of the state variables. Second, there might be no predetermined initial or terminal value for the co-state. And finally, co-state variables can be discontinuous under certain conditions. A construction of the optimal trajectory can therefore be arduous. In order to overcome these difficulties, the following general approach is used. Exploiting necessary conditions presented in Section 2.3.2, general properties of an optimal policy can be derived in Section 2.3.3 by defining *cases* with respect to state variables and deriving respective optimal decision structures. Since there can also be other influencing conditions, further sub-cases might be distinguished. Within such defined intervals closed form expressions for the control, state and co-state functions can be stated. Since such a case or policy will in general not be optimal for the whole planning horizon, in Section 2.3.4 it is determined under which conditions a *transition* from one case to another one occurs. These points in time are identified as so called *transition points*. Then, properties of optimal intervals where stock-keeping is present are derived in Section 2.3.5 and finally

a solution algorithm is given in Section 2.3.6 that numerically determines the optimal solution.

For ease of the following discussion we first restrict to the case where both initial inventories are equal to zero $(y_s^0 = y_u^0 = 0)$, and therefore the jump parameter w_0 must be equal to zero. In Section 2.4 we show what changes when initial stocks have to be considered. For the remainder of this chapter, we omit time indices where appropriate for reasons of simplicity.

2.3.2 Necessary Conditions

There are two equivalent possibilities to formulate necessary conditions. The first way is to use the so-called *present-value formulation* in where the static objective (Hamiltonian) explicitly includes the discount factor $e^{-\alpha t}$. Instead of that, we prefer to state conditions by using the so-called *current-value formulation* which has the advantage that both, the static objective and the static optimality conditions do not change in time.

The current-value Hamiltonian $H(y_s, y_u, p, r, w, t, \lambda_0, \lambda_s, \lambda_u) = H(.)$ for a maximization problem corresponding to (2.9) is given by multiplication of the (negative) objective by a constant λ_0 and the state transitions by adjoint variables $\lambda_s(t)$, $\lambda_u(t)$ respectively

$$H(.) = \lambda_0 (-c_p p - c_r r - c_w w - h_s y_s - h_u y_u) + \lambda_s (p + r - d) + \lambda_u (u - r - w).$$
(2.10)

In the appendix we show that $\lambda_0 = 1$. Then, the adjoints or co-state variables $\lambda_s(t)$ and $\lambda_u(t)$ are unequivocally defined and can be interpreted as shadow price of a change in the corresponding inventory level. Since there are several ways to change the inventory level, the co-states also have the following meaning: $\lambda_s(t)$ is the value of a produced item or the negative value of an additionally demanded item at t. $\lambda_u(t)$ can be seen as value of another returned item or negative value of a disposed item at t. $\lambda_s(t) - \lambda_u(t)$ can be interpreted as value of a remanufactured item. We will also refer to the adjoints as the value of a stored item.

In order to account for the control and state inequality constraints, Lagrange multipliers $\mu_i(t)$, i = 1, 2, 3 for the three control variables and $k_j(t)$, j = 1, 2 for the two state variables are defined. The corresponding Lagrangian $L(., \mu_1, \mu_2, \mu_3, k_1, k_2) = L(..)$ is given by

$$L(..) = H(.) + \mu_1 \cdot p + \mu_2 \cdot r + \mu_3 \cdot w + k_1 \cdot y_s + k_2 \cdot y_u.$$
(2.11)

The derivation of the necessary conditions for the optimal solution is made in accordance with the methodology put forth in Feichtinger and Hartl (1986), pp. 164-169.

Let (y_s^*, y_u^*) represent the optimal trajectory of the state variables and (p^*, r^*, w^*) the (piecewise continuous) trajectory of optimal control policies

to problem (2.9). Then, there exist piecewise continuous functions of time λ_s , λ_u , μ_i , i = 1, 2, 3, k_j , j = 1, 2 and two sets of time points $\theta_s \in \Theta_s$ and $\theta_u \in \Theta_u$ where the co-states λ_s and λ_u jump with corresponding height parameters $\eta_s(\theta_s)$ and $\eta_u(\theta_u)$. Except for points of discontinuity in the controls and junction points, i.e. points where one of the non-negativity constraints for inventories become or leave the state of being binding, the following necessary conditions (2.12)–(2.24) have to hold.

Due to its linearity in all controls, the Hamiltonian is maximized, if

$$p^* = \begin{cases} 0 \quad \lambda_s < c_p \\ \text{singular } \lambda_s = c_p \\ \infty \quad \lambda_s > c_p \end{cases}$$
(2.12)

$$r^* = \begin{cases} 0 \quad \lambda_s - \lambda_u < c_r \\ \text{singular } \lambda_s - \lambda_u = c_r \\ \infty \quad \lambda_s - \lambda_u > c_r \end{cases}$$
(2.13)

$$w^* = \begin{cases} 0 \quad \lambda_u > -c_w \\ \text{singular } \lambda_u = -c_w \\ \infty \quad \lambda_u < -c_w \end{cases}$$
(2.14)

Note that infinite values for the controls can be excluded as shown below. The so-called bang-bang equations (2.12)–(2.14) can be interpreted as follows. If production costs c_p exceed the value of a stored serviceables item λ_s , the production rate obtains its lower bound of zero. If unit production cost and adjoint variable are equal, the production rate can be positive but it is not to be determined from this condition alone ('singular'). Analogously, the remanufacturing rate is zero if the increase in value of a used product when remanufacturing it (value of a serviceables item minus the value of the recoverable item) is smaller than remanufacturing cost c_r and positive when both are equal. If the value of a returned item is larger than its salvage revenue $(-c_w)$, no items are disposed of.

Further necessary conditions account for inequality conditions for the control variables. Equations (2.15)-(2.17) maximize the Lagrangian. Non-negativity as well as complementary slackness conditions (2.18)-(2.20) must apply:

$$\frac{\partial L}{\partial p} = -c_p + \lambda_s + \mu_1 = 0 \tag{2.15}$$

$$\frac{\partial L}{\partial r} = -c_r + \lambda_s - \lambda_u + \mu_2 = 0 \tag{2.16}$$

$$\frac{\partial L}{\partial w} = -c_w - \lambda_u + \mu_3 = 0 \tag{2.17}$$

$$\mu_1 \ge 0 \quad \mu_1 \cdot p^* = 0 \tag{2.18}$$

- $\mu_2 \ge 0 \ \ \mu_2 \cdot r^* = 0 \tag{2.19}$
- $\mu_3 \ge 0 \ \ \mu_3 \cdot w^* = 0 \tag{2.20}$

From (2.15)–(2.17) together with (2.18)–(2.20) it follows that the infinite upper values for p^* , r^* and w^* in (2.12)–(2.14) cannot occur in an optimal solution because $\lambda_s \leq c_p$ from (2.15) and (2.18), $\lambda_u \geq -c_w$ from (2.16) and (2.19), and $\lambda_s - \lambda_u \leq c_r$ from (2.17) and (2.20).

Equations (2.21)–(2.22) represent optimal co-state transitions, and (2.23)–(2.24) are complementary-slackness conditions for the state variables:

$$\dot{\lambda}_s = \alpha \lambda_s - \frac{\partial L}{\partial y_s} = \alpha \lambda_s + h_s - k_1 \tag{2.21}$$

$$\dot{\lambda}_u = \alpha \lambda_u - \frac{\partial L}{\partial y_u} = \alpha \lambda_u + h_u - k_2 \tag{2.22}$$

$$k_1 \ge 0 \ k_1 \cdot y_s^* = 0 \tag{2.23}$$

$$k_2 \ge 0 \quad k_2 \cdot y_u^* = 0. \tag{2.24}$$

Conditions (2.21)–(2.22) together with (2.23)–(2.24) imply that if serviceables inventory is positive (therefore $k_1 = 0$), the marginal increase of the value of a serviceables item equals interest on the current value which would be realized by selling the item plus the serviceables out-of-pocket holding cost rate and if recoverables inventory is positive ($k_2 = 0$), the increase of the corresponding value equals the sum of interests to be paid on the current value and the recoverables out-of-pocket holding cost rate.

Now we address times where above conditions might not hold and where adjoints may be discontinuous, i.e. where jumps in λ_s or λ_u and in the Hamiltonian occur. At junction points where a state constraint is binding or it reaches (entry time), leaves (exit time), or just touches (contact time) the boundary, e.g. $y_s^* = 0$, there can be jumps in λ_s at time points $\theta_s \in \Theta_s$ of height $\eta_s(\theta_s)$ such that

$$\lambda_s(\theta_s^{-}) = \lambda_s(\theta_s^{+}) + \eta_s(\theta_s)$$
where $\eta_s(\theta_s) \ge 0$ and $\eta_s(\theta_s) \cdot y_s^*(\theta_s) = 0.$

$$(2.25)$$

The same holds for time points $\theta_u \in \Theta_u$ with $y_u^* = 0$

$$\lambda_u(\theta_u^-) = \lambda_u(\theta_u^+) + \eta_u(\theta_u)$$
where $\eta_u(\theta_u) \ge 0$ and $\eta_u(\theta_u) \cdot y_u^*(\theta_u) = 0.$

$$(2.26)$$

Note that only downward jumps are possible for λ_s and λ_u .

Using the necessary conditions described in this section, we can deduce the structure of the optimal solution, which is done in the following subsection.

2.3.3 The Structure of an Optimal Solution

The trajectory of optimal decisions (p^*, r^*, w^*) is determined by the development of the state of the system (consisting of optimal values for state and co-state variables). Especially, the control policy can be given by simple rules which depend on whether a state restriction is binding or not. We therefore distinguish between four different cases with respect to their serviceables and recoverables inventory status. Where necessary, cases are further subdivided, if different control policies are applicable. As a result, we provide optimal decisions within each of the distinguished (sub-)cases and other required properties concerning co-states and demand/return rate developments. Since, in general, none of the cases will be optimal for the entire planning horizon, the question of under which circumstances to enter or to leave a certain case has to be answered as well, which will be discussed later. Thus, the following propositions should provide optimal decisions within an interval of the respective case and given that one does not switch into another case. In the remainder, let $\theta_{e,i}$ and $\theta_{x,i}$ denote the entry and exit time of a Case i interval.

Proposition 2.2 (Optimal decisions in Case 1 intervals).

If both, serviceables and recoverables inventory are positive $(y_s^* > 0, y_u^* > 0)$, no items are produced $(p^* = 0)$, remanufactured $(r^* = 0)$, or disposed of $(w^* = 0)$.

Under linear cost functions and positive serviceables inventory (to satisfy demand), holding costs can be saved by postponing production to a point in time where inventory is depleted. Remanufacturing cannot be optimal because recoverables holding cost would be substituted by higher holding cost for serviceables. Disposal cannot be optimal because if decisions before t were optimal, items held in the recoverables inventory could have been disposed of in the past realizing recoverables holding cost savings. Applying optimal decisions, the state variables develop as follows

$$\dot{y}_s = -d < 0 \text{ and } \dot{y}_u = u > 0.$$
 (2.27)

In a Case 1 interval, the adjoint variables increase (from (2.21) and (2.22)) with rates

$$\dot{\lambda}_s = \alpha \lambda_s + h_s \text{ and } \dot{\lambda}_u = \alpha \lambda_u + h_u$$
(2.28)

and therefore by assuming the interval to start at $\theta_{e,1}$, solving above first order differential equations (2.28) yields the values of stored serviceables and recoverables

$$\lambda_s(t) = \left(\lambda_s(\theta_{e,1}^+) + \frac{h_s}{\alpha}\right) e^{\alpha(t-\theta_e)} - \frac{h_s}{\alpha}$$
(2.29)

$$\lambda_u(t) = \left(\lambda_u(\theta_{e,1}^+) + \frac{h_u}{\alpha}\right) e^{\alpha(t-\theta_e)} - \frac{h_u}{\alpha}$$
(2.30)

where $\lambda_s(\theta_{e,1}^+)$, $\lambda_u(\theta_{e,1}^+)$ have to be determined later. Optimal decisions in Case 1 intervals imply the following limitations on the co-states which follow from (2.12)-(2.14)

$$\lambda_s < c_p, \ \lambda_s - \lambda_u < c_r, \ \text{and} \ \lambda_u > -c_w.$$
 (2.31)

Proposition 2.3 (Optimal decisions in Case 2 intervals).

If serviceables inventory is zero and recoverables inventory is positive $(y_s^* = 0, y_u^* > 0)$, the optimal policy is not to produce $(p^* = 0)$ and not to dispose $(w^* = 0)$. The remanufacturing rate equals the demand rate $(r^* = d)$.

Under zero serviceables and positive recoverables inventory, demand is satisfied from remanufacturing only. Disposal is not optimal for the same reason given for Case 1. Thus, co-states are limited by the following conditions

$$\lambda_s = \lambda_u + c_r \text{ and } - c_w < \lambda_u < c_p - c_r.$$
(2.32)

In a Case 2 interval starting at $\theta_{e,2}$, both co-states rise at the same rate

$$\dot{\lambda}_s = \dot{\lambda}_u = \alpha \lambda_u + h_u, \qquad (2.33)$$

and the adjoint variables (from (2.16) and (2.22)) are given by

$$\lambda_s(t) = \lambda_u(t) + c_r, \qquad (2.34)$$

$$\lambda_u(t) = \left(\lambda_u(\theta_{e,2}^+) + \frac{h_u}{\alpha}\right) e^{\alpha(t-\theta_{e,2})} - \frac{h_u}{\alpha}.$$
 (2.35)

The recoverables inventory develops according to

$$\dot{y}_u = u - d, \tag{2.36}$$

where the direction of this change depends on the relation of demand and return rate.

Proposition 2.4 (Optimal decisions in Case 3 intervals).

If serviceables inventory is positive and recoverables inventory is zero $(y_s^* > 0, y_u^* = 0)$, the optimal policy is not to produce $(p^* = 0)$ and not to remanufacture $(r^* = 0)$. All returns are disposed of $(w^* = u)$.

Production cannot be optimal under positive serviceables inventory since holding cost can be saved by postponing production until inventory becomes zero. For the same reason, remanufacturing cannot occur as long as no costs can be saved by shipping recoverables to serviceables inventory. Collecting returns instead of disposing of them implies to terminate Case 3 and to move to Case 1.

In a Case 3 interval starting at $\theta_{e,3}$, the value of a serviceables item is restricted by

$$\lambda_s < c_r - c_w, \tag{2.37}$$

and it increases with rate

$$\dot{\lambda}_s = \alpha \lambda_s + h_s, \tag{2.38}$$

19

while the value of an item put into the recoverables inventory stays constant at $-c_w$ since an additional return must be disposed of. The adjoint variables (from (2.17) and (2.21)) are

$$\lambda_s(t) = \left(\lambda_s(\theta_e^+) + \frac{h_s}{\alpha}\right) e^{\alpha(t-\theta_e^+)} - \frac{h_s}{\alpha}, \quad \lambda_u(t) = -c_w.$$
(2.39)

As in Case 1, the serviceables stock must decrease, since production and remanufacturing do not take place, i.e.

$$\dot{y}_s = -d < 0.$$
 (2.40)

Proposition 2.5 (Optimal decisions in Case 4 intervals).

If serviceables and recoverables inventories are zero $(y_s^* = 0, y_u^* = 0)$, optimal decisions depend on how demand relates to the return rate and two subcases can be distinguished:

Subcase $4(1) \Leftrightarrow d \leq u$

Demand is satisfied completely by remanufacturing returns $(r^* = d)$ and excess returns are disposed of $(w^* = u - d)$. No items are produced $(p^* = 0)$. **Subcase 4(2)** $\Leftrightarrow u < d$

All returns are remanufactured $(r^* = u)$ and the missing items are produced $(p^* = d - u)$. No items are disposed of $(w^* = 0)$.

The case of zero inventories is characterized by demand and return synchronized activities. In absence of inventory holding cost, simultaneous production and disposal is suboptimal because of the general recovery advantage assumption. Situations with excess demand require production, while situations with excess returns require disposal.

With zero inventories, the co-states can be interpreted as financial implications of an additional return (λ_u) or additional demand (λ_s) . If returns exceed the demand rate $(d(t) \leq u(t))$, the adjoint variables (from (2.16) and (2.17)) are

$$\lambda_s(t) = c_r - c_w, \quad \lambda_u(t) = -c_w, \tag{2.41}$$

i.e. an additional demand unit would be satisfied from remanufacturing a returned unit which would have otherwise been disposed of, whereas another returned item has to be disposed of because it can neither be stored nor remanufactured.

For u(t) < d(t) we find (from (2.15) and (2.16))

$$\lambda_s(t) = c_p, \quad \lambda_u(t) = c_p - c_r. \tag{2.42}$$

Since there are not enough returns available to satisfy demand completely, demand for another item is satisfied by additional production, while an additional return could be used to replace production by remanufacturing.

The main results of the four cases are summarized in Table 2.1.

	p^*	r^*	w^*	\dot{y}_s	\dot{y}_u	λ_s	$\dot{\lambda}_s$	λ_u	$\dot{\lambda}_u$		
Case 1: $y_s > 0, y_u > 0$											
	0	0	0	-d	u	$< c_p$	$\alpha\lambda_s + h_s$	$-c_w <$	$\alpha \lambda_u + h_u$		
Case 2: $y_s = 0, y_u > 0$											
	0	d	0	0	u-d	$\lambda_u + c_r$	$\alpha\lambda_u + h_u$	$-c_w < < < c_p - c_r$	$\alpha \lambda_u + h_u$		
Case 3: $y_s > 0, y_u = 0$											
	0	0	u	-d	0	$< c_r - c_w$	$\alpha\lambda_s + h_s$	$-c_w$	0		
Case 4: $y_s = 0, y_u = 0$											
(1) $(d \le u)$ (2) $(u < d)$						-		$-c_w$ $c_p - c_r$			
Case 4 generalized: $p^* = \max\{d - r^*, 0\}, r^* = \min\{u, d\}, w^* = \max\{u - r^*, 0\}$											

Table 2.1. Main results of optimal cases in the basic model.

2.3.4 Optimal Transitions Between Cases and Subcases

After determining the optimal production, remanufacturing, and disposal policy for each of the four inventory situations it is necessary to derive conditions for the possible order of different intervals. First, this order is derived from inventory non-negativity constraints which may be binding. Second, a transition from one case to another can be necessary in order to maximize the Hamiltonian, i.e. the change is associated with a higher value of the Hamiltonian. At this point, it is important to distinguish between two kinds of possible transitions, namely *forced* and *automatic*. An automatic transition occurs when applying the optimal control rule for the current case leads to a transition to another case. For instance, the serviceables inventory running empty would necessitate an automatic transition from Case 1 to Case 2. In contrast to this a forced transition is result of a decision. This happens when inventory is build up in anticipation of an impeding use.

There is a relation between the kind of transition and the co-state variables λ_s and λ_u because some transitions can only occur if the co-state variables are not continuous. These transitions are further called *discontinuous* as opposed to *continuous* case changes. For this reason, it is necessary to investigate the continuity of the adjoints.

Proposition 2.6. λ_s and λ_u are continuous, i.e. jump parameters η_s and η_u vanish everywhere, except at time points $\theta \in \Theta$ where $y_s(\theta) = y_u(\theta) = 0$ and $u(\theta) = d(\theta)$ holds.

Proposition 2.6 implies that any transition between cases with the exception of transitions from and to (and within) Case 4 intervals must be continuous w.r.t. the co-states. If it is not, from (2.25) and (2.26) we know that the co-states must fall at the jump point. This immediately excludes transitions from Case 4(1) to Case 4(2).

Corollary 2.1. Within Case 4 Subcase 4(2) is followed by Subcase 4(1). This automatic and discontinuous transition requires demand to equal the return rate. A transition in the opposite direction is not possible.

From the optimal decisions in Case 1 and Case 3 intervals it follows, that it is not possible to produce or to remanufacture an amount that exceeds current demand. Thus, at any time t where $y_s(t) > 0$ holds, the inventory level must decrease, i.e. $\dot{y}_s(t) = -d(t) \leq 0$.

Corollary 2.2. It is never optimal to build up a serviceables inventory.

Corollary 2.2 excludes any transition from Cases 2 or 4 to 1 or 3. Further, the optimal policy in Case 1 requires that this case must terminate in a Case 2 interval. Transitions from Case 1 to Cases 3 or 4 are therefore not possible.

Corollary 2.3. A Case 1 interval automatically and continuously terminates in another of Case 2.

A direct transition from a Case 3 $(y_s > 0, y_u = 0)$ to a Case 2 $(y_s = 0, y_u > 0)$ interval is not possible because there must be a time of Case 4 $(y_s = 0, y_u = 0)$ between Case 3 and 2. This leaves two possible transitions at the end of a Case 3 interval, as stated in the following corollary.

Corollary 2.4. There exist two types of transitions starting at a Case 3 interval. A continuous and automatic transition to Case 4(1) occurs when the serviceables inventory is depleted, and a forced and continuous transition to Case 1 when a decision is made to stop disposing of the returns.

It remains to be seen which transitions are possible between Case 2 and Case 4 intervals. Here, we have 4 possibilities as stated in the following corollary.

Corollary 2.5. In order to build up recoverables inventory, a forced transition from Case 4 to Case 2 is necessary. This can happen either continuously (starting in Subcase 4(1)) or discontinuously (starting in Subcase 4(2)).

When the recoverables inventory is depleted, an automatic transition from Case 2 to Case 4 takes place. This can be either continuous (terminating in Subcase 4(2)) or discontinuous (terminating in Subcase 4(1)).

Figure 2.3 summarizes all possible transitions.

From Corollary 2.2 it follows that only in the model with initial inventories Cases 1 and 3 appear in an optimal time path. Therefore, Cases 2 and 4 and transitions between them (highlighted through a grey shaded area in Figure 2.3) build the focal point of the model and are examined further in the next section.



Fig. 2.3. Optimal case transitions in the basic model

2.3.5 Properties of Optimal Return Collection Intervals

After showing which transitions are possible in an optimal solution the question arises at what time these transitions take place. In case of automatic transitions the changes are triggered by following the optimal control policy within a case. But in case of a forced transition the decision maker has to determine the optimal switching time. Therefore, the problem has to be solved to determine the optimal entry point $\theta_{e,2}$ and the optimal exit point $\theta_{x,2}$ of a Case 2 interval. Unfortunately, it is not possible to give an analytical expression for these points, but we can describe the optimal Case 2 intervals and we can provide some properties, especially their optimal entry and exit time. Based on these results an algorithm can be constructed in order to compute these values numerically.

The first property results from the fact that only downward jumps of λ_s and λ_u are permitted inside Case 4 intervals. This leads to the following corollary.

Corollary 2.6 (Location Property).

Let θ denote a time point where $d(\theta) = u(\theta)$. If returns cross demand from above, i.e. there exist $\varepsilon_1 > 0$ and $\varepsilon_2 > 0$ with $d(\theta - \varepsilon_1) < u(\theta - \varepsilon_1)$ and $d(\theta + \varepsilon_2) > u(\theta + \varepsilon_2)$ as well as $\varepsilon_1 + \varepsilon_2$ being sufficiently small, then it is always optimal to have a positive recoverables inventory at (and around) time θ ($y_u(\theta) > 0$).

Corollary 2.6 follows immediately from the cost advantage of holding a returned item for an infinitesimal (sufficiently small) time and identifies the time where returns (that otherwise would be disposed of) can be saved before θ and used with higher benefits after θ by remanufacturing it instead of producing a new item. Considering that it might only be beneficial to store returns for a limited time (this aspect is discussed below in more detail), there



Fig. 2.4. Example of a Case 2 interval which has maximal length.

might be a special case, that returns equal demands for some period. In this case, we have to ensure that this period does not exceed the maximum time for which it is beneficial to store returns.

In the following the maximal length of a Case 2 interval is investigated. Inside this case, a lower bound for the value of a returned item λ_u is given by $-c_w$ and an upper bound is $c_p - c_r$. Due to the increasing co-state λ_u and its continuity, inserting these minimal and maximal values into the costate development equation within a Case 2 interval (2.35) yields the following condition which is illustrated in Figure 2.4.

Proposition 2.7 (Maximal Length Property).

The maximal length τ_u of a Case 2 interval $I = (\theta_{e,2}, \theta_{x,2})$ is given by

$$(\theta_{x,2} - \theta_{e,2}) \le \tau_u := \frac{1}{\alpha} \ln \left(\frac{\alpha(c_p - c_r) + h_u}{-\alpha c_w + h_u} \right).$$

The Maximal Length Property represents a marginal criterion for collecting one returned item which can be interpreted as follows. An additionally collected return at $\theta_{e,2}$ is needed for reuse at time $\theta_{x,2}$. Therefore, it will be stocked for a time interval of length $\theta_{x,2} - \theta_{e,2}$. This can only be optimal if the corresponding holding costs do not exceed the recovery cost advantage and also accounting for discounting and different time values of the respective payments. τ_u will further be referred to as *Maximal Holding Time* (of returns).

It can easily be shown, that

$$\lim_{\alpha \to 0} \tau_u = \frac{c_p + c_w - c_r}{h_u} \tag{2.43}$$

holds which constitutes the maximal length property as shown by Minner and Kleber (2001) for the undiscounted case. In the absence of time value considerations, an item will be stored until there is equality of holding costs $h_u \cdot \tau_u$ and the direct recovery cost advantage $c_p + c_w - c_r$.

The question arises if all optimal Case 2 intervals have maximal length or when such an interval does not have maximal length. Before we answer this question we provide another property of the optimal Case 2 intervals. Obviously, at the beginning and the end of a Case 2 interval the recoverables inventory is empty. Therefore, cumulative returns and cumulative remanufacturing rate over the entire time interval have to be equal. Further, during the time interval, cumulative returns have to exceed cumulative remanufacturing rates in order to remain in Case 2. This is intuitively clear and leads to the following proposition (presented without proof).

Proposition 2.8 (Inventory Conditions).

Let $I = (\theta_{e,2}, \theta_{x,2})$ be an open time interval where $y_u > 0$ and $y_u(\theta_{e,2}) = y_u(\theta_{x,2}) = 0$. Then,

(i) cumulative demand equals cumulative returns over the whole interval

$$\int_{\theta_{e,2}}^{\theta_{x,2}} (d(t) - u(t))dt = 0, \qquad (2.44)$$

(ii) at any point $\theta \in I$, cumulative returns must be larger than cumulative demand

$$\int_{\theta_{e,2}}^{\theta} (u(t) - d(t))dt > 0.$$
(2.45)

As an implication of Proposition 2.8 it must hold that at $\theta_{e,2}$, the return rate must not be smaller than the demand rate in order to start collecting returns and at $\theta_{x,2}$, the opposite must hold.

In order for a feasible solution which holds under above inventory conditions, lengths of optimal Case 2 intervals may be smaller than the maximal length given in Proposition 2.7. There are two reasons for this. First, it is possible that there are not enough returns available to save further units for later remanufacturing. In such cases the interval starts at a point in time where returns equal demand rate or at the begin of the planning period (at time zero). Otherwise, it would have been possible to start collecting returns earlier and thereby, increasing the Case 2 interval length. A second reason for non-maximal interval lengths is no excess demands. Then, the Case 2 interval must end at a point where returns equal demands or at the planning horizon T. If this is not the case, additional demand could have been satisfied by extending the Case 2 interval.

Proposition 2.9. If the interval I does not reach its maximum length, i.e. if $(\theta_{x,2} - \theta_{e,2}) < \tau_u$ holds, and if $\theta_{e,2} > 0$ and $\theta_{x,2} < T$, then (i) $u(\theta_{e,2}) = d(\theta_{e,2})$ or (ii) $u(\theta_{x,2}) = d(\theta_{x,2})$ must hold.
Up to now we have shown how given optimal entry and exit points of Case 2 intervals can be characterized. How these results can be used in order to determine the optimal entry and exit points of optimal Case 2 intervals (without having this information) is shown in the following section where an algorithm for this problem is provided.

2.3.6 Solution Algorithm

The Cases 2 and 4 build the focal point of the model with zero initial inventories, because from Proposition 2.2 it follows that Cases 1 and 3 will never appear in an optimal solution. From the initial and terminal conditions we know that at both points, t = 0 and t = T, Case 4 must apply. Note that if there exists no time where Corollary 2.6 holds, it is never optimal to have a positive recoverables inventory. For constructing an optimal time path we suggest a forward algorithm that proceeds as follows.

Algorithm 2.1

Step 1 Identify all K time points for which Corollary 2.6 holds. let θ^k denote the intersection points k = 1, ..., K. Step 2 let k := 1. while $k \leq K$ do Apply Propositions 2.7, 2.8, and 2.9 to determine the corresponding interval $I^k = (\theta_{e,2}^k, \theta_{x,2}^k)$: let $\theta_{e,2}^k = \theta_{x,2}^k = \theta^k$. Decrease $\theta_{e,2}^k$ and increase $\theta_{x,2}^k$ so that Proposition 2.8 always holds until one of the terminating conditions given in Propositions 2.7 or 2.9 (i) or (ii) are achieved, $\theta_{e,2}^k = 0 \text{ or } \theta_{x,2}^k = T.$ let k := k + 1. end while Step 3 while there exist two consecutive intervals with $\theta_{x,2}^k = \theta_{e,2}^j$, j > k and $\theta_{x,2}^j - \theta_{e,2}^k < \tau_u$ do let $\theta_{x,2}^k := \theta_{x,2}^j$. Delete point *j* from the set of construction intervals. Decrease $\theta_{e,2}^k$ and increase $\theta_{x,2}^k$ so that Proposition 2.8 always holds until one of the terminating conditions given in Propositions 2.7 or 2.9 (i) or (ii) are achieved or $\theta_{e,2}^k = 0$ or $\theta_{x,2}^k = T$. end while



Fig. 2.5. Example for applying the forward algorithm

The following example with a constant return rate and a seasonal demand pattern as depicted in Figure 2.5 is used to illustrate Algorithm 2.1. First, Step 1 identifies two intersection points k = 1, 2 where returns intersect demands from above. According to Step 2, the collection intervals $(\theta_{e,2}^1, \theta_{x,2}^1)$ for k =1 and $(\theta_{e,2}^2, \theta_{x,2}^2)$ for k = 2 are extended until one of the stopping criteria holds. In this case, the interval construction ends with criterion (i) and (ii) from Proposition 2.9 because at the time where demands intersect returns from above, the first interval cannot be extended to the right and the second interval cannot be extended to the left without violating the integral condition. Note that (in general) if we assume a rather small value for τ_u , that both interval constructions may not reach this point and the interval construction would terminate because of maximal length reached. Because this example also covers the special situation dealt with by Step 3, we find two intervals that directly follow each other. Let us assume that the cumulative length of both intervals is still smaller than τ_u . Then, a joint interval is constructed by extending the interval to $\tilde{\theta}_{e,2}^1$ on the left and to $\tilde{\theta}_{x,2}^1$ on the right. Here, it is assumed that the construction ends because maximum length of a collection interval is reached, i.e. $\tilde{\theta}_{x,2}^1 - \tilde{\theta}_{e,2}^1 = \tau_u$.

2.4 Dealing with Initial Inventories

The main difference that arises from allowing for initial stocks is that the jump variable w_0 does not necessarily vanish in problem (2.9). w_0 stands for a disposal quantity (impulse) of unwanted returns which immediately decreases recoverables stock by a quantity (jump). This scenario requires the generalized maximum principle that allows for jumps in state variables, i.e. in our case y_u . Since we know that a jump in y_u is possible only at time zero, the following necessary conditions have to be added (see Feichtinger and Hartl (1986)).

Define the impulse Hamiltonian function H_0 (being valid at time zero)

$$H_0(y_s(0), y_u(0^-), w_0, \lambda_u(0)) = -c_w \cdot w_0 - \lambda_u(0) \cdot w_0.$$
(2.46)

Due to its linearity, (2.46) is maximized if (omitting time indices)

$$w_0^* = \begin{cases} 0 \quad \lambda_u > -c_w \\ singular \quad \lambda_u \le -c_w. \end{cases}$$
(2.47)

Since the states do not appear in the impulse Hamiltonian the adjoints do not jump at time zero, i.e. $\lambda_u(0^-) = \lambda_u(0^+)$. It is clear that w_0 vanishes if the value of stored returns exceeds (negative) payments ('revenue') for disposal $(\lambda_u > -c_w)$. This implies that the disposal rate equals zero. Otherwise, the height of the jump w_0 must be determined by searching for a desired inventory level \tilde{y}_u^0 followed by setting the optimal disposal quantity equal to w_0^* = $\max\{0, y_u^0 - \tilde{y}_u^0\}.$

Proposition 2.10. The desired initial recoverables inventory level \tilde{y}_u^0 is given by the net demand during a time interval $[0, \theta^u]$

$$\tilde{y}_{u}^{0} = \max\left\{0, \int_{0}^{\theta^{u}} (d(t) - u(t))dt - y_{s}^{0}\right\}.$$

 $\theta^u \leq \tau_u$ is the largest value for which holds at any point $\theta < \theta^u$ cumulative demand exceeds cumulative returns

$$\int_{\theta}^{\theta^u} (d(t) - u(t))dt \ge 0.$$
(2.48)

Except for the trivial case $y_s \ge \int_0^T d(t) dt$, a starting interval with $y_s > 0$ will always (at time $\theta^s \in (0,T)$) terminate in an interval with $y_s = 0$ after using all serviceables inventory to satisfy demand, i.e.

$$\theta^s : \int_0^{\theta^s} d(t)dt = y_s^0.$$
(2.49)

Let us consider the four cases outlined in Section 2.3.3 as initial conditions (after disposing of quantity w_0^*). Case 4 initial conditions $y_s^0 = \tilde{y}_u^0 = 0$ lead to the results derived before. If the system starts in Case 1 $(y_s^0 > 0, \tilde{y}_u^0 > 0)$ it will turn to Case 2 $(y_s = 0, y_u > 0)$ at some time θ^s and finally (at time $\theta^u > \theta^s$) terminate automatically into a Case 4 interval. For a starting Case 2 interval, the same holds with $\theta^s = 0$.

The most interesting case is a starting interval of Case 3 where $y_s^0 > 0$, $\tilde{y}_u^0 = 0$ holds. While the above starting inventory (except Case 4) configuration always terminates with both state variable constraints binding, the end of a Case 3 starting interval can also be driven by a cost advantage of collecting returns. The alternatives are (1) to continue with the Case 3 policy of satisfying demand from serviceables inventory and to dispose of all returns and (2) switch to Case 1 and start collecting excess returns for future recovery within a subsequent Case 2 interval. The conditions for switching from Case 3 to Case 1 can be determined by applying the following modification of Proposition 2.8 (presented without proof).

Proposition 2.11. Let $I = (\theta_{e,1}, \theta_{x,2})$ be an open time interval (of Case 1 immediately followed by Case 2) where $y_u > 0$ and $y_u(\theta_{e,1}) = y_u(\theta_{x,2}) = 0$. Then,

(i) cumulative demand equals the sum of cumulative returns over the whole interval plus serviceables inventory at $\theta_{e,1}$

$$\int_{\theta_{e,1}}^{\theta_{x,2}} (d(t) - u(t))dt - y_s(\theta_{e,1}) = 0,$$

(ii) at any point $\theta \in I$ the sum of cumulative returns and serviceables inventory at $\theta_{e,1}$ must be larger than cumulative demand

$$y_s(\theta_{e,1}) + \int_{\theta_{e,1}}^{\theta} (u(t) - d(t))dt > 0.$$

After collecting recoverables in a Case 1 time interval, these stocks are used to remanufacture for demand in a Case 2 interval with positive initial recoverables inventory.

The solution algorithm changes in the following way. First, one determines the desired initial recoverables inventory \tilde{y}_u^0 and thus obtains the initial disposal quantity. Considering that until some time point θ^s the demand can be satisfied by using the serviceables inventory, we define the adjusted demand rate

$$\tilde{d}(t) = \begin{cases} 0 & \text{if } t \le \theta^s \\ d(t) & \text{otherwise.} \end{cases}$$
(2.50)

Afterwards, one continues with the proposed solution algorithm using $\tilde{y}_s^0 = 0$, \tilde{y}_u^0 , i.e. the transformed initial conditions and \tilde{d} are used. Until time θ^u , we always have a time interval with $y_u > 0$. Note, that due to the possible jump in the demand rate at time θ^s , the necessary conditions do not hold at this point, which can lead to a jump in the adjoint variables. At time θ^s it is therefore necessary to check whether \tilde{d} jumps above the return rate which implies that it is optimal to have a recoverables inventory at this time.

2.5 Numerical Examples

In this section the results of the basic model are illustrated by using three examples (Examples 2.1-2.3) based on the demand/return scenario introduced in Section 2.2. According to the Location Property, intervals with positive recoverables inventory will be located around time points $\theta^k \in \{5.92, 12.21\}$. All three examples differ with respect to initial stock levels in both serviceables and recoverables inventories as shown in Table 2.2. Cash flow parameters are set as follows: $c_p = 2$, $c_r = 1$, $c_w = 1$, $h_u = 1$, and $\alpha = 0.1$. Therefore, a Case 2 interval length will not exceed 2.00 units time.

Table 2.2. Initial inventory levels in Examples 2.1-2.3

Example	y_s^0	y_u^0
2.1	0	0
2.2	1	1
2.3	2	1

For now, we consider the case of zero initial inventories (Example 2.1). Using the solution algorithm shown in Section 2.3.6 the following two collection intervals are determined: $[\theta_{e,2}^1, \theta_{x,2}^1] = [4.85, 6.85]$ and $[\theta_{e,2}^2, \theta_{x,2}^2] = [11.82, T]$. The first interval has maximum length, but the second one is shorter because it reaches the end of the planning horizon where no further demand is available. Figure 2.6 shows optimal Case 2 intervals and the optimal development of the value of returns λ_u . The resulting inventory levels are depicted in Figure 2.7.

Outside optimal Case 2 intervals demand is satisfied either from remanufacturing and production if demand exceeds the return rate, or from remanufacturing alone if the opposite applies. In the first case, an additional return would have a value of $c_p - c_r = 1$ since it could be used immediately to forego production. In the second case, the additional return ought to be disposed of, thus having a value of $-c_w = -1$. Inside Case 2 intervals demand is met by remanufacturing currently or previously collected returns, which both are valued equally. The value of an additional (marginal) return increases with time, since the time it spends in inventory until usage decreases and thus, corresponding out-of-pocket holding costs and opportunity costs would reduce.



Fig. 2.6. Optimal Case 2 intervals and co-state λ_u development in Example 2.1.



Fig. 2.7. Optimal inventory levels in Example 2.1.

Let us now presume two scenarios with positive initial inventory levels for serviceables and recoverables. Under such situations there are two additional questions to be answered. First, it has to be determined whether to keep all recoverables available at time zero or to dispose of a certain quantity w_0 and secondly, we want to know how operative decisions change.

In Example 2.2 where $y_s^0 = 1$, $y_u^0 = 1$, the serviceables inventory is depleted first which completes at time $\theta_s = 0.83$. Thus, the desired recoverables inventory level at time zero (which can be used up within economically reasonable time) amounts to $\tilde{y}_u^0 = 0.81$. A quantity of $w_0 = 0.19$ is therefore disposed of at time zero. The remaining returns are kept and further accumulated using incoming returns until after θ_s , upon which recoverables inventory is reduced by filling all demand from remanufacturing. In other words, at θ_s a transition from a Case 1 to a Case 2 interval occurs. The latter finishes at time $\theta_{x,2}^0 = \tau_u = 2$ as the inventory runs empty. Then, the same policy is optimal as described in Example 2.1. Optimal Case 1 and 2 intervals as well as the changed evolution of the value of returns (λ_u) can be seen from Figure 2.8. The corresponding serviceables and recoverables inventory levels are depicted in Figure 2.9.

Since in Example 2.3 initial serviceables stock is higher than in the previous example, it is not used up before $\theta_s = 1.52$. It is possible to acquire sufficient returns during this initial period making it superfluous to have any returns available at time zero ($\tilde{y}_u^0 = 0$). All initial returns are disposed of ($w_0 = 1$). The system therefore starts in a Case 3 situation. Under these circumstances one wants to know whether to have a positive recoverables inventory at θ_s and at what point to start collecting returns, i.e. when to switch from Case 3 to Case 1. Since after θ_s demand surpasses currently available returns, stockkeeping of returns is profitable and it starts at time $\theta_{e,2}^0 = 0.11$. At θ_s a transition from a Case 1 to a Case 2 interval takes place, and since stock-keeping of returns here yields maximal length at $\theta_{x,2}^0 = 2.11$ the recoverables inventory is depleted. As in the previous example, subsequent optimal decisions follow the same pattern as in Example 2.1, being depicted on top of optimal costate developments in Figure 2.10. The changed stock levels are to be found in Figure 2.11.



Fig. 2.8. Optimal Case 1/2 intervals and co-state λ_u development in Example 2.2.



Fig. 2.9. Optimal inventory levels in Example 2.2.

33



Fig. 2.10. Optimal policies and co-state λ_u development in Example 2.3.



Fig. 2.11. Optimal inventory levels in Example 2.3.

2.6 Comparison of Holding Cost and Discounted Cash Flow

In inventory control Holding Cost (HC, also called Average Cost, AC, for cyclic infinite horizon problems) models often are used as approximation to the as correct recognized Discounted Cash Flow (DCF) approach because they are easier to solve. It has been shown for a number of models (see, e.g., Grubbström (1980), Corbey et al. (1999)) that the HC framework leads to almost the same optimal decision as the DCF approach when choosing the HC holding cost rate appropriately. While in DCF models actual cash flows are considered and holding an item in inventory causes out-of-pocket-costs only, the cost oriented HC approach also includes the opportunity costs of capital tied up in stock. Thus, valuation of inventories becomes crucial in order to find almost correct inventory control decisions.

A lot of effort has been spent in order to determine the 'right' holding cost rate and to correctly value stored items (see, e.g., Teunter et al. (2000), Teunter and van der Laan (2004)). Although there exists general criticism of such undertaking (see, e.g., Fleischmann (2001a)), the deterministic framework presented above can also be used to find a holding cost rate that leads to the same results when comparing above DCF and the corresponding total undiscounted costs solutions provided by Minner and Kleber (2001). Since there is no motivation for having a positive stock in the serviceables inventory in this model, the results are restricted to the holding cost rate for recoverables. In contrast to the usual approach where the HC objective constitutes a first order approximation of the respective DCF objective (Fleischmann (2001a)), in the sequel we deal with properties of the respective optimal solutions.

2.6.1 Holding Cost Results

A main result from solving both discounted and undiscounted models is that the value of stored returns as expressed by the co-state λ_u is not constant over time, and therefore no unequivocal value and with it no 'right' holding cost rate exists. However, given identical parameters for all other operational cash flows and costs a recoverables holding cost rate h_u^{HC} can be determined leading to the same results as are obtained in the DCF approach with outof-pocket holding cost rate h_u^{DCF} by valuing the returns with some kind of average value.

When solving the HC or the corresponding DCF problem a difference occurs in setting the holding time, i.e. the maximum length an interval with positive recoverables inventory can possibly have. Since α vanishes in (2.22), the co-state λ_u rises linearly with rate $\dot{\lambda}_u = h_u$ in a Case 2 interval where $y_u > 0$. Thus, the Maximal Holding Time in the HC case is given by (see Minner and Kleber (2001), compare with (2.43)) 2.6 Comparison of Holding Cost and Discounted Cash Flow

$$\tau_u^{HC} = \frac{c_p + c_w - c_r}{h_u^{HC}},$$
(2.51)

and in the DCF case it is (see Proposition 2.7)

$$\tau_u^{DCF} = \frac{1}{\alpha} \ln \left(\frac{\alpha(c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}} \right).$$
(2.52)

If $\tau_u^{DCF} = \tau_u^{HC}$, both approaches lead to the same solution.

The difference between both holding cost rates can be modeled as a markup $m_h = h_u^{HC} - h_u^{DCF}$. In the following the precise mark-up for the recoverables holding cost rate as well as two approximations for small values of α will be determined. Several numerical examples are used in order to assess the quality of the approximations.

2.6.2 A Comparison of Different Approximations

Now, let us determine the value of the mark-up m_h that assures

$$\tau_u^{HC} = \tau_u^{DCF} \tag{2.53}$$

by assuming identity of all parameters except for the holding cost rates. Further, it is required that out-of-pocket holding costs exceed costs from postponing disposal $(h_u^{DCF} > \alpha c_w)$ in order to assure a finite value of τ_u^{DCF} . For instance this is always true in situations where disposal earns a salvage revenue $(c_w < 0)$.

Exact solution. For finite values of τ_u^{DCF} , solving equation (2.53) for m_h gives the exact value of the mark-up

$$m_h = \frac{\alpha(c_p + c_w - c_r)}{\ln\left(\frac{\alpha(c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}}\right)} - h_u^{DCF}, \qquad (2.54)$$

for which two approximations are elaborated.

Approximation 1. Replacing the logarithm term in (2.54) by its first order Taylor series approximation $(\ln x \approx x - 1)$ which is valid for small values of α yields

$$m_h \approx \frac{\alpha(c_p + c_w - c_r)}{\frac{\alpha(c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}} - 1} - h_u^{DCF}$$
$$= \frac{\alpha(c_p + c_w - c_r)}{\frac{\alpha(c_p - c_r) + h_u^{DCF} + \alpha c_w - h_u^{DCF}}{-\alpha c_w + h_u^{DCF}}} - h_u^{DCF} = -\alpha c_w.$$

A nearly optimal HC recoverables holding cost rate is given by out of pocket costs, reduced by the interests on the disposal cost rate, i.e. $h_u^{DCF} - \alpha c_w$.

35

This value can be interpreted as follows. Holding costs only play a role during intervals where the recoverables inventory is positive. If an additional return arrives during such a collection interval, there are two options, it can be either be stocked (which will be done in order to shorten the collection interval) or it can be disposed of at cost c_w . Therefore, the value of this item is given by the opportunity cost saved by storing it, i.e. the negative disposal cost rate. Holding an item becomes cheaper by (saved) interests on disposal $-\alpha c_w$ or more expensive by (lost) interests on the salvage value.

Approximation 2. A first order MacLaurin series approach applied to the whole term (2.54) yields the following approximation being valid for small values of α

$$m_h = \alpha \left(\frac{c_p - c_r}{2} + \frac{-c_w}{2}\right) + O(\alpha^2).$$
 (2.55)

For a complete derivation see page 44. In order to obtain results similar to the DCF solution, the value of a returned item should be set to the unweighted average of the difference of direct production and remanufacturing costs $c_p - c_r$ and the opportunity cost saved by not disposing the item $-c_w$. This makes sense if one considers that m_h has been derived by using a marginal criterion regarding the first item to be put on stock. The value of this (marginal) returned item (represented by the co-state variable) continuously increases during the interval where it is kept on inventory and for small values of α (no discounting) this movement is approximately linear. Since the minimum value of the stored return is given by $-c_w$ (otherwise it would be disposed of) and as a maximum $c_p - c_r$ may be reached (otherwise production would be preferred), above average value results.

Comparison. It remains to be seen whether these approximations perform well or not. In order to answer this question, optimal and approximated holding times are plotted against the discount rate using a number of different scenarios. These were derived by using a factorial design that helps to find out how the parameters interact with each other. Since unit production and remanufacturing costs only appear as a difference in all relevant terms, these parameters have been fixed in such a way that the difference $(c_p - c_r) \in \{-1, 1, 3, 5\}$. The disposal parameter uses the following values in order to account for high, zero, and negative (i.e. positive salvage revenue) per unit cost rates $c_w \in \{-2, 0, 2\}$. Finally, we considered two levels (high and low) for the out-of-pocket holding cost rate $h_u^{DCF} \in \{0.1, 1\}$. By excluding invalid parameter soft the examined scenarios as well as the maximal allowed discount rate α_{max} for a finite Maximal Holding Time and average values of the recoverables on stock according to Approximations 1 and 2 can be found in Table 2.3.

In order to gain insight into the quality of the approximations, a maximum permissible discount rate for a given approximation error in the holding time is calculated for maximum relative errors of 1%, 10%, 20% and 50%, see Tables 2.4 and 2.5.

Scenario	$c_p - c_r$	c_w	h_u^{DCF}	α_{max}	$-c_w$	$\frac{c_p - c_r}{2} - \frac{c_w}{2}$
1	-1	2	0.1	.05	-2	-1.5
2	-1	2	1	.5	-2	-1.5
3	1	0	0.1	∞	0	.5
4	1	0	1	∞	0	.5
5	1	2	0.1	.05	-2	5
6	1	2	1	.5	-2	5
7	3	-2	0.1	∞	2	2.5
8	3	-2	1	∞	2	2.5
9	3	0	0.1	∞	0	1.5
10	3	0	1	∞	0	1.5
11	3	2	0.1	.05	-2	.5
12	3	2	1	.5	-2	.5
13	5	-2	0.1	∞	2	3.5
14	5	-2	1	∞	2	3.5
15	5	0	0.1	∞	0	2.5
16	5	0	1	∞	0	2.5
17	5	2	0.1	.05	-2	1.5
18	5	2	1	.5	-2	1.5

Table 2.3. Cost parameters in examined scenarios.

From comparing values in Tables 2.4 and 2.5 it follows that Approximation 2 performs considerably better than Approximation 1 in all scenarios, which is understandable as it uses a more sophisticated method. Further it can be stated, that the performance of the preferable Approximation 2 depends on the parameters in such a way that it tends to perform better under circumstances that lead to a small (correct) Maximal Holding Time, i.e. for smaller $c_p - c_r$, higher salvage revenues (negative disposal costs) $-c_w$, or higher out-of-pocket holding cost rates h_u^{DCF} (see e.g. Scenario 8). This result is intuitive, since a smaller holding time also implies a smaller impact of compound interest. On the other hand, the approximation should not be used if a high Maximal Holding Time is expected (Scenario 17).

The first approximation tends to perform better as its valuation of returns gets closer to that of the second, i.e. where $-c_w \approx c_p - c_r$ holds (Scenarios 1 to 4, 7 and 8). Although it seems to be (without knowing the technical details) more intuitive than the second, it should generally not be used.

Our results also show that one has to be very careful when designing experiments for determining the 'right' holding cost rate in more complex models, which can not be solved analytically, e.g. when stochastic influences or setup costs are considered. For instance, Teunter et al. (2000) compared five different reasonable methods for setting the holding cost rates in a model with stochastic demand and return rates as well as setup costs for manufacturing and remanufacturing by using simulation methods, but these did not include valuation methods like that one which we derived by using Approximation 2.

Scenario	1%	10%	20%	50%
1	0.0019	0.0146	0.0230	0.0348
2	0.0193	0.1461	0.2297	0.3479
3	0.0020	0.0206	0.0425	0.1144
4	0.0201	0.2065	0.4250	1.1440
5	0.0007	0.0060	0.0110	0.0216
6	0.0066	0.0605	0.1104	0.2163
7	0.0021	0.0352	0.2835	> 100
8	0.0209	0.3517	2.8351	> 100
9	0.0007	0.0069	0.0142	0.0381
10	0.0067	0.0688	0.1417	0.3813
11	0.0004	0.0038	0.0073	0.0157
12	0.0040	0.0381	0.0727	0.1570
13	0.0007	0.0080	0.0198	0.1607
14	0.0068	0.0798	0.1977	1.6069
15	0.0004	0.0041	0.0085	0.0229
16	0.0040	0.0413	0.0850	0.2288
17	0.0003	0.0028	0.0054	0.0123
18	0.0029	0.0279	0.0541	0.1232

Table 2.4. Maximum discount rates α_{max} for different error levels when using Approximation 1.

Table 2.5. Maximum discount rates α_{max} for different error levels when using Approximation 2.

Scenario	1%	10%	20%	50%
1	0.0227	0.0408	0.0454	0.0495
2	0.2274	0.4079	0.4538	0.4945
3	0.0417	0.2214	0.4906	4.5063
4	0.4170	2.2143	4.9064	45.0633
5	0.0109	0.0298	0.0383	0.0484
6	0.1088	0.2981	0.3829	0.4839
7	0.2510	> 100	> 100	> 100
8	2.5103	> 100	> 100	> 100
9	0.0139	0.0738	0.1635	1.5021
10	0.1390	0.7381	1.6355	15.0211
11	0.0071	0.0235	0.0331	0.0474
12	0.0715	0.2349	0.3312	0.4737
13	0.0192	> 100	> 100	> 100
14	0.1925	> 100	> 100	> 100
15	0.0083	0.0443	0.0981	0.9013
16	0.0834	0.4429	0.9813	9.0127
17	0.0053	0.0194	0.0292	0.0464
18	0.0532	0.1938	0.2918	0.4640

2.7 Managerial Insights

In this chapter, we presented an optimization approach that accounts for demand and return dynamics in deterministic recovery management under a linear discounted cost regime. We showed that there exist another motivation, besides of cycle stock, safety, and seasonal stocks, for keeping inventory, which arises from a dynamic environment. An *anticipation stock* is used if returns arrive at times when they can not be used immediately, but are stored in order to satisfy demand some time later and therefrom realizing a cost advantage.

The model presented allows the determination of return collection and recovery time intervals. The optimal control framework seems fairly promising to investigate several other aspects of reverse logistics. First, the simultaneous determination of the adjoint trajectories to serviceables and recoverables inventories yields a dynamic economic valuation of returned products. Such value is not simply given by the remanufacturing cost advantage $c_p + c_w - c_r$ but also must account for holding costs from return to reuse period. The first return not disposed of in a collection time interval has a value of $\lambda_u = -c_w$ because of the assumption that undesired returns cannot be rejected. The last collected return, which is immediately recovered and sold, has a value equal to the difference of production and remanufacturing cost rate $\lambda_u = c_p - c_r$. For returns being collected and kept in inventory, the respective inventory holding costs have to be subtracted. This reasoning yields a property regarding a Maximal Holding Time of returns (τ_{μ}) which together with inventory conditions was used to construct a forward solution algorithm. Another interesting implication is that the Maximal Length Property can be used to determine the least required length of the planning horizon within a rolling planning environment (see e.g. Inderfurth et al. (2004) for an overview on production planning for product recovery management). This is due to that fact that only demand and return information for the next τ_u periods are needed in order to answer the question of whether to keep a currently returned item or not.

Further, an optimal way to deal with initial inventories was presented. Since it was shown that it is never optimal to build up a serviceables inventory, after depletion of a given initial stock by filling complete demand for a certain period, production and remanufacturing rates are synchronized with demand rate, which is always possible in the absence of capacity constraints. Regarding an initial recoverables stock, a desired initial recoverables inventory level was determined, which gives the maximum amount of returns that can be used within (economically) reasonable time. If initial stock exceeds desired stock, excess returns have to be disposed of.

In order to investigate commonalities and differences of undiscounted cost (see Minner and Kleber (2001)) and discounted cash flow approach, we compared the results and determined an HC holding cost rate leading to the same optimal solution as our DCF approach. Using an appropriate approximation method, this holding cost rate could be separated into out-of-pocket holding and interest rate based opportunity cost of capital, where the 'value' of returned products is given as an (unweighted) average of the direct remanufacturing advantage $c_p - c_r$ and the opportunity costs saved by not disposing of the item $-c_w$.

The assumption of linear unit costs for production, remanufacturing, and disposal concerns most business practitioners and with appropriate extension it allows for more insights with respect to capacity aspects and economies of scale. When explicitly considering capacity constraints, an optimal production/inventory policy smoothes demand variations. These aspects are considered in more detail in Chapter 3. Economies of scale are often justified by cost degression under learning curve effects, which are discussed in Chapter 4.

2.8 Proofs and Derivations

Proof (Proof of Proposition 2.1).

The proof is divided into three parts. We show that, (i) a production impulse and (ii) a remanufacturing impulse are never optimal, whereas (iii) a disposal impulse can be excluded for all t > 0.

(i) Assume that there exists a time point θ where a jump in y_s takes place by an impulse production quantity $v_p > 0$. This quantity is used to satisfy demands from θ until $\theta + \omega$, $\omega > 0$. Therefore, $v_p = \int_{\theta}^{\theta+\omega} d(t)dt$. The difference $\Delta NPV_{(i)}$ between the impulse control strategy and the financial impact that arise from synchronizing demand with production is given by

$$\Delta NPV_{(i)} = e^{-\alpha\theta}c_p v_p + \int_{\theta}^{\theta+\omega} e^{-\alpha t} h_s \left(v_p - \int_{\theta}^{t} d(s) ds \right) dt - \int_{\theta}^{\theta+\omega} e^{-\alpha t} c_p d(t) dt \,.$$

Since

$$e^{-\alpha\theta}c_pv_p - \int_{\theta}^{\theta+\omega} e^{-\alpha t}c_p d(t) \, dt > e^{-\alpha\theta}c_pv_p - e^{-\alpha\theta}c_p \int_{\theta}^{\theta+\omega} d(t) \, dt = 0$$

 $e^{-\alpha b}$ it follows

$$\Delta NPV_{(i)} > h_s \int_{\theta}^{\theta+\omega} e^{-\alpha t} \left(v_p - \int_{\theta}^{t} d(s) ds \right) dt > 0.$$

Thus, synchronizing demand with production is always superior to impulse production.

(*ii*) Assume that there exists a time point θ where an impulse remanufacturing quantity v_r leads to jumps both in y_s and y_u . Then, the difference between this strategy and synchronizing demand and remanufacturing rate is

$$\Delta NPV_{(ii)} = e^{-\alpha\theta}c_r v_r + \int_{\theta}^{\theta+\omega} e^{-\alpha t} h_s \left(v_r - \int_{\theta}^{t} d(s)ds\right) dt - \int_{\theta}^{\theta+\omega} e^{-\alpha t} h_u \left(v_r - \int_{\theta}^{t} d(s)ds\right) dt - \int_{\theta}^{\theta+\omega} e^{-\alpha t} c_r d(t) dt > (h_s - h_u) \int_{\theta}^{\theta+\omega} e^{-\alpha t} \left(v_r - \int_{\theta}^{t} d(s)ds\right) dt,$$

which is larger than zero because of $h_u < h_s$.

(*iii*) Assume that there exists a time point $\theta > 0$ where an impulse disposal quantity $v_w > 0$ leads to a jump in y_u . Because the return rate is finite, there

exists a point $\theta - \omega$, $\omega > 0$ where $y_u(\theta - \omega) > 0$. Then, an earlier disposal $w(\theta - \omega) = \min\{v_w, y_u(\theta - \omega)\}$ saves holding costs of $\int_{\theta - \omega}^{\theta} e^{-\alpha t} h_u w(\theta - \omega) dt > 0$ and the objective value changes by

$$\begin{split} \Delta NPV_{(iii)} &= e^{-\alpha(\theta-\omega)} c_w w(\theta-\omega) - \int_{\theta-\omega}^{\theta} e^{-\alpha t} h_u w(\theta-\omega) dt - e^{-\alpha \theta} c_w w(\theta) \\ &= \left(e^{-\alpha(\theta\omega)} - e^{-\alpha \theta}\right) c_w w(\theta-\omega) - \int_{\theta-\omega}^{\theta} e^{-\alpha t} h_u w(\theta-\omega) dt \\ &= \left(e^{-\alpha(\theta\omega)} - e^{-\alpha \theta}\right) c_w w(\theta-\omega) + \frac{h_u}{\alpha} \left(e^{-\alpha \theta} - e^{-\alpha(\theta-\omega)}\right) w(\theta-\omega) \\ &= \left(e^{-\alpha(\theta\omega)} - e^{-\alpha \theta}\right) \left(c_w - \frac{h_u}{\alpha}\right) w(\theta-\omega) \end{split}$$

 $NPV_{(iii)}$ exceeds zero because of assumption (2.6). Disposal activities will therefore be carried out as soon as possible. Thus, a positive impulse quantity can only appear at time zero, as no earlier disposal is possible.

Proof (Proof of $\lambda_0 = 1$).

We only consider the cases $\lambda_0 = 0$ and $\lambda_0 = 1$. All other cases can be transformed to the case $\lambda_0 = 1$ by rescaling of λ_s and λ_u . First, let us assume $\lambda_0 = 0$. Then, the Lagrangian simplifies to

$$L(..) = (\lambda_s + \mu_1)p + (\lambda_s - \lambda_u + \mu_2)r + (-\lambda_u + \mu_3)w - \lambda_s d + \lambda_u r + k_1 y_s + k_2 y_u,$$
(2.56)

and necessary conditions (2.15)-(2.17) change to

$$\frac{\partial L}{\partial p} = \lambda_s + \mu_1 = 0, \qquad (2.57)$$

$$\frac{\partial L}{\partial r} = \lambda_s - \lambda_u + \mu_2 = 0, \qquad (2.58)$$

$$\frac{\partial L}{\partial w} = -\lambda_u + \mu_3 = 0. \tag{2.59}$$

From non-negativity of multipliers μ_1, μ_2, μ_3 it follows that

 $\lambda_s \le 0, \lambda_s - \lambda_u \le 0, \lambda_u \ge 0.$

Condition $(\lambda_0, \lambda_s, \lambda_u) \neq 0$ for a non-trivial solution requires at least λ_s or λ_u to be different from zero. A positive remanufacturing rate r > 0 would require $\lambda_s = \lambda_r = 0$ which contradicts non-triviality (because of $\mu_2 = 0$). Therefore, remanufacturing cannot take place, which leads to an obviously non-optimal solution as $c_p + c_w > c_r$.

Proof (Proof of Proposition 2.2).

From (2.23) and (2.24) it follows that $k_1 = k_2 = 0$. Thus, (2.21) and (2.22) imply

$$\dot{\lambda}_s = \alpha \lambda_s + h_s \text{ and } \dot{\lambda}_u = \alpha \lambda_u + h_u.$$
 (2.60)

p > 0 requires $\mu_1 = 0$ in (2.18) which yields $\lambda_s = c_p$ in (2.15). It follows that $\dot{\lambda}_s = 0$ which contradicts (2.60). u > 0 requires $\mu_2 = 0$ in (2.19) which yields $\lambda_s - \lambda_u = c_r$ in (2.16). It follows that $\dot{\lambda}_s = \dot{\lambda}_u$ which contradicts the assumption that $h_s > h_u$. w > 0 requires $\mu_3 = 0$ in (2.20) which yields $\lambda_u = -c_w$ in (2.17). It follows that $\dot{\lambda}_u = 0$ which contradicts assumption (2.8). Proof (Proof of Proposition 2.3). From (2.24) it follows that $k_2 = 0$. Thus, (2.22) implies

$$\dot{\lambda}_u = \alpha \lambda_u + h_u. \tag{2.61}$$

The proof for $w^* = 0$ is the same as in Case 1. p > 0 and (simultaneously) r > 0 requires $\mu_1 = \mu_2 = 0$ from (2.18) and (2.19) which yields $\lambda_s = c_p$ in (2.15) and $\lambda_u = c_p - c_r$ in (2.16). It follows that $\dot{\lambda}_u = 0$ which contradicts (2.61). In any interval where p > 0 and r = 0 hold, the definition of Case 2 ($\dot{y}_s = 0$) requires p = d and $\dot{y}_u = u \ge 0$. From (2.12) and (2.13) it follows $\lambda_u > c_p - c_r$ and $\dot{\lambda}_u = h_u$. Continuity of λ_u within a Case 2 interval (which follows from (2.26)) requires that the optimal control never switches back to r > 0, and therefore, y_u will never decrease, which contradicts the final condition $y_u(T) = 0$. The optimal policy in Case 2 is therefore given by $p^* = 0$ and $r^* = d$. (2.13) necessitates $\lambda_s = \lambda_u + c_r$ and $\dot{\lambda}_s = \dot{\lambda}_u$.

Proof (Proof of Proposition 2.4). From (2.23) it follows that $k_1 = 0$. (2.21) implies

$$\dot{\lambda}_s = \alpha \lambda_s + h_s. \tag{2.62}$$

p > 0 requires $\mu_1 = 0$ which yields $\lambda_s = c_p$. It follows that $\dot{\lambda}_s = 0$ which contradicts (2.62). r = 0, w = 0 requires $\dot{y}_u = u$ which is not feasible as long as $y_u = 0$. r > 0, w > 0 requires $\mu_2 = \mu_3 = 0$ in (2.19) and (2.20) which yields $\lambda_u = -c_w$ in (2.17) and $\lambda_s = c_r - c_w$ in (2.16). It follows that $\dot{\lambda}_s = 0$ which again contradicts (2.62). r > 0, w = 0 requires $\mu_2 = 0$ in (2.19) which yields $\lambda_s - \lambda_u = c_r$ in (2.16). It follows that $\dot{\lambda}_s = \dot{\lambda}_u$. Inserting (2.62) and (2.22) yields $k_2 = \alpha(-c_r) + h_u - h_s$ and $k_2 \ge 0$ in (2.24) contradicts the assumptions that $h_s > h_u$ and $c_r > 0$. The optimal policy is therefore given by $r^* = 0$ and $w^* = u$, which is required in order to stay in Case 3 ($\dot{y}_u = 0$ has to hold), and from (2.14), $\lambda_u = -c_w$.

Proof (Proof of Proposition 2.5).

From the definition of Case 4 it follows $\dot{y}_s = \dot{y}_u = 0$ which implies p + r = dand r + w = u. When neglecting boundary situations, i.e. where demand or return rate equal zero, this already excludes the alternatives (p = 0, r = 0) and (r = 0, w = 0). p > 0, r > 0, w > 0 requires $\mu_i = 0 \forall i$ which yields $\lambda_s = c_p$ in (2.15), $\lambda_u = -c_w$ in (2.17). Then, (2.16) becomes $c_p + c_w = c_r$ which contradicts the assumption of a positive recovery advantage (2.6). Similarly, for p > 0, r = 0, w > 0 we find $\lambda_s = c_p, \lambda_u = -c_w$ and $\mu_2 = c_r - c_p - c_w$ which has to be non-negative (2.19). Again, this contradicts assumption (2.6). For p > 0, r > 0, w = 0 we find r = u to ensure $\dot{y}_u = 0$ and then p = d - u to ensure $\dot{y}_s = 0$. This configuration is only feasible if $d \ge u$. For p = 0, r > 0, w > 0 we find r = d to ensure $\dot{y}_s = 0$ and then w = u - d to ensure $\dot{y}_u = 0$.

Proof (Proof of Proposition 2.6).

In the following we use the results derived in Feichtinger and Hartl (1986),

especially Corollary 6.3 (p.168). From (2.25) we know that λ_s is always continuous when $y_s > 0$ holds, i.e. inside Cases 1 and 3. The same is true for λ_u if $y_u > 0$ inside Cases 1 and 2. Thus, we will restrict our analysis to examine (*i*) time points inside intervals where $y_s = 0$ or $y_u = 0$ holds and (*ii*) entry and exit points of such intervals.

(i) A constraint qualification guarantees the continuity of the adjoint variables λ_s or λ_u inside intervals defined by $y_s = 0$ or $y_u = 0$, respectively. Thus, continuity of the respective adjoint variables is given, if the matrix (with line numbers given on the right hand side)

$$\begin{pmatrix} 1 & 0 & 0 & p & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & r & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & w & 0 & 0 \\ 1 & 1 & 0 & 0 & 0 & y_s & 0 \\ 0 & -1 & -1 & 0 & 0 & 0 & y_u \end{pmatrix} \begin{bmatrix} I \\ III \\ IV \\ V \end{bmatrix}$$
(2.63)

has full rank of five. First, we note that the first three rows are always independent. Analysis of the above matrix yields that the constraint qualification is not satisfied in three situations.

- If p = 0, r = 0 and $y_s = 0$ then I + II = IV. This can only happen in Cases 2 and 4 if d = 0. As we already know, in Case 2, λ_u is continuous and $\lambda_s = \lambda_u + c_r$ is therefore also continuous. If one assumes that in Case 4, u > d = 0 holds, then $\lambda_s = c_r - c_w$ proves continuity.
- If r = 0, w = 0 and $y_u = 0$ then -II III = V. This can only happen in Cases 3 and 4 if u = 0. In Case 3, $\lambda_u = -c_w$ and in Case 4, under the condition d > u = 0, $\lambda_u = -c_p - c_r$ holds. Thus, λ_u does not jump.
- If p = 0, w = 0, $y_s = 0$ and $y_u = 0$ then I III = IV + V. This situation occurs in Case 4 ($y_s = 0, y_u = 0$) when demand equals returns and the policy switches from p > 0, r > 0 to r > 0, w > 0 or vice versa. Only in this case we find a discontinuity of both adjoint variables. The height of both jumps is $\eta_s = \eta_u = c_p + c_w - c_r$. Note, this situation includes the conditions neglected before (u = d = 0).

(*ii*) Let θ_s^1 be the entry time of an interval where $y_s = 0$ holds. Then, λ_s is continuous at this time point if y_s enters this interval in a non-tangential way, i.e. $\dot{y}_s = p + r - d$ jumps. This happens if one of the controls p and r jumps at θ_s^1 . An entry point θ_s^1 can only be present at a switch from Cases 1/3 to Cases 2/4. There, r jumps if $d(\theta_s^1) > 0$ which must hold to ensure leaving the first cases. An exit time θ_s^2 does not exist because it will never be optimal to build up a serviceables stock, which can easily be seen from the production and remanufacturing decisions which will in sum never exceed the demand rate.

Let θ_u^1 be the entry time and θ_u^2 the exit time of an interval, where $y_u = 0$ holds. Then, λ_u is continuous at these time points if y_u enters or leaves this

interval in a non-tangential way, i.e. $\dot{y}_u = u - r - w$ jumps. That happens again if one of the controls r and w jumps at θ_u^1 or θ_u^2 , respectively. This will always be the case except for u = d at one of the time points.

Proof (Proof of Proposition 2.7).

From (2.14) we get a minimal value for λ_u of $\lambda_u^{\min} = -c_w$. For Case 2 we can give an upper bound for $\lambda_u^{\max} = c_p - c_r$ which together with the continuity of λ_u (Proposition 2.6) and (2.35) yields

$$c_p - c_r \ge \left(-c_w + \frac{h_u}{\alpha}\right)e^{\alpha \tau_u} - \frac{h_u}{\alpha}$$

where $\tau_u = t - \theta_{e,2}$. Solving for τ_u yields

$$\alpha(c_p - c_r) + h_u \ge (-\alpha c_w + h_u) e^{\alpha \tau_u} \Leftrightarrow e^{\alpha \tau_u} \le \frac{\alpha(c_p - c_r) + h_u}{-\alpha c_w + h_u}$$
$$\Leftrightarrow \tau_u \le \frac{1}{\alpha} \ln \left(\frac{\alpha(c_p - c_r) + h_u}{-\alpha c_w + h_u} \right)$$

Proof (Proof of Proposition 2.9).

Assume that the inequality in Proposition 2.9 holds and that at both time points $u(\theta_{e,2}) > d(\theta_{e,2})$ and $u(\theta_{x,2}) < d(\theta_{x,2})$, which is necessary to build up a stock or to decrease it, respectively. From Proposition 2.6 it follows that λ_u will not jump at these time points and $\lambda_u(\theta_{e,2}) = -c_w$, $\lambda_u(\theta_{x,2}) = c_p - c_r$. Considering that there can be no jump point of λ_u inside the interval I and that λ_u grows at rate $\dot{\lambda}_u = \alpha \lambda + h_u$ yields $\theta_{x,2} - \theta_{e,2} = \tau_u$ which contradicts the assumption.

Proof (Proof of Proposition 2.10). It is optimal to replace production by remanufacturing as long as the cost advantage is larger than the costs that arise from holding recoverables inventory, i.e. for at most τ_u time units. Equation (2.48) ensures, that initial inventories are in fact needed to satisfy demand within the interval.

Proof (Derivation of Equation (2.55)).

For a first order MacLaurin approximation of (2.54) (A) the value of m_h as well as (B) its first derivative at $\alpha = 0$ have to be determined. Thereby, the following relationship will be used

$$\frac{\partial}{\partial \alpha} \ln \left(\frac{\alpha(c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}} \right) = \frac{h_u^{DCF}(c_p + c_w - c_r)}{(\alpha(c_p - c_r) + h_u^{DCF})(-\alpha c_w + h_u^{DCF})}$$

(A) Since in the first term of (2.54) both numerator and denominator tend to 0 as α tends to 0, l'Hôpital's rule must be applied

$$\lim_{\alpha \to +0} m_h = \frac{"0"}{0} - h_u^{DCF} = \lim_{\alpha \to +0} \frac{c_p + c_w - c_r}{\frac{h_u^{DCF}(c_p + c_w - c_r)}{(\alpha(c_p - c_r) + h_u^{DCF})(-\alpha c_w + h_u^{DCF})}} - h_u^{DCF}$$
$$= \frac{(h_u^{DCF})^2}{h_u^{DCF}} - h_u^{DCF} = 0.$$

(B) The first derivative of (2.54) is given by

$$\frac{\partial m_h}{\partial \alpha} = \frac{\left(c_p + c_w - c_r\right)\ln X - \frac{\alpha h_u^{DCF} \left(c_p + c_w - c_r\right)^2}{\left(\alpha \left(c_p - c_r\right) + h_u^{DCF}\right)\left(-\alpha c_w + h_u^{DCF}\right)}}{\left[\ln X\right]^2} \quad (2.64)$$

where $X = \left(\frac{\alpha(c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}}\right)$. Again, l'Hôpital's rule is used in order to find the right hand side limit of (2.64) as α tends to 0, i.e.

$$\begin{split} \lim_{\alpha \to +0} & \frac{\partial m_h}{\partial \alpha} = \frac{``0"}{0} \\ &= \lim_{\alpha \to +0} \frac{(c_p + c_w - c_r) \left(1 - \frac{(h_u^{DCF})^2 + \alpha^2 c_w (c_p - c_r)}{(\alpha (c_p - c_r) + h_u^{DCF})(-\alpha c_w + h_u^{DCF})}\right)}{2 \ln \left(\frac{\alpha (c_p - c_r) + h_u^{DCF}}{-\alpha c_w + h_u^{DCF}}\right)} = \frac{``0"}{0} \\ &= \lim_{\alpha \to +0} \frac{-2\alpha c_w (c_p - c_r) + \frac{((h_u^{DCF})^2 + \alpha^2 c_w (c_p - c_r))(h_u^{DCF} (c_p - c_r) - hc_w - 2\alpha c_w (c_p - c_r))}{(\alpha (c_p - c_r) + h_u^{DCF})(-\alpha c_w + h_u^{DCF})} \\ &= \frac{\left(\frac{(h_u^{DCF})^2 (h_u^{DCF} (c_p - c_r) - hc_w)}{2h_u^{DCF}}\right)}{2h_u^{DCF}} = \frac{c_p - c_r}{2} + \frac{-c_w}{2}. \end{split}$$

Thus, the first order MacLaurin approximation of m_h is given by

$$m_h = \left(\frac{c_p - c_r}{2} + \frac{-c_w}{2}\right)\alpha + O(\alpha^2).$$

On the Effects of Capacity Constraints in Product Recovery

3.1 Motivation

In the previous chapter, a product recovery system with dynamic demand and return streams was investigated. It was found that a recoverables inventory is built up if there are some excess returns which can later be used to replace production with remanufacturing and thereby realizing a cost advantage of recovering returned items, which is for a certain time period larger than the resulting out-of-pocket holding costs and the decrease of the recovery cost advantage due to discounting. When reconsidering the optimal policies and their dynamics, there are a number of reasons which necessitate highly flexible processes. Firstly, since the serviceables inventory is not used, production and remanufacturing rates are synchronized with demand and undergo (in sum) the same variations. Secondly, in situations where it is optimal to have a positive recoverables inventory, demand is served from remanufacturing alone, which also means that the production rate is zero. Finally, at time points where stock-keeping starts or where the recoverables inventory is depleted, both production and remanufacturing rates undergo sudden and considerable changes, e.g. the production rate jumps from zero up to the difference of demand and return rates.

In many practical situations, processes are not that flexible and therefore, a smoothing of manufacturing and remanufacturing volumes is required. In medium-range aggregate production planning of a pure production system, if it is not possible to influence the productive capacity ('hard' constraint), a level strategy (see Silver et al. (1998), Chapter 14) is employed where a serviceables inventory is built up at times when the constraint is not binding and used up when it is. An application of optimal control theory to a production/inventory model is presented e.g. by Gaimon (1988). There, in addition to production quantities, optimal prices and capacity expansion are determined in order to maximize total profits. Another modeling option for inflexible processes is to use convex cost functions, which represent 'soft' constraints where regular capacity can be extended at increasing (overtime) unit costs. An application of convex cost functions has been presented by Kistner and Dobos (2000) for a product recovery system without disposal option. But there it is assumed that the number of returns at t directly depends on the current demand rate at t, i.e. it is modeled as a fraction of demand. Finally, process levels are smoothed if there are costs of changing e.g. the production rate or deviating from an optimal level. A production/inventory model as presented by Thompson and Sethi (1980) for instance penalizes the deviation from a target production rate and inventory level by using quadratic cost functions.

In this chapter, we deal with 'hard' capacity constraints that cannot be influenced. These restrictions can be present in different ways. If the processes take place in different facilities, independent capacity constraints, i.e. maximum production $\bar{p} > 0$ or remanufacturing rate $\bar{r} > 0$, are given. A similar approach can be applied if, due to legislative regulations, also the disposal rate is limited. If production and remanufacturing take place in a common facility, the joint capacity usage can be limited. Given a total capacity $\bar{a} > 0$ and capacity requirements coefficients $a_p > 0$ and $a_r > 0$ for both processes this can be modeled as

$$a_p p(t) + a_r r(t) \le \bar{a}. \tag{3.1}$$

The analysis of bottleneck situations, i.e. situations where a constraint becomes binding, adds another motive for holding inventory because in situations where e.g. demand exceeds the current total capacity of the system, it is necessary to have a positive (serviceables or recoverables) inventory. An optimal solution must therefore answer the question which inventory to use. Going one step further, an issue of operations strategy would be to choose from different capacity levels or expansion paths. In such a case, capacity constraints would be functions of time. A discussion of the connection of classical inventory management and capacity expansion strategies is for instance to be found in Slack and Lewis (2003). The optimal control framework used to solve dynamic problems of product recovery is able to deal with capacity dynamics because there is no general reliance on static parameters. Although not explicitly stated below, the following analysis is therefore not only restricted to stationary constraints but these can also be assumed to be (given) continuous functions of time.

In order to account for the medium and long term horizon of capacity expansion, the application of a discounted cash flow approach as introduced in Chapter 2 is continued in the following. Changes that are relevant when assuming a zero discount rate are stated explicitly. A simple application to product recovery is presented in Chapter 5, where the questions are answered on when to make an (unrestricted) remanufacturing process available, if at all.

The remainder of this chapter is organized as follows. Single capacity constraints for production and remanufacturing are considered and pure effects (i.e. under exclusion of situations where stock-keeping already took place in the basic model) of introducing the respective restriction are discussed. The investigation is supported by several numerical examples. For technical reasons and in order to simplify the discussion, demand and return rates are assumed to be strictly positive, i.e.

$$d(t) > 0 \text{ and } u(t) > 0.$$
 (3.2)

3.2 Limited Production Capacity

3.2.1 Changes to the Basic Model

In this section, the effects of a single capacity constraint are shown. Thereby, it is assumed that production and remanufacturing take place in different facilities and the latter process is only limited by available returns. Thus, solely the production process is restricted

$$p(t) \le \bar{p}(t). \tag{3.3}$$

An optimization problem is given by (2.9) and an additional constraint (3.3).

Since both sources to satisfy demand are limited, a feasible solution exists if for every point in time $0 \le t \le T$ the cumulative demand up to this point reduced by initial serviceables does not exceed the sum of maximal production and cumulative returns, i.e.

$$\int_0^t d(s)ds - y_s^0 \le y_u^0 + \int_0^t (\bar{p}(s) + u(s))ds \quad \forall t \in [0, T].$$
(3.4)

In order to simplify our discussion it is assumed that there exist no 'points of contact' of demand, return and constraint functions, e.g. if demand and constraint rates meet $(d(t) = \bar{p}(t))$ then they truly intersect $(\dot{d}(t) \neq \dot{\bar{p}}(t))$.

3.2.2 Properties of an Optimal Solution

Necessary Conditions

As only a pure control constraint was added to control optimization model (2.9), the Hamiltonian (2.10) does not change. By defining a Langrange multiplier μ_4 in order to account for the additional restriction (3.3), the Lagrangian is

$$L(..) = H(.) + \mu_1 \cdot p + \mu_2 \cdot r + \mu_3 \cdot w + \mu_4 \cdot (\bar{p} - p) + k_1 \cdot y_s + k_2 \cdot y_u.(3.5)$$

The proof of $\lambda_0 = 1$ can be accomplished in the same way as in the basic model and is therefore not treated here. Proceeding as in Section 2.3.2, necessary conditions (2.12)–(2.24) change as follows. The Hamiltonian maximizing condition (2.12) now reads

50 3 On the Effects of Capacity Constraints in Product Recovery

$$p^* = \begin{cases} 0 & \lambda_s < c_p \\ \text{singular } \lambda_s = c_p \\ \bar{p} & \lambda_s > c_p \end{cases}$$
(3.6)

equation (2.15) is replaced by

$$\frac{\partial L}{\partial p} = -c_p + \lambda_s + \mu_1 - \mu_4 = 0, \qquad (3.7)$$

and because of constraint (3.3) we get an additional non-negativity as well as a complementary slackness condition

$$\mu_4 \ge 0 \ \mu_4 \cdot (\bar{p} - p^*) = 0. \tag{3.8}$$

All other conditions remain unaltered.

In contrast to the basic model, production at the upper bound \bar{p} takes place, if the value of an item to be added to the serviceables inventory (λ_s) exceeds production unit costs. Consequently, also the value of a used product can be higher than $c_p - c_r$, as it would (arriving at a time where $\lambda_s > c_p$) allow for a higher cost advantage when replacing production.

The Structure of an Optimal Solution

Although we still distinguish between four cases with respect to the serviceables and recoverables inventory status, Cases 1 and 2 may show different optimal policies than provided in Propositions 2.2 and 2.3. Thus, the following propositions provide further sub-cases within intervals of the respective case.

Proposition 3.1 (Optimal decisions in Case 1 intervals).

If both serviceables and recoverables inventory are positive $(y_s^* > 0, y_u^* > 0)$, no items are remanufactured $(r^* = 0)$ or disposed of $(w^* = 0)$. The optimal decision on whether to produce new items or not depends on the co-state $\lambda_s < \lambda_u + c_r$ and two subcases can be distinguished:

Subcase $1(1) \Leftrightarrow \lambda_s < c_p \text{ and } \lambda_u > -c_w$ No items are produced $(p^* = 0)$. Subcase $1(2) \Leftrightarrow \lambda_s > c_p$ and $\lambda_u > c_p - c_r$ Production takes place at its upper bound $(p^* = \bar{p})$.

In contrast to the model without a capacity constraint, Subcase 1(2) shows that it is possible to produce even if there are serviceables available and if there exists a positive recoverables stock from which current demand could be satisfied completely. Such a decision only makes sense if one considers that there might be a later bottleneck situation for which both stocks are held for. It requires that the value of a serviceables item is higher than actual unit production costs and that the value of a recoverables item exceeds the difference of production and remanufacturing costs. Production takes place at its upper limit. The co-states development takes place as shown in the basic model, i.e.

$$\dot{\lambda}_s = \alpha \lambda_s + h_s \text{ and } \dot{\lambda}_u = \alpha \lambda_u + h_u.$$
 (3.9)

Because of the different decision structure in Subcase 1(2) and in contrast to the first subcase, the serviceables inventory level changes with rate $\bar{p} - d$, and it increases if demand rate is smaller than the manufacturing constraint.

Proposition 3.2 (Optimal decisions in Case 2 intervals).

If serviceables inventory is zero and recoverables inventory is positive $(y_s^* = 0, y_u^* > 0)$, no items are disposed of. The optimal decisions on production and remanufacturing depend on the co-state λ_u and on the relationship between demand and return rates. Three subcases are distinguished:

Subcase 2(1) $\Leftrightarrow -c_w < \lambda_u < c_p - c_r, \ \lambda_s < c_p$

No items are produced $(p^* = 0)$ and the remanufacturing rate equals the demand rate $(r^* = d)$.

Subcase 2(2) $\Leftrightarrow \lambda_u > c_p - c_r, \ \lambda_s = c_p, \ and \ d \leq \bar{p}$

Production equals demand rate $(p^* = d)$ and no items are remanufactured $(r^* = 0)$.

Subcase $2(3) \Leftrightarrow \lambda_u > c_p - c_r, \lambda_s > c_p$, and $\bar{p} < d$ Production takes place at its upper bound $(p^* = \bar{p})$ and remanufacturing is used to fill remaining demand $(r^* = d - \bar{p})$.

Under zero serviceables and positive recoverables inventory, there are two new subcases when considering a restricted production process. Both have in common that production becomes favorable, since the value of a serviceables item at least equals production unit costs ($\lambda_s \ge c_p$). Therefore, as many items as possible are manufactured without building up a serviceables stock ($p^* = \min\{d, \bar{p}\}$) and thus, more returns can be saved for later use as would be the case in Subcase 2(1). While in Subcase 2(2) no items are remanufactured, because their increase in value would be smaller than remanufacturing costs. In Subcase 2(3) one is indifferent between satisfying demand from remanufacturing or production of new items ($\lambda_s = \lambda_u + c_r$).

Since there is no change in optimal decisions in Cases 3 and 4, Propositions 2.4 and 2.5 are still applicable. Table 3.1 summarizes the main results of the four cases.

Optimal Transitions Between Cases and Subcases

From the optimal decisions inside Case 4 intervals it can be seen that the maximal possible demand rate that can be satisfied immediately without having inventory available is $u + \bar{p}$. If this *current capacity* of the product recovery system is exceeded by the demand rate, either serviceables have to be available in stock (as it would be the case in a pure production system) or recoverables must have been collected before in order to satisfy excess demand. Intervals where the current capacity of the system is smaller than demand are further

	p^*	r^*	w^*	\dot{y}_s	\dot{y}_u	λ_s	$\dot{\lambda}_s$	λ_u	$\dot{\lambda}_u$
Case 1: y_s	Case 1: $y_s > 0$, $y_u > 0$								
(1)	0	0	0	-d	u	$< c_p$	$\alpha\lambda_s + h_s$	$-c_w <$	$\alpha \lambda_u + h_u$
						$(\lambda_s < \lambda_u + c_r)$			
(2)	\bar{p}	0	0	$\bar{p} - d$	u	*	$\alpha\lambda_s + h_s$	$c_p - c_r <$	$\alpha\lambda_u + h_u$
						$(\lambda_s < \lambda_u + c_r)$			
Case 2: y_s	= 0, -	$y_u > 0$)						
(1)	0	d	0	0				$-c_w < \lambda_u < c_p - c_r$	$\alpha \lambda_u + h_u$
						$(\lambda_s = \lambda_u + c_r)$			
(2) $(d \le \bar{p})$		0	0		u	c_p	0		$\alpha \lambda_u + h_u$
$(3) \ (\bar{p} < d)$	\bar{p}	$d-\bar{p}$	0	0				$c_p - c_r <$	$\alpha \lambda_u + h_u$
						$(\lambda_s = \lambda_u + c_r)$			
Subcases 2(Subcases 2(2) and (3) generalized: $p^* = \min\{d, \bar{p}\}, r^* = \max\{0, d - \bar{p}\}, w^* = 0$								
Case 3: y_s	Case 3: $y_s > 0, y_u = 0$								
	0	0	u	-d	0	$< c_r - c_w$	$\alpha\lambda_s + h_s$	$-c_w$	0
Case 4: $y_s = 0, y_u = 0$									
(1) $(d \le u)$	0	d	u - d	0	0	$c_r - c_w$	0	$-c_w$	0
(2) $(u < d)$			0	0	0	c_p	0	$c_p - c_r$	0
Case 4 generalized: $p^* = \max\{d - r^*, 0\}, r^* = \min\{u, d\}, w^* = \max\{u - r^*, 0\}$									

 Table 3.1. Main results of optimal cases when considering a manufacturing constraint.

52

53

called *bottleneck situations*. The required serviceables and recoverables to deal with such situations are referred to as *bottleneck stock*.

In Section 2.3.4, we distinguished between two types of case transitions, namely forced and automatic. Here, a third type is introduced, which helps to differentiate between forced transitions that are made to build up anticipation stock (those already known from the basic model) and others which are required to deal with bottleneck situations. These are called constraint-forced transitions. Constraint-forced transitions require co-states to have a higher value than was allowed in the basic model. This becomes possible because the necessity to fill demand may require us to keep stock for a longer time period than was possible in the basic model where production always was able to satisfy demand immediately. As before, continuous and discontinuous transitions are distinguished. Subsequently, $A \to B$ reads as a transition from a Case A to a Case B interval.

In order to determine dynamic properties of the system, especially on possible case and subcase transitions, it is necessary to know under which conditions the co-states, λ_s and λ_u , are allowed to jump. Proposition 3.3 collects all situations, in which one or both co-state variables may be discontinuous.

Proposition 3.3 (Continuity of λ_s and λ_u).

 λ_s and λ_u are continuous, i.e. jump parameters η_s and η_u vanish everywhere, except at time points θ where one of the following conditions holds: (i) $2(3) \rightarrow 2(2)$:

 $\lambda_s \text{ jumps if } d(\theta) = \bar{p}(\theta) \text{ and } \dot{d}(\theta) < \dot{p}(\theta).$

$$(ii) 4(2) \to 4(1)$$
:

 $\lambda_s \text{ and } \lambda_u \text{ jump if } u(\theta) = d(\theta) \text{ and } \dot{d}(\theta) < \dot{u}(\theta).$

(iii) $1(2) \rightarrow 2(2)$ as well as $2(3) \rightarrow 1(2)$:

$$\lambda_s \text{ jumps if } d(\theta) = \bar{p}(\theta) \text{ and } d(\theta) < \dot{\bar{p}}(\theta).$$

- $(iv) 2(1) \rightarrow 4(1)$ as well as $4(2) \rightarrow 2(1)$:
- $\lambda_s \text{ and } \lambda_u \text{ jump if } d(\theta) = u(\theta) \text{ and } \dot{d}(\theta) < \dot{u}(\theta).$ (v) 2(3) \rightarrow 4(2) :

$$\lambda_s \text{ and } \lambda_u \text{ jump if } d(\theta) = u(\theta) + \bar{p}(\theta) \text{ and } d(\theta) < \dot{u}(\theta) + \dot{\bar{p}}(\theta).$$

The following Corollaries 3.1–3.7 have been developed by using results regarding optimal decisions within the subcases as well as co-states properties, especially by using Proposition 3.3. A transition from one case to another is excluded if (1) optimal decisions and requirements on parameter relation together do not allow for accumulation or depletion of a certain inventory, (2) an upward jump in one or both co-states is required, or (3) downward jumps are necessitated for other switches than those stated above.

Corollary 3.1. Within Case 1 Subcase 1(1) can be followed by Subcase 1(2). This constraint-forced and continuous transition requires the value of a serviceables item to equal production unit costs ($\lambda_s = c_p$). At switching time, the value of recoverables must exceed the difference of unit production and remanufacturing costs ($\lambda_u > c_p - c_u$). Case 1(1) only allows for a maximum value of the serviceables held in stock. When surpassing c_p without emptying the serviceables stock, a switch to Subcase 1(2) must take place. A reverse transition is not possible because it would require a jump in the serviceables co-state λ_s which is forbidden in an interval with positive inventory level.

Corollary 3.2. Within Case 2 the following transitions are possible: $2(1) \rightarrow 2(2)$. This constraint-forced and continuous transition requires $\lambda_s =$

- $c_p, \lambda_u = c_p c_r, and d \leq \bar{p}.$
- $2(1) \rightarrow 2(3)$. This constraint-forced and continuous transition requires $\lambda_s = c_p$, $\lambda_u = c_p c_r$, and $d > \bar{p}$.
- $2(3) \rightarrow 2(2)$. This automatic and discontinuous transition requires demand to intersect the production constraint from above $(d = \bar{p} \text{ and} \dot{d} < \dot{\bar{p}})$.

As the value of a recoverables item reaches the difference of unit production and remanufacturing cost, a transition from Subcase 2(1) to one of the other subcases takes place. Whether Subcase 2(2) or 2(3) is reached depends on the relation between demand rate and constraint. If this relation changes, i.e. demand gets smaller than the constraint, a transition from Subcase 2(3) to 2(2) takes place.

Corollary 3.3. Within Case 4 Subcase 4(2) is followed by Subcase 4(1). This automatic and discontinuous transition requires demand to intersect the return rate from above $(d = u \text{ and } \dot{d} < \dot{u})$.

Other transitions between subcases within a case are not optimal. Now, we turn to transitions between different cases.

Corollary 3.4. Starting at a Case 1 interval where $y_s > 0$ and $y_u > 0$ the following transitions are possible:

- $1(1) \rightarrow 2(1)$. This automatic and continuous transition requires $\lambda_s = \lambda_u + c_r$ and $\lambda_u < c_p - c_u$.
- $1(1) \rightarrow 2(2)$. This constraint-forced and continuous transition requires $\lambda_s = c_p$, $\lambda_u > c_p c_u$, and $d \leq \bar{p}$.
- 1(1) \rightarrow 2(3). This constraint-forced and continuous transition requires $\lambda_s = c_p, \ \lambda_u > c_p c_u, \ and \ d > \bar{p}.$
- $1(2) \rightarrow 2(2)$. This automatic and discontinuous transition requires $d = \bar{p}$ and $\dot{d} < \bar{p}$.
- $1(2) \rightarrow 2(3)$. This automatic and continuous transition requires $\lambda_s = \lambda_u + c_r$ and $d > \bar{p}$.

Since returns are neither remanufactured nor disposed of, a Case 1 interval terminates into a Case 2 subcase when depleting the serviceables inventory. Depending on the value of recoverables λ_u , Case 1(1) switches either to Case 2(1) (if $\lambda_u < c_p - c_u$) or to Cases 2(2)/2(3). Case 1(2) can be left if demand exceeds the manufacturing constraint at the end if this interval. Since $\lambda_u > c_p - c_u$ also holds there, only transitions to Cases 2(2)/2(3) are possible.

Corollary 3.5. Starting at a Case 2 interval where $y_s = 0$ and $y_u > 0$ the following transitions are possible:

- $2(1) \rightarrow 4(1)$. This automatic and discontinuous transition requires d = uand $\dot{d} \leq \dot{u}$.
- $2(1) \rightarrow 4(2)$. This automatic and continuous transition requires $u \leq d$, $\lambda_s = c_p$, $\lambda_u = c_p c_r$.
- $2(2) \rightarrow 1(2)$. This constraint-forced and continuous transition requires $d < \bar{p}$.
- $2(3) \rightarrow 1(2)$. This constraint-forced and discontinuous transition requires $d = \bar{p}, \ \dot{d} < \dot{\bar{p}}.$
- $2(3) \rightarrow 4(2)$. This automatic and discontinuous transition requires $d = \bar{p}+u$, $\dot{d} \leq \dot{p} + \dot{u}$.

In addition to the case transitions known from the basic model, a switch from Case 2(2)/(3) to Case 1(2) is possible, requiring demand not to be larger than the remanufacturing constraint $(d \le \bar{p})$. Further, a depleting recoverables inventory allows for a transition from Case 2(3) to 4(2). This requires demand to be larger than the current capacity of the system at the end of the Case 2(3) interval as well as the opposite inside the Case 4(2) interval in order to be able to satisfy demand without having a positive stock.

Corollary 3.6. Starting at a Case 3 interval where $y_s > 0$ and $y_u = 0$ the following transitions are possible:

 $3 \rightarrow 1(1)$. This forced and continuous transition requires $d \leq u$.

 $3 \rightarrow 4(1)$. This automatic and continuous transition requires $\lambda_s = c_r - c_w$ and $d \leq u$.

Transitions starting at a Case 3 interval are the same as discussed for the basic model.

Corollary 3.7. Starting at a Case 4 interval where $y_s = y_u = 0$ the following transitions are possible:

- $4(1) \rightarrow 2(1)$. This forced and continuous transition requires $d \leq u$.
- $4(2) \rightarrow 2(1)$. This forced and discontinuous transition requires d = u and $\dot{d} < \dot{u}$.
- $4(2) \rightarrow 2(2)$. This constraint-forced and continuous transition requires $d \leq \bar{p}$.
- $4(2) \rightarrow 2(3)$. This constraint-forced and continuous transition requires $d > \bar{p}$.

Adding to the already known switches, Case 4(2) can be left for a transition into a Case 2(2)/2(3) interval in order to build up a bottleneck stock.

As in the basic model, an accumulation of serviceables through a transition from any subcase of Cases 2/4 to 1(1) or 3 is excluded because building up stock would require $\dot{y}_s > 0$ which contradicts optimal decisions within the mentioned subcases with positive serviceables inventory level (1(1) and 3). Since also 1(2) does not terminate into those cases, they are only possible during an initial period. We therefore distinguish between transitions that can only occur at the beginning of the planning horizon when having a positive initial inventory level (depicted in Figure 3.1) and others that occur during



Fig. 3.1. Optimal case transitions at the beginning of the planning horizon when considering a manufacturing constraint

the planning period after using up initial stock (see Figure 3.2).

In what follows, we concentrate on the latter kind of case changes by assuming $y_s^0 = y_u^0 = 0$. A discussion on how to deal with initial stock would proceed in a similar manner as been applied for the basic model (see Section 2.4). Additional transitions have to be considered if parts of initial stock are used during a subsequent bottleneck situation.



Fig. 3.2. Optimal case transitions when considering a manufacturing constraint

3.2.3 Pure Effects of a Manufacturing Constraint

Figure 3.2 distinguishes between two reasons for keeping stock. On the one hand, as in the basic model, the anticipation of a later change in the demand/return relationship leads to an accumulation of returns in a Case 2(1) interval (shaded area in Figure 3.2). This motive only allows for using the recoverables inventory. Another cause that may also require a serviceables stock is to prepare for a bottleneck situation using Cases 2(2), 2(3) and 1(2). The required constraint-forced transitions are shown on the right hand side of Figure 3.2. A general solution to our problem requires the simultaneous consideration of both motivations, because a bottleneck situation for instance may not become binding if there is enough anticipation stock available to satisfy demand from remanufacturing returns.

To begin with, we deal with *pure effects* of a manufacturing constraint. Therefore, it is assumed that returns do not exceed demand during the planning horizon, i.e.

$$d(t) > u(t) \ \forall t \in [0, T] \tag{3.10}$$

If the opposite holds, demand can always be satisfied from (unrestricted) remanufacturing alone. Later, *combined effects* are discussed, that occur when allowing demand and return rate to intersect.

Limitation (3.10) immediately excludes Case 4(1) and Case 2(1) intervals, where the cost advantage of remanufacturing motivates stock keeping. Therefore, the number of case transitions reduces to those depicted in Figure 3.3. Two further implications result. Firstly, since recoverables can immediately



Fig. 3.3. Optimal case transitions when considering a manufacturing constraint assuming u < d

be used to serve demand, they obtain a minimal value of $\lambda_u \ge c_p - c_r$. Likewise, since production is necessary to fill part of the demand, serviceables

are not valued less than production costs, i.e. $\lambda_s \geq c_p$. Secondly, production capacity always becomes binding in intervals where demand exceeds the current production and remanufacturing capacity. This leads to the following proposition.

Proposition 3.4 (Location of a bottleneck interval).

If demand exceeds current capacity of the system, i.e.

$$d(\theta) > \bar{p}(\theta) + u_r(\theta), \qquad (3.11)$$

it is necessary to have a positive (serviceables or recoverables) inventory at θ .

Proof. Proposition 3.4 directly follows from the definition of Case 4 intervals.

In order to simplify the discussion we consider a situation with a single bottleneck interval as well as no planning horizon effects, i.e. we do not deal with situations where the treatment of a bottleneck interval requires to take the borders of the planning period into account. Furthermore, it is assumed that a valid solution exists, i.e. feasibility condition (3.4) holds.

Let $\theta_{e,b}$ the begin and $\theta_{x,b}$ the end of a bottleneck situation as described in the preceding proposition with

$$\begin{aligned} \theta_{e,b} &: d(\theta_{e,b}) = \bar{p}(\theta_{e,b}) + u(\theta_{e,b}) \text{ and } d(\theta_{e,b}) > \bar{p}(\theta_{e,b}) + \dot{u}(\theta_{e,b}), \\ \theta_{x,b} &: d(\theta_{x,b}) = \bar{p}(\theta_{x,b}) + u(\theta_{x,b}) \text{ and } \dot{d}(\theta_{x,b}) < \bar{p}(\theta_{x,b}) + \dot{u}(\theta_{x,b}) \end{aligned}$$

Reconsidering the limited current capacity of the system, at $\theta_{e,b}$ a certain number of serviceables or recoverables have to be stocked in advance to satisfy *bottleneck demand* D_B . It is defined as that part of demand during the bottleneck interval ($\theta_{e,b}, \theta_{x,b}$) which can not be satisfied by using current capacity of the system, i.e.

$$D_B = \int_{\theta_{e,b}}^{\theta_{x,b}} d(t) - (u(t) + \bar{p}(t)) dt.$$
(3.12)

Now, the question will be answered how to collect stock and under which circumstances to use each of the inventories.

In an optimal solution Case 4(2) cannot be followed by Case 1(2) or vice versa, which means that before starting an interval where both inventories are positive (Case 1(2)), there must be an interval where only returns are collected (Case 2). In a Case 1(2) interval returns are collected and only a transition back to Case 2 is possible. Therefore, the following sequences can occur in an optimal solution

Case
$$2 \rightarrow$$
 Case $4(2) \rightarrow$ Case 2 and Case $2 \rightarrow$ Case $1(2) \rightarrow$ Case 2

As no further stock is required at the end of the bottleneck interval $(\theta_{x,b})$, Case 4(2) is present afterwards. Since $d > \bar{p}$ holds at this point, a transition from Subcase 2(3) to Case 4(2) must take place. Let θ_c denote the start time of a corresponding *collection interval* ($\theta_c, \theta_{e,b}$). This time point depends on how many items have to be stored to cover bottleneck demand and how long it takes to collect them. Case 4(2) is present before θ_c . Depending on the relation between demand and maximum production rate, at θ_c either a transition from Case 4(2) to Case 2(2) or to Case 2(3) takes place. Within the Case 2 interval, recoverables inventory increases with rate

$$\dot{y}_u = \min\{u, u - (d - \bar{p})\}$$
 (inside Case 2(2)/2(3)) (3.13)

and since both switches are continuous transitions, co-states develop starting with

$$\lambda_s(\theta_c) = c_p \text{ and } \lambda_u(\theta_c) = c_p - c_r.$$
 (3.14)

During the collection interval, building up a serviceables stock and thus switching from Case 2 to Case 1(2) becomes preferable in a period where demand falls below the capacity constraint followed by another where the opposite holds. This situation technically reflects the impossibility of a transition from Subcase 2(2) to 2(3) without passing a Case 1(2) interval, and leads to the following corollary.

Corollary 3.8 (Location property of a Case 1(2) interval). Let θ denote a time point inside a collection interval where $d(\theta) = \bar{p}(\theta)$. If demand crosses the manufacturing capacity from below, i.e. $\dot{d}(\theta) > \dot{\bar{p}}(\theta)$, then it is always optimal to have a positive serviceables inventory at time θ .

In a Case 1(2) interval, the recoverables inventory increases with rate

$$\dot{y}_u = u \text{ (inside Case 1(2))} \tag{3.15}$$

which is larger than it would have been when staying in Case 2 where $d(\theta) > \bar{p}(\theta)$. Therefore, using the serviceables inventory allows to save additional returns for later use during the bottleneck interval and likewise decreases the length of the collection interval thus saving holding costs in the recoverables inventory. This reasoning will later be used to develop a maximal length criterion.

Since there might be several points meeting Corollary 3.8 within the collection period, we assume that there are $n \geq 0$ Case 1(2) intervals $J^i = (\theta^i_{e,1(2)}, \theta^i_{x,1(2)}), i = 1, 2, ..., n$. Analogously to Proposition 2.8, the following inventory conditions must hold (presented without proof).

Proposition 3.5 (Inventory Conditions of Case 1(2) intervals).

Let $J^{i} = (\theta^{i}_{e,1(2)}, \theta^{i}_{x,1(2)})$ be an open time interval where $y_{s} > 0$, $y_{u} > 0$ and $y_{s}(\theta^{i}_{e,1(2)}) = y_{s}(\theta^{i}_{x,1(2)}) = 0$. Then,

(i) cumulative production equals cumulative demand over the whole interval

$$\int_{\theta_{e,1(2)}^{i}}^{\theta_{x,1(2)}^{i}} (\bar{p}(t) - d(t)) dt = 0, \qquad (3.16)$$

(ii) at any point $\theta \in J^i$, cumulative production must be larger than cumulative demand

$$\int_{\theta_{e,1(2)}^{i}}^{\theta} (\bar{p}(t) - d(t))dt > 0.$$
(3.17)

Comparing the optimal return collection between Case 2 and a Case 1(2) intervals to be seen in (3.13) and (3.15), in the latter case additional returns are stored with a total quantity of $\int_{\theta_{x,1(2)}}^{\theta_{x,1(2)}} (\max\{\bar{p}(t) - d(t), 0\}) dt$. Using this result, inventory conditions regarding the use of a recoverables inventory can be given (presented without proof and omitting the non-negativity condition).

Proposition 3.6 (Inventory Condition for the joint collection and bottleneck interval).

Let $I = (\theta_c, \theta_{x,b})$ be an open time interval where $y_u > 0$ with $y_s(\theta_c) = y_u(\theta_c) = y_s(\theta_{x,b}) = y_u(\theta_{x,b}) = 0$ and $J^i = (\theta^i_{e,1(2)}, \theta^i_{x,1(2)}) \subset I$, i = 1, 2, ..., n be open time intervals where $y_s > 0$, $y_u > 0$ and for which Proposition 3.5 applies. Then, cumulative stored returns must equal bottleneck demand

$$\int_{\theta_c}^{\theta_{x,b}} (u(t) - \max\{d(t) - \bar{p}(t), 0\}) dt + \sum_{i=1}^n \int_{\theta_{e,1(2)}^i}^{\theta_{x,1(2)}^i} (\max\{\bar{p}(t) - d(t), 0\}) dt = 0.$$
(3.18)

As already mentioned, increasing the length of a Case 1(2) interval leads to a decreasing length of the collection period. But this is only profitable as long as the induced serviceables holding costs lead to a higher reduction of recoverables holding costs. Using this trade-off a condition for maximal Case 1(2) interval lengths can be given.

Proposition 3.7 (Maximal Length of a Case 1(2) interval).

The maximal length of a Case 1(2) interval $J^i = (\theta^i_{e,1(2)}, \theta^i_{x,1(2)})$ is time dependent and it is given by

$$\theta_{x,1(2)}^i - \theta_{e,1(2)}^i \le \frac{1}{\alpha} \ln \left(\frac{\alpha c_r + h_s - h_u}{\alpha \left(c_p - \lambda_u(\theta_{e,1(2)}^i) \right) + h_s - h_u} \right).$$
(3.19)

The reasoning behind marginal criterion (3.19) is as follows. Assume there are two possibilities to satisfy the last demand unit of a bottleneck interval. This can be done either by reducing θ_c (increase the length of the collection period) or by decreasing $\theta_{e,1(2)}^i$ (increase length of Case 1(2) interval). Independent of this (marginal) decision, the same amount of returns will be on stock at $\theta_{x,1(2)}^i$, i.e. at the end of the Case 1(2) interval. In an optimal solution, one has to be indifferent between both options and holding costs up to this time need to be balanced. A Case 1(2) interval with maximal length, however, implies the following sequence

Case
$$2(2) \rightarrow$$
 Case $1(2) \rightarrow$ Case $2(3)$.

There are two interesting implications of Proposition 3.7. On the one hand, maximal Case 1(2) interval lengths increase during a collection period. And on the other, there might be a time after which the maximal length of the Case 1(2) interval becomes infinity, because a long collection period leads to such a high value of recoverables that the serviceables co-state is not able to catch up. This is the case if the denominator in the logarithm term of 3.19 becomes negative.

Corollary 3.9. There exists no maximal holding time for a Case 1(2) interval $J^i = (\theta^i_{e,1(2)}, \theta^i_{x,1(2)})$ if it starts at a time point $\theta^i_{e,1(2)}$ which is larger than a critical value $\theta^{crit}_{e,1(2)}$

$$\theta_{e,1(2)}^{crit} : \lambda_u(\theta_{e,1(2)}^{crit}) > c_p + \frac{h_s - h_u}{\alpha}$$
(3.20)

There exist two other types of situations where (3.19) does not hold with equality. First, the collection interval cannot start earlier because there is no excess capacity available (demand would exceed capacity constraint) and thus, inventory condition (Proposition 3.5) would be violated. This situation requires $d(\theta_{e,1(2)}^i) = \bar{p}(\theta_{e,1(2)}^i)$ and $\dot{d}(\theta_{e,1(2)}^i) < \bar{p}(\theta_{e,1(2)}^i)$ and Case 1(2) is reached originating in a Case 2(3) interval. Second, a Case 1(2) interval may not be extended because of absent excess demand. This would require a switch from Case 1(2) to Case 2(2).

In rather short term problems, discounting can be neglected and co-state developments inside intervals with positive inventory simplify to linear functions, i.e. $\dot{\lambda}_u = h_u$ in intervals where $y_u > 0$ and $\dot{\lambda}_s = h_s$ if $y_s > 0$. The results with respect to cases and case transitions remain unaffected but the maximal holding time criterion given in Proposition 3.7 changes as follows

$$\theta_{x,1(2)}^{i} - \theta_{e,1(2)}^{i} \le \frac{\lambda_u(\theta_{e,1(2)}^{i}) - (c_p - c_r)}{h_s - h_u}.$$
(3.21)

Maximum holding time is reached if equality holds. As in the equivalent criterion under consideration of discounting, there is indifference between (1) increasing the Case 1(2) interval and accepting additional holding costs in the serviceables inventory or (2) enlarging the length of collection interval.
3.2.4 Mixed Effects of a Manufacturing Constraint

When allowing for intersections of demand and return functions, both the anticipation as well as the capacity motive for keeping stock play a role, if for instance during an interval where returns are collected for use during a later bottleneck interval the demand and return rates intersect. Thus, it might happen that returns exceed demand before the start of the collection interval θ_c . These excess returns can be used to replace production which would take place right after θ_c by remanufacturing of returns that have been stored before this time. Then, the collection interval is preceded by a Case 2(1) interval, where anticipation stock is built up. The same trade-off has to be struck which is already known from the basic model and the maximal length of a Case 2(1) interval is given by Proposition 2.7.

Since part of the anticipation stock is used to fill bottleneck demand, inventory condition (3.18) must be adapted as follows.

Proposition 3.8 (Inventory Condition for the joint collection and bottleneck interval, adapted for mixed effects).

Let $I = (\theta_{e,2(1)}, \theta_{x,b})$ be an open time interval where $y_u > 0$ with $y_s(\theta_c) = y_u(\theta_c) = y_s(\theta_{x,b}) = y_u(\theta_{x,b})$ and $J^i = (\theta_{e,1(2)}^i, \theta_{x,1(2)}^i) \subset I$, i = 1, 2, ..., n be open time intervals where $y_s > 0$, $y_u > 0$ and for which Proposition 3.5 applies. Let θ_c denote the time of leaving Case 2(1). Then, cumulative stored returns must equal bottleneck demand

$$\int_{\theta_{e,2(1)}}^{\theta_{c}} \left(u(t) - d(t) \right) dt + \int_{\theta_{c}}^{\theta_{x,b}} \left(r(t) - \max\{d(t) - \bar{p}(t), 0\} \right) dt \qquad (3.22)$$
$$+ \sum_{i=1}^{n} \int_{\theta_{e,1(2)}}^{\theta_{x,1(2)}^{i}} \left(\max\{\bar{p}(t) - d(t), 0\} \right) dt = 0.$$

Note that it might happen that a bottleneck situation will not become binding when demand is completely satisfied by remanufacturing anticipation stock.

3.2.5 Numerical Examples

Three examples are used to show the essential properties of optimal solutions in the presence of a limited production rate. Let demand, return, capacity constraint over a planning horizon of $T = 2\pi$ be given as stated in Table 3.2 (approximately 6 months). The discount rate is set to zero ($\alpha = 0$) and holding cost parameters are $h_s = 1.5$ and $h_u = 1$. The remaining relevant cash flow parameters as well as the beginning and end of the respective bottleneck intervals can also be found in Table 3.2.

In Example 3.1 (see Figure 3.4) demand d is generally larger than (constant) return rate u and capacity constraint \bar{p} , and for some positive time it even becomes larger than the current production and remanufacturing capacity of the system $\bar{p} + u$. Therefore, an interval with positive recoverables



 Table 3.2. Demands, returns and capacity constraint functions as well as cash flow

 parameters in three considered scenarios

Fig. 3.4. Demands, returns and capacity constraint in Example 3.1

inventory must start and terminate in Subcase 4(2) and optimal decisions in the collection interval follow Subcase 2(3). For determining the start time θ_c of the collection interval, inventory condition (3.18) can be used which simplifies to

$$\int_{\theta_c}^{\theta_{x,b}} (r(t) - \max\{d(t) - \bar{p}(t), 0\}) \, dt = 0 \tag{3.23}$$

since there is no Case 1(2) interval to be considered (n = 0). Solving (3.23) yields $\theta_c = 0.7$. Optimal decisions in Example 3.1 can be found in Figure 3.5, and the optimal co-state movements are depicted in Figure 3.6.



Fig. 3.5. Optimal solution of Example 3.1



Fig. 3.6. Optimal co-state developments in Example 3.1

In Example 3.2, which is depicted in Figure 3.7, there is a point satisfying Corollary 3.8 inside the collection interval (at 2.9) and thus, it is optimal to have a Case 1(2) interval at (and around) this time. The optimal solution must satisfy inventory conditions (3.16), (3.17), (3.18), as well as Maximal Length Condition for a Case 1(2) interval (3.21), and it is depicted in Figure 3.8. The collection interval starts at $\theta_c = 0.45$, a switch to Case 1(2) takes place at $\theta_{e,1(2)} = 1.65$ and back to Case 2(3) is located at $\theta_{x,1(2)} = 4.06$.

Since the extension of both collection and Case 1(2) intervals is not limited and because of a zero discount rate, the recoverables co-state at the begin of the Case 1(2) interval is given by $\lambda_u(\theta_{e,1(2)} = 1.65) = c_p - c_r + h_u(\theta_{e,1(2)} - t_c) =$ 3.2 and (3.21) holds with equality, yielding maximal length of the Case 1(2) interval of $\theta_{x,1(2)} - \theta_{e,1(2)} \leq 1.2/.5 = 2.4$. Equality holds except for a rounding error. The optimal co-state developments can be seen from Figure 3.9.



Fig. 3.7. Demands, returns and capacity constraint in Example 3.2



Fig. 3.8. Optimal solution of Example 3.2



Fig. 3.9. Optimal co-state developments in Example 3.2



Fig. 3.10. Demands, returns and capacity constraint in Example 3.3

Now, we will examine Example 3.3 (see Figure 3.10), where it is optimal to have a positive inventory even in the unrestricted case because demand and return functions intersect. There the maximal holding time in the recoverables inventory $\tau_u = 2$. The optimal unrestricted solution is depicted in Figure 3.11, showing a period where the capacity constraint is violated (grey shaded area).



Fig. 3.11. Optimal solution of the unrestricted problem for Example 3.3

The optimal solution under consideration of the manufacturing constraint consists of a sequence $4(1) \rightarrow 2(1) \rightarrow 2(3) \rightarrow 4(2)$. Using inventory condition (3.22) (again with n = 0, see Example 3.1) as well as maximum length property of Case 2(1) intervals both optimal time point when to switch from Case

4(1) to Case 2(1) $\theta_{e,2(1)}$ as well as time θ_c after which manufacturing takes place are determined: $\theta_{e,2(1)} = 0.6$ and $\theta_c = 2.6$. The solution of Example 3.3 is depicted in Figure 3.12, and the optimal co-state movements are to be found in Figure 3.13.



Fig. 3.12. Optimal solution of Example 3.3



Fig. 3.13. Optimal co-state developments in Example 3.3

A number of more complex examples, also including several successive bottleneck situations within the planning horizon, can be found in Baranczyk (2001). In this work, a discrete time approximation of the constraint model was solved using linear programming methods.

3.2.6 Managerial Insights

Adding a production capacity constraint to the basic model leads to an additional motivation for keeping stock, namely to prevent against bottleneck situations where current capacity of the system does not suffice to satisfy demand. In contrast to the results presented in Chapter 2, generally both inventories can be used. Since there exists a holding cost advantage, bottlenecks are primarily served using previously collected returns. The size of the bottleneck determines how many returns to save and when to start collecting returns. Building up a serviceables stock during the collection interval is beneficial in periods where excess production capacity is available. It is used in order to shorten the collection period, because more returns are accumulated. The optimal policy must balance a trade-off regarding the holding costs which can be expressed by an interval length property.

Depending on the size of the bottleneck situation and the time that is required to collect returns, holding costs might achieve substantial dimensions and it is questionable whether the assumption to satisfy all demand should be kept. When allowing for a backlogging of demands, improvements are possible because in such an environment parts of the demand during a bottleneck situation are collected and satisfied after its end. See Kiesmüller et al. (2000) for a discussion of the profitability of having backorders in a product recovery system without capacity restrictions. One step further, the question has to be answered which parts of the demand to satisfy at all. This can be answered by using a profit maximizing approach.

3.3 Limited Remanufacturing Capacity

3.3.1 Changes to the Basic Model

This section deals with the influence of a remanufacturing constraint $\bar{r}(t) > 0$ on optimal decisions in the product recovery system introduced in Chapter 2. The corresponding optimization problem is given by (2.9) and an additional constraint

$$r(t) \le \bar{r}(t). \tag{3.24}$$

Since production is unrestricted, demand can always be satisfied without producing in advance. Therefore, a condition ensuring the existence of a feasible solution as provided in Section 3.2 is not needed.

In what follows, since we are not dealing with boundary effects, starting inventory levels are set to zero, i.e. $y_s^0 = y_u^0 = 0$. As an additional simplifying condition, it is assumed that there exists no joint intersection of demand, return and maximal remanufacturing rate within the planning horizon, i.e.

$$\nexists t \in [0,T] \text{ with } d(t) = u(t) = \bar{r}. \tag{3.25}$$

Without restricting generality too much, (3.25) together with the assumption of strict positive demand and return rates leads to a considerable reduction of possible case transitions.

3.3.2 Properties of an Optimal Solution

Necessary Conditions

Proceeding in the same way as in the model with a remanufacturing constraint (see Section 3.2), the Lagrangian now reads

$$L(..) = H(.) + \mu_1 \cdot p + \mu_2 \cdot r + \mu_3 \cdot w + \mu_4 \cdot (\bar{r} - r) + k_1 \cdot y_s + k_2 \cdot y_u.$$
(3.26)

Necessary conditions (2.12)-(2.24) known from the basic model are adapted as follows. The Hamiltonian maximizing condition (2.13) changes to

$$r^* = \begin{cases} 0 \quad \lambda_s - \lambda_u < c_r \\ \text{singular } \lambda_s - \lambda_u = c_r \\ \bar{r} \quad \lambda_s - \lambda_u > c_r \end{cases}$$
(3.27)

equation (2.16) is replaced by

$$\frac{\partial L}{\partial r} = -c_r + \lambda_s - \lambda_u + \mu_2 - \mu_4 = 0, \qquad (3.28)$$

and constraint (3.24) leads to an additional non-negativity as well as complementary slackness condition

$$\mu_4 \ge 0 \ \ \mu_4 \cdot (\bar{r} - r^*) = 0. \tag{3.29}$$

The remaining conditions remain unchanged.

When introducing a remanufacturing constraint, the difference of the values of a serviceables item λ_s and a recoverables item λ_u can be higher than remanufacturing unit costs c_r because the latter might not be usable due to the capacity restriction. Since production and disposal rate are unbounded, the maximal value of serviceables is given by unit production costs ($\lambda_s \leq c_p$), and the value of returns cannot be lower than disposal revenue ($\lambda_u \geq -c_w$).

The Structure of an Optimal Solution

The introduction of a remanufacturing constraint allows for additional subcases in each of the four cases being distinguished with respect to their serviceables and recoverables status.

Proposition 3.9 (Optimal decisions in Case 1 intervals).

If both, serviceables and recoverables inventory are positive $(y_s^* > 0, y_u^* > 0)$, no items are produced $(p^* = 0)$ or disposed of $(w^* = 0)$. The optimal decision on whether to remanufacture or not depends on the relation of both co-states and two subcases can be distinguished:

Subcase $1(1) \Leftrightarrow \lambda_s - \lambda_u < c_r$ No items are remanufactured $(r^* = 0)$. Subcase $1(2) \Leftrightarrow \lambda_s - \lambda_u > c_r$ Remanufacturing takes place at its upper bound $(r^* = \bar{r})$.

In analogy to the model with a restricted production rate, it is now possible to remanufacture even if demand could be served from serviceables stock. In addition to Subcase 1(1) which is already known from the basic model there is another Subcase 1(2). There the remanufacturing rate obtains its upper bound if the change of the value of a recoverables item when remanufacturing it to serviceables stock $(\lambda_s - \lambda_u)$ exceeds unit remanufacturing costs. It requires that the value of a serviceables λ_s is located in the interval $(c_r - c_w, c_p)$ and for the value of a returned item $\lambda_u \in (-c_w, c_p - c_r)$ must hold. The inventory levels change with rates $\dot{y}_s = \bar{r} - d$ and $\dot{y}_u = u - \bar{r}$.

Proposition 3.10 (Optimal decisions in Case 2 intervals).

If serviceables inventory is zero and recoverables inventory is positive $(y_s^* = 0, y_u^* > 0)$, no items are disposed of. The optimal decisions on production and remanufacturing depend on the co-states and on the relationship between demand and return rates. Three subcases are distinguished:

Subcase $2(1) \Leftrightarrow c_r - c_w < \lambda_s < c_p, \ \lambda_u < c_p - c_r, \ d < \bar{r}$ No items are produced $(p^* = 0)$ and the remanufacturing rate equals the de-

mand rate $(r^* = d)$.

Subcase $2(2) \Leftrightarrow \lambda_s = c_p, \ \lambda_u > c_p - c_r$

Production equals demand rate $(p^* = d)$ and no items are remanufactured $(r^* = 0)$.

Subcase $2(3) \Leftrightarrow \lambda_s = c_p, \lambda_u < c_p - c_r \text{ and } \bar{r} < d$ Remanufacturing takes place at its upper bound $(r^* = \bar{r})$ and production is used to fill remaining demand $(p^* = d - \bar{r})$.

Similar to the previous section, two additional cases have to be considered in which production takes place. In Subcase 2(2) demand is filled completely from producing new items and thus, the recoverables inventory level must rise ($\dot{y}_u = u > 0$). As will be shown below, this subcase is not relevant for determining the optimal solution. Subcase 2(3) shows both remanufacturing with maximal rate as well as production. This makes sense in situations where demand is so high that it cannot be satisfied using the preferred mode.

Proposition 3.11 (Optimal decisions in Case 3 intervals).

If serviceables inventory is positive and recoverables inventory is zero $(y_s^* > 0, y_u^* = 0)$, the optimal policy is not to produce $(p^* = 0)$. The value of a returned item equals the disposal revenue $(\lambda_u = -c_w)$, and the optimal decision on whether to remanufacture or not depends on the co-state λ_s . The following subcases are possible:

Subcase $3(1) \Leftrightarrow \lambda_s < c_r - c_w$

No items are remanufactured $(r^* = 0)$ and all returns are disposed of $(w^* = u)$.

Subcase $3(2) \Leftrightarrow c_r - c_w < \lambda_s < c_p \text{ and } u > \bar{r}$

Remanufacturing takes place at its upper bound $(r^* = \bar{r})$ and remaining returns are disposed of $(w^* = u - \bar{r})$.

Although it is not optimal to produce when having a positive serviceables inventory, in a new Subcase 3(2) remanufacturing takes place with maximal possible rate. The serviceables stock develops inside this case with rate $\dot{y}_s = \bar{r} - d$ and it increases if the constraint exceeds demand and decreases if the opposite holds. Such a decision is reasonable if one considers that serviceables are accumulated by remanufacturing more items than currently required in order to satisfy demand in a situation where the remanufacturing constraint is binding. Returns that exceed the capacity are disposed of if they can not be used within a reasonable period.

Proposition 3.12 (Optimal decisions in Case 4 intervals).

If serviceables and recoverables inventories are zero $(y_s^* = 0, y_u^* = 0)$, optimal decisions depend on how demand rate, return rate, and remanufacturing constraint relate to each other and three subcases can be distinguished:

Subcase 4(1) $\Leftrightarrow d \leq u \text{ and } d < \bar{r}$

Demand is satisfied completely by remanufacturing returns $(r^* = d)$ and excess returns are disposed of $(w^* = u - d)$. No items are produced $(p^* = 0)$. Subcase $4(2) \Leftrightarrow u < d$ and $u < \bar{r}$

All returns are remanufactured $(r^* = u)$ and the missing items are produced $(p^* = d - u)$. No items are disposed of $(w^* = 0)$.

Subcase 4(3) $\Leftrightarrow \bar{r} < u \text{ and } \bar{r} < d$

Remanufacturing takes place at its upper bound $(r^* = \bar{r})$ and the missing items are produced $(p^* = d - \bar{r})$. Remaining returned items are disposed of $(w^* = u - \bar{r})$.

In the absence of both recoverables and serviceables stock, the decision on remanufacturing depends on the relation of demand, return and maximum remanufacturing rate. It is given by the minimum of those three values $(r^* = \min\{p, u, \bar{r}\})$ and the determination of how many items to produce and/or to dispose of directly follows this decision.

The value of an additional return λ_u is either given by $c_p - c_r$ if it can be used immediately (Subcase 4(2)), or it is $-c_w$ if the returned item cannot be used due to insufficient demand or capacity. This happens in Subcases 4(1) and 4(3). The value of a serviceables item λ_s is $c_r - c_w$ if demand can be satisfied from remanufacturing alone (Subcase 4(1)), requiring both enough returns and remanufacturing capacity, or it is given by c_p if the last demanded unit was satisfied from production. This is relevant in Subcases 4(2) and 4(3).

Table 3.3 summarizes the results of the four cases.

71

	p^*	r^*	w^*	\dot{y}_s	\dot{y}_u	λ_s	$\dot{\lambda}_s$	λ_u	$\dot{\lambda}_u$
Case 1: $y_s > 0, y_u > 0$									
(1)	0	0	0	-d	u - d	$< c_p$	$\alpha\lambda_s + h_s$	$-c_w <$	$\alpha \lambda_u + h_u$
						$(\lambda_s < \lambda_u + c_r)$			
(2)	0	\bar{r}	0	$\bar{r} - d$	$u-\bar{r}$	$c_r - c_w < \lambda_s < c_p$	$\alpha\lambda_s + h_s$	$-c_w < \lambda_u < c_p - c_r$	$\alpha \lambda_u + h_u$
						$(\lambda_u + c_r < \lambda_s)$			
Case 2: $y_s = 0, y_u > 0$									
$(1) \ (d < \bar{r})$	0	d	0	0	u - d	$c_r - c_w < \lambda_s < c_p$	$\alpha \lambda_u + h_u$	$-c_w < \lambda_u < c_p - c_r$	$\alpha \lambda_u + h_u$
						$(\lambda_s = \lambda_u + c_r)$			
(2)	d	0	0	0	u	c_p	0	$c_p - c_r <$	$\alpha \lambda_u + h_u$
	$d - \bar{r}$		0		$u - \bar{r}$			$-c_w < \lambda_u < c_p - c_r$	$\alpha \lambda_u + h_u$
Subcases 2(1) and 2(3) generalized: $p^* = \max\{0, d - \bar{r}\}, r^* = \min\{d, \bar{r}\}, w^* = 0$									
Case 3: $y_s > 0, y_u = 0$									
(1)	0	0	u	-d	0	$< c_r - c_w$	$\alpha\lambda_s + h_s$	$-c_w$	0
$(2) \ (\bar{r} < u)$	0	\bar{r}	$u - \bar{r}$	$\bar{r} - d$	0	$c_r - c_w < \lambda_s < c_p$	$\alpha\lambda_s + h_s$	$-c_w$	0
Case 4: $y_s = 0, y_u = 0$									
(1) $(d < u, d < \bar{r})$	0	d	u - d	0	0	$c_r - c_w$	0	$-c_w$	0
(2) $(u < d, u < \bar{r})$					0	c_p	0	$c_p - c_r$	0
$(3) \ (\bar{r} < u, \bar{r} < d)$	$d - \bar{r}$	\bar{r}	$u - \bar{r}$	0	0	c_p	0	$-c_w$	0
Case 4 generalized: $p^* = \max\{d - r^*, 0\}, r^* = \min\{u, d, \bar{r}\}, w^* = \max\{u - r^*, 0\}$									

 Table 3.3. Main results of optimal cases when considering a remanufacturing constraint.

Optimal Transitions Between Cases and Subcases

Because of the additional subcases, there is a substantial increase of the number of possible case transitions. Therefore, cases are excluded that never occur in an optimal solution (especially when neglecting initial stocks) before describing relevant transitions. Similar to the case of a manufacturing constraint, Subcases 1(1) and 3(1) can only be present at the beginning of the planning period. This is because from the optimal serviceables state developments inside Case 1(1) and Case 3(1) intervals ($\dot{y}_s < 0$) it follows that any transition from a case where $y_s = 0$ to one of these two cases is forbidden. Moreover, because of co-state continuity if the respective inventory is non-empty, there is no possibility for transitions from Case 3(2) to Case 1(1)/3(1) or from Case 1(2) to Case 1(1), because a downward jump in λ_s would be required. Thus, the occurrence of both cases requires positive initial inventory conditions, leading to similar effects as discussed for the basic model.

In any interval where Case 2(2) applies, the recoverables stock must increase $(\dot{y}_u > 0)$. Therefore, a switch to another case where $y_u = 0$ is excluded. Since the value of a recoverables item exceeds the difference of unit production and remanufacturing cost rate $(\lambda_u > c_p - c_r)$, a transition to another case where $y_u > 0$ (Cases 1(2), 2(1), and 2(3)) cannot take place. There the co-state requires a lower value and a downward jump in the co-state is excluded when having positive recoverables inventory. Case 2(2) will therefore never be left contradicting final condition $y_u(T) = 0$. The following corollary collects results regarding the exclusion of cases.

Corollary 3.10 (Cases that do not occur in an optimal solution).

It is never optimal to have intervals of Cases 1(1), 3(1), and 2(2) in a solution to a problem with zero initial and terminal conditions.

Following our usual procedure, Proposition 3.13 collects all situations, under which one or both co-states are allowed to jump.

Proposition 3.13 (Continuity of λ_s and λ_u).

 λ_s and λ_u are continuous, i.e. jump parameters η_s and η_u vanish everywhere, except at time points θ where one of the following conditions hold:

(i) λ_s jumps if $d(\theta) = \bar{r}(\theta)$ and $d(\theta) < \dot{\bar{r}}(\theta)$ hold in situations where $y_s = 0$, allowing for the following transitions:

 $\begin{array}{l} 2(3) \rightarrow 2(1), \ 4(3) \rightarrow 4(1), \ 4(3) \rightarrow 2(1), \ 1(2) \rightarrow 2(1), \ 3(2) \rightarrow 4(1), \\ 2(3) \rightarrow 1(2), \ 4(3) \rightarrow 1(2), \ and \ 4(3) \rightarrow 3(2). \end{array}$

- (ii) λ_s and λ_u jump if $u(\theta) = d(\theta)$ and $\dot{d}(\theta) < \dot{u}(\theta)$ hold in situations where $y_s = y_u = 0$, allowing for the following transitions: $4(2) \rightarrow 4(1), 2(1) \rightarrow 4(1), \text{ and } 4(2) \rightarrow 2(1).$
- (iii) λ_u jumps if $u(\theta) = \bar{r}(\theta)$ and $\dot{u}(\theta) > \dot{\bar{r}}(\theta)$ hold in situations where $y_u = 0$, allowing for the following transitions: $4(2) \rightarrow 4(3), 1(2) \rightarrow 3(2), 1(2) \rightarrow 4(3), 2(3) \rightarrow 4(3), and 4(2) \rightarrow 2(3).$

The following Corollaries 3.11–3.16 state possible case and subcase transitions and are derived by using the same methods described in the previous section.

Corollary 3.11. Within Case 2 Subcase 2(3) can be followed by Subcase 2(1). This automatic and discontinuous transition requires $d = \bar{r}$ and $\dot{d} < \dot{\bar{r}}$.

In Case 2(3) as many items as possible are remanufactured $(r = \bar{r})$. This is only feasible without accumulating a serviceables stock as long as demand exceeds the remanufacturing constraint. As the demand rates falls below the constraint, decisions must change and an automatic transition to Case 2(1) takes place.

Corollary 3.12. Within Case 4 the following transitions are possible: $4(2) \rightarrow 4(1)$. This automatic and discontinuous transition requires $d = u < \bar{r}$ and $\dot{d} < \dot{u}$.

- 4(2) \rightarrow 4(3). This automatic and discontinuous transition requires $u = \bar{r} < d$ and $\dot{u} > \dot{\bar{r}}$.
- $4(3) \rightarrow 4(1)$. This automatic and discontinuous transition requires $d = \bar{r} < u$ and $\dot{d} < \dot{r}$.

Since there is no stock available in Case 4 intervals, the minimum of demand, return rate and constraint determines the remanufacturing rate. Therefore, as another component begins to decide about this rate, also the subcase of Case 4 must change.

Now, possible transitions between different cases are presented.

Corollary 3.13. Starting at a Case 1(2) interval where $y_s > 0$ and $y_u > 0$ the following transitions are possible:

- $1(2) \rightarrow 2(1)$. This automatic and discontinuous transition requires $d = \bar{r}$ and $\dot{d} < \dot{\bar{r}}$.
- $1(2) \rightarrow 2(3)$. This automatic and continuous transition requires $\lambda_s = c_p$ and $\bar{r} < d$.
- $1(2) \rightarrow 3(2)$. This automatic and discontinuous transition requires $u = \bar{r}$ and $\dot{u} > \dot{\bar{r}}$.
- 1(2) \rightarrow 4(3). This automatic and discontinuous transition requires $\lambda_s = c_p$, $u = \bar{r} < d \text{ and } \dot{u} > \dot{\bar{r}}.$

In Case 1(2) remanufacturing takes place at its upper bound. The serviceables inventory decreases if demand exceeds the constraint and recoverables inventory depletes if more recoverables are remanufactured than currently returned. Depending on the development of demand, return, and constraint rates, one of both inventories depletes first, and a switch to Case 2 or Case 3 occurs. As a special case, a transition to Case 4(3) takes place if both inventories are emptied simultaneously.

Corollary 3.14. Starting at a Case 2 interval where $y_s = 0$ and $y_u > 0$ the following transitions are possible:

- $2(1) \rightarrow 1(2)$. This constraint-forced and continuous transition requires $d < \bar{r}$.
- 2(1) \rightarrow 4(1). This automatic and discontinuous transition requires $d = u < \bar{r}, \ \dot{d} \leq \dot{u}$
- 2(1) \rightarrow 4(2). This automatic and continuous transitions requires $\lambda_s = c_p$, $\lambda_u = c_p - c_r$, and $u < d < \bar{r}$.
- $2(3) \rightarrow 1(2)$. This constraint-forced and discontinuous transition requires $d = \bar{r}, \, \dot{d} < \dot{\bar{r}}.$
- $2(3) \rightarrow 4(2)$. This automatic and continuous transition requires $\lambda_u = c_p c_r$ and $u < \bar{r} < d$.
- $2(3) \rightarrow 4(3)$. This automatic and discontinuous transition requires $u = \bar{r} < d$ and $\dot{u} > \dot{\bar{r}}$.

Similar to the model with a limited production rate, a Case 2 interval may terminate into Case 1(2), requiring demand not to exceed the constraint ($d \leq \bar{r}$). A switch to Case 4 either takes place under conditions known from the basic model (2(1) \rightarrow 4(1)/4(2), $u \leq d < \bar{r}$), or it occurs starting from a Case 2(3) interval, which necessitates $u \leq \bar{r} < d$.

Corollary 3.15. Starting at a Case 3(2) interval where $y_s > 0$ and $y_u = 0$ the following transitions are possible:

- $3(2) \rightarrow 1(2)$. This automatic and continuous transition requires $u > \bar{r}$.
- $3(2) \rightarrow 4(1)$. This automatic and discontinuous transition requires $d = \bar{r} < u$ and $\dot{d} < \dot{\bar{r}}$.
- $3(2) \rightarrow 4(3)$. This automatic and continuous transition requires $\lambda_s = c_p$, $\bar{r} < u$, and $\bar{r} < d$.

As before, after completing a Case 3(2) interval, either (1) a constraint-forced switch to Case 1(2) is possible, if the return rate is larger than the remanufacturing constraint, or (2) a depleting serviceables inventory leads to a transition to Case 4 requiring demand to exceed the remanufacturing constraint. **Corollary 3.16.** Starting at a Case 4 interval where $y_s = y_u = 0$ the following transitions are possible:

- $4(1) \rightarrow 1(2)$. This constraint-forced and continuous transition requires $d < \bar{r} < u$.
- $4(1) \rightarrow 2(1)$. This forced and continuous transition requires d < u and $d < \bar{r}$.
- $4(1) \rightarrow 3(2)$. This constraint-forced and continuous transition requires $d < \bar{r} < u$.
- 4(2) \rightarrow 2(1). This forced and discontinuous transition requires $d = u < \bar{r}$, $\dot{d} < \dot{u}$
- $4(2) \rightarrow 2(3)$. This constraint-forced and discontinuous transition requires $u = \bar{r} < d$ and $\dot{u} > \dot{\bar{r}}$.
- $4(3) \rightarrow 1(2)$. This constraint-forced and discontinuous transition requires $d = \bar{r} < u$ and $\dot{d} < \dot{r}$.
- $4(3) \rightarrow 2(1)$. This forced and discontinuous transition requires $d = \bar{r} < u$ and $\dot{d} < \dot{r}$.
- 4(3) \rightarrow 2(3). This constraint-forced and continuous transition requires $d > \bar{r}$ and $u > \bar{r}$.
- $4(3) \rightarrow 3(2)$. This constraint-forced and discontinuous transition requires $d = \bar{r} < u$ and $\dot{d} < \dot{r}$.

Starting in a Case 4 interval, a broad range of forced transitions is possible. These can be organized into four groups. The first is already known from the basic model and uses the anticipation of a change in the relation between demands and returns motive. It terminates in a Case 2(1) interval. A second group, culminating in a Case 2(3) interval uses a related motivation. Here, in a situation with high demand $(d > u \text{ and } d > \bar{r})$, a recoverables stock is collected in a period where $u > \bar{r}$ in anticipation of a return rate falling below the constraint. In the third group, where a switch to a Case 3(2) interval takes place, situations with a high return rate $(d < u, u > \bar{r})$ require to use a serviceables inventory in anticipation of a demand increasing above the constraint. A combination of the last two motivations leads to a transition to Case 1(2), necessitating that both collecting returns and building serviceables stock can be done simultaneously. This requires $d \leq \bar{p} < u$ at switching time.

All possible transitions are depicted in Figure 3.14.

3.3.3 Pure Effects of a Remanufacturing Constraint

When considering a remanufacturing constraint, the current capacity of the product recovery system always suffices to immediately fill demand. Therefore, there is no such inherent reason that requires us to build up stock in order to obtain feasibility as it has been when assuming a limited manufacturing rate. Production in advance never takes place as can be seen from optimal decisions in cases where the serviceables inventory is positive. On the other side, the recovery cost advantage suggests to efficiently use available capacity and thus there are situations where it might be preferable to keep stock to 'smooth'



Fig. 3.14. Optimal case transitions when considering a remanufacturing constraint

its usage. The corresponding *constraint-forced* transitions are discussed in the following. As a result, additional motivations for keeping stock are developed.

Since we are concentrating an pure effects of a limited remanufacturing rate, and in order to reduce the potential complexity of the solution, we restrict ourselves to two scenarios with respect to the relation of demand and return rates. Thus not all transitions present in Figure 3.14 are considered. First, pure effects of a remanufacturing constraint are discussed by assuming a situation with *high demand*, i.e. the return rate is smaller than the demand rate during the whole planning period. Afterwards, we deal with a situation where the opposite holds (*low demand*). In doing so, we neglect situations where the anticipation motive leads to keeping stock. Some of the results are applied in a model for product recovery strategy and presented in Section 5.5.

Pure Effects of a Remanufacturing Constraint when u < d

In a planning situation where demand exceeds the return rate, Case 4(1) plays no role. Exclusion of all switches between cases that require $d \leq u$ leaves transitions as depicted in Figure 3.15.

Here, available returns can immediately be used to fill demand as long as these do not exceed the remanufacturing capacity (Case 4(2)). If they do, all returns exceeding the constraint cannot immediately be used, and are therefore called *bottleneck returns*. An optimal solution must answer the question on how to deal with these items. For simplifying the discussion, let us assume there exists a single interval where returns do not exceed maximal remanufacturing rate amid of two adjacent *bottleneck intervals* during the considered planning period. This situation is sufficient to explain all of the relevant effects. Let $\theta_{x,b}^u$ represent the end of the preceding and $\theta_{e,b}^u$ the begin of the subsequent bottleneck interval with



Fig. 3.15. Optimal case transitions when considering a remanufacturing constraint assuming high demand

$$\begin{aligned} \theta^{u}_{x,b} &: u(\theta^{u}_{x,b}) = \bar{r}(\theta^{u}_{x,b}) \text{ and } \dot{u}(\theta^{u}_{x,b}) < \dot{\bar{r}}(\theta^{u}_{x,b}), \\ \theta^{u}_{e,b} &: u(\theta^{u}_{e,b}) = \bar{r}(\theta^{u}_{e,b})) \text{ and } \dot{u}(\theta^{u}_{e,b}) > \dot{\bar{r}}(\theta^{u}_{e,b}). \end{aligned}$$

At the end of the preceding bottleneck period it might be preferable to keep returns and use them at a time where the return rate has fallen below the constraint. There, the stored returns can be used to maximize remanufacturing capacity usage and replace an otherwise necessary production of new items. The following corollary results from the fact that Case 4(3) can only terminate into a Case 2(3) interval.

Corollary 3.17 (Location of an interval with positive recoverables inventory when demand is high).

Let θ denote a time point where $u(\theta) = \bar{r}(\theta) < d(\theta)$. If returns cross the constraint from above, i.e. $\dot{u}(\theta) < \dot{\bar{r}}(\theta)$, then it is always optimal to have a positive recoverables inventory at (and around) time θ ($y_u(\theta) > 0$).

Let $\theta_{e,u}$ denote the start of a bottleneck *collection period*, i.e. the time of a transition from Case 4(3) to Case 2(3) at which accumulation of returns starts. Then, at $\theta_{e,u}$ the return rate must not be smaller than the remanufacturing constraint $(u(\theta_{e,u}) \geq \bar{r}(\theta_{e,u}))$ in order to start building up stock. The end of a bottleneck interval $\theta_{x,b}^{u}$ (which satisfies Corollary 3.17) separates the collection from a *consumption period* where stored recoverables are used up, being completed at time $\theta_{x,u}$.

Figure 3.16 shows a situation where demand exceeds both the return rate as well as the remanufacturing constraint. There exists a point satisfying Corollary 3.17 which therefore separates a bottleneck return collection period from a return consumption period. Before $\theta_{e,u}$ part of the returns, i.e. those exceeding the constraint, are disposed of and after $\theta_{x,u}$ the production rate gets higher than would be required when remanufacturing with highest



Fig. 3.16. An interval where returns are kept during a bottleneck collection interval for later use within a consumption period with maximal length.

possible rate. Therefore, a trade-off is balanced which is similar to that one being known from the basic model for collecting a returned item and later using it for replacing production by remanufacturing, instead of disposing of it. A corresponding marginal criterion leads to a maximal holding time condition for recoverables as presented in the following proposition.

Proposition 3.14 (Maximal Length of an interval where $y_u > 0$). The maximal length of an interval $I = (\theta_{e,u}, \theta_{x,u})$ where $y_u > 0$ holds is given by

$$\theta_{x,u} - \theta_{e,u} \le \tau_u \tag{3.30}$$

where τ_u is determined in Proposition 2.7.

A bottleneck collection and consumption period having maximal length is exemplified in Figure 3.16 showing a transition sequence $4(3) \rightarrow 2(3) \rightarrow 4(2)$. Here also optimal co-state movements can be seen. Since production takes place in all cases, the serviceables value is c_p . The value of a returned item rises during the Case 2(3) interval starting with $-c_w$ at $\theta_{e,u}$ and reaching $c_p - c_r$ at $\theta_{x,u}$.

Intervals with positive recoverables inventory not having maximal length require intersections of return rate and constraint at either $\theta_{e,u}$ or $\theta_{x,u}$. In the first case, the collection period cannot start earlier, since there are not sufficient returns that exceed the remanufacturing capacity. A corresponding case sequence would be $4(2) \rightarrow 2(3) \rightarrow 4(2)$. Secondly, the consumption period must end, because there is no remaining excess capacity available to remanufacture stored returns resulting in a sequence $4(3) \rightarrow 2(3) \rightarrow 4(3)$.

Another interesting question would be what happens if demand intersects the constraint. By definition this is only possible inside a consumption period because otherwise the assumption that demands exceed the return rate would be contradicted. As demand falls below the constraint, it is no longer necessary to remanufacture with maximum rate and a transition from Case 2(3) to 2(1) takes place at a time θ with $d = \bar{r}$ and $\dot{d} < \dot{\bar{r}}$. An example where the joint interval (where $y_u > 0$) has maximal length is depicted in Figure 3.17. It shows a sequence $4(3) \rightarrow 2(3) \rightarrow 2(1) \rightarrow 4(2)$. Since production is not



Fig. 3.17. A consumption period showing a Case 2(1) interval

required inside Case 2(1) intervals, the value of serviceables drops down to the sum of the value of recoverables and the remanufacturing cost rate at the transition time. The same reasons as explained above allow also in this case for Case 2 intervals of less than maximal length corresponding to sequences $4(2) \rightarrow 2(3) \rightarrow 2(1) \rightarrow 4(2)$ and $4(3) \rightarrow 2(3) \rightarrow 2(1) \rightarrow 4(3)$.

If inside a consumption period demand crosses the constraint from below, it is preferable to remanufacture with maximum rate before the intersection time and to keep serviceables in stock, since it now allows to fill additional demand from remanufactured and kept items (see Figure 3.18). The following corollary results from the impossibility of a transition from Case 2(1) to Case 2(3).

Corollary 3.18 (Location of an interval with positive serviceables inventory when demand is high).

Let θ denote a time point where $u(\theta) < d(\theta) = \bar{r}(\theta)$ and $y_u(\theta) > 0$. If demand crosses the constraint from below, i.e. $\dot{d}(\theta) > \dot{r}(\theta)$, then it is always optimal to have a positive serviceables inventory at (and around) time θ ($y_s(\theta) > 0$).

There might be several time points meeting Corollary 3.18 within a consumption period. Therefore, it is assumed that there are n > 0 intervals $J^i = (\theta^i_{e,s}, \theta^i_{x,s}), i = 1, 2, ..., n$ where it is optimal to keep serviceables $(y_s > 0)$.



Fig. 3.18. A consumption period showing a Case 1(2) interval.

For each one of these, the following inventory conditions must hold (presented without proof).

Proposition 3.15 (Inventory Conditions of intervals with positive serviceables inventory).

Let $J^i = (\theta^i_{e,s}, \theta^i_{x,s})$ be an open time interval where $y_s > 0$ and $y_s(\theta^i_{e,s}) = y_s(\theta^i_{x,s}) = 0$. Then,

(i) cumulative remanufacturing equals cumulative demand over the whole interval

$$\int_{\theta_{e,s}^{i}}^{\theta_{x,s}^{i}} (\bar{r}(t) - d(t)) dt = 0, \qquad (3.31)$$

(ii) at any point $\theta \in J^i$, cumulative remanufacturing must be larger than cumulative demand

$$\int_{\theta_{e,s}^i}^{\theta} (\bar{r}(t) - d(t))dt > 0.$$

$$(3.32)$$

Inside intervals with serviceables in stock, more demand is satisfied from remanufacturing than inside Case 2 and thus, additional returns are used adding up to $\int_{\theta_{e,s}^i}^{\theta_{x,s}^i} (\max\{\bar{r}(t) - d(t), 0\}) dt$. This result can be used in order to derive inventory conditions for the recoverables inventory (presented without proof and omitting the non-negativity condition).

Proposition 3.16 (Inventory Condition for the joint collection and consumption interval).

Let $I = (\theta_{e,u}, \theta_{x,u})$ be an open time interval where $y_u > 0$ with $y_s(\theta_{e,u}) = y_u(\theta_{e,u}) = y_u(\theta_{x,b}) = 0$ and $J^i = (\theta^i_{e,s}, \theta^i_{x,s})$, i = 1, 2, ..., n be open time intervals where $y_s > 0$, $y_u > 0$ and for which Proposition 3.15 applies. Then, cumulative stored returns during bottleneck periods must equal consumed returns

$$\int_{\theta_{e,u}}^{\theta_{x,u}} (u(t) - \min\{d(t), \bar{r}(t)\}) dt - \sum_{i=1}^{n} \int_{\theta_{e,s}^{i}}^{\theta_{x,s}^{i}} (\max\{\bar{r}(t) - d(t), 0\}) dt = 0.$$
(3.33)

As an implication from consuming more recoverables, also the consumption period reduces when increasing the length of an interval with a positive serviceables stock. This lowers costs as long as the induced serviceables holding costs are accompanied by a higher reduction of recoverables holding costs. This trade-off yields the following proposition.

Proposition 3.17 (Maximal Length of a Case 1(2) interval).

The maximal length of a Case 1(2) interval $J^i = (\theta^i_{e,s}, \theta^i_{x,s})$ is time dependent and it is given by

$$\theta_{x,s}^{i} - \theta_{e,s}^{i} \le \frac{1}{\alpha} \ln \left(\frac{\alpha c_{p} + h_{s}}{\alpha \left(\lambda_{u}(\theta_{e,s}^{i}) + c_{r} \right) \right) + h_{s}} \right).$$
(3.34)

In short term problems when assuming $\alpha = 0$, condition (3.34) changes to

$$\theta_{x,s}^i - \theta_{e,s}^i \le \frac{c_p - \lambda_u(\theta_{e,s}^i) - c_r}{h_s}.$$
(3.35)

Marginal criterion (3.34) (respectively (3.35)) can be interpreted as follows. Assume there are two possibilities to use a collected returned unit available at $\theta_{e,s}^i$. This can be either (1) keeping it until $\theta_{x,u}$ where it is remanufactured to fill demand at that point (increase the consumption period) or (2) remanufacturing it immediately to stock at $\theta_{e,s}^i$ for satisfying demand at $\theta_{x,s}^i$ and thus, increasing the length of the interval where $y_s > 0$. In an optimal solution, one needs to be indifferent between both options except for situations explained later. As another implication of Proposition 3.17, the maximal length of an interval with positive serviceables inventory must decrease during a consumption period, because the later $\theta_{e,s}^i$ the higher the recoverables co-state and thus, the right hand side of (3.34) must fall. This is because remanufacturing cost advantage is balanced against total holding costs for a marginal item. The later a stored return is remanufactured and transferred to serviceables stock the higher are already accumulated recoverables holding costs and thus less serviceables holding cost are allowed to be added.

Since the serviceables co-state increases faster than the recoverables value inside a Case 1(2) value, the serviceables inventory must be emptied before



Fig. 3.19. A consumption period terminating into a Case 3(2) interval.

depleting the recoverables inventory except for one reason. Namely that if the time period, where the recoverables inventory is positive is limited because the return rate rises above the remanufacturing constraint at the end of this interval $\theta_{x,u}$, then the consumption period does not reach maximal length. There, it might be preferable to hold serviceables even longer than until $\theta_{x,u}$. This allows for sequences (at the end of a consumption period) $1(2) \rightarrow 4(3)$, if both inventories are depleted simultaneously and $1(2) \rightarrow 3(2) \rightarrow 4(3)$ if recoverables inventory is emptied first. An example is depicted in Figure 3.19.

Pure Effects of a Remanufacturing Constraint when u > d

Planning situations under which the return rate always exceeds the demand rate never show Case 4(2) intervals, and excluding all transitions requiring $d \ge u$ leaves switches as depicted in Figure 3.20. In what follows, we concentrate on a main effect of a remanufacturing constraint when demand is low by additionally assuming $u \ge \overline{r}$. Thus, there are always sufficient returns available for remanufacturing and stockkeeping is not performed under use of a recoverables inventory. Under this assumption therefore only sequences

Case
$$4 \rightarrow$$
 Case $3(2) \rightarrow$ Case 4

occur in an optimal solution. Further results could be obtained when allowing for more complex interactions as has been accomplished in the case of high demand.



Fig. 3.20. Optimal case transitions when considering a remanufacturing constraint assuming low demand.

As long as demand does not exceed the remanufacturing constraint, it is satisfied from current remanufacturing, and excess returns are disposed of (Case 4(1)). An optimal solution must answer the question from which source to satisfy demand that exceeds the remanufacturing capacity, which is therefore called *bottleneck demand*. Analogously to the previously considered situation, our discussion is restricted to a single interval where $d < \bar{r}$. Let $\theta_{x,b}^d$ be the end of the preceding and $\theta_{e,b}^d$ the beginning of a subsequent *bottleneck interval* with

$$\begin{aligned} \theta^d_{x,b} &: d(\theta^d_{x,b}) = \bar{r}(\theta^d_{x,b}) \text{ and } d(\theta^d_{x,b}) < \dot{\bar{r}}(\theta^d_{x,b}), \\ \theta^d_{e,b} &: d(\theta^d_{e,b}) = \bar{r}(\theta^d_{e,b})) \text{ and } \dot{d}(\theta^d_{e,b}) > \dot{\bar{r}}(\theta^d_{e,b}). \end{aligned}$$

Before $\theta_{e,b}^d$ it is favorable to remanufacture as many returns as possible and to build up a serviceables stock. These items can later be used to replace otherwise necessary production. Since Case 4(3) can only be preceded by a Case 3(2) interval, Corollary 3.19 follows.

Corollary 3.19 (Location of an interval with positive serviceables inventory when demand is low).

Let θ denote a time point where $d(\theta) = \bar{r}(\theta) < u(\theta)$. If demand crosses the constraint from below, i.e. $\dot{d}(\theta) > \dot{\bar{r}}(\theta)$, then it is always optimal to have a positive serviceables inventory at (and around) time θ ($y_s(\theta) > 0$).

Let $\theta_{x,s}$ denote the end of a serviceables consumption period, i.e. the time of a transition from Case 3(2) to Case 4. At $\theta_{x,s}$, demand must not be smaller than the remanufacturing constraint in order to deplete the serviceables inventory. The beginning of a bottleneck interval $\theta_{e,b}^d$ separates a serviceables collection from the respective consumption period. The first is assumed to



Fig. 3.21. An interval where returns are remanufactured with maximal rate during a serviceables accumulation interval for later use within a demand bottleneck with maximal length.

start at $\theta_{e,s}$. For the interval $(\theta_{e,s}, \theta_{x,s})$ an inventory condition as stated in Proposition 3.15 must hold.

Figure 3.21 shows a situation where the return rate exceeds both demand and remanufacturing constraint. A point satisfying Corollary 3.19 separates a serviceables accumulation period from a consumption period. Before $\theta_{e,s}$, all returns exceeding current demand are disposed of. After $\theta_{x,s}$, the part of demand which is higher than the remanufacturing capacity is served from production. Therefore, a marginal criterion applies for remanufacturing a returned item to stock and later using it for serving demand from inventory and replacing production instead of disposing of it. A maximal holding time condition for serviceables results as presented in the following proposition.

Proposition 3.18 (Maximal Length of an interval where $y_s > 0$).

The maximal length of an interval $I = (\theta_{e,s}, \theta_{x,s})$ where $y_s > 0$ holds is given by

$$\theta_{x,s} - \theta_{e,s} \le \tau_s = \frac{1}{\alpha} \ln \left(\frac{\alpha c_p + h_s}{\alpha (c_r - c_w) + h_s} \right).$$
(3.36)

When assuming a short planning period and setting α to zero, (3.36) simplifies to

$$\theta_{x,s} - \theta_{e,s} \le \tau_s = \frac{c_p + c_w - c_r}{h_s} \tag{3.37}$$

Figure 3.21 illustrates an interval where $y_s > 0$ with maximal length showing a sequence $4(1) \rightarrow 3(2) \rightarrow 4(3)$ as well as optimal co-state movements. While the serviceables value steadily increases, the value of returns is given by $-c_w$ because in all cases returns must partly be disposed of. There are two reasons for not having a Case 3(2) interval with maximal length. Either there is no remaining capacity available at $\theta_{e,s}$ and thus, the serviceables accumulation period cannot be extended, or consumption of serviceables ends because there exists no further bottleneck demand at $\theta_{x,s}$. The first case leads to a sequence $4(3) \rightarrow 3(2) \rightarrow 4(3)$ and the second to $4(1) \rightarrow 3(2) \rightarrow 4(1)$.

3.3.4 Managerial Insights

Although the current capacity of the system always suffices to satisfy demand without using inventories, a limitation of the cheaper mode, namely remanufacturing, motivates stock-keeping in order to improve the usage of available capacity. Even when restricting to pure effects, i.e. neglecting interactions with the anticipation stock motive, this motivation allows for a number of different situations where stock-keeping takes place, but all of these jointly require that there are periods where the remanufacturing rate is restricted through another circumstance which can be either insufficient returns or demand.

In situations with high demand (d > u), returns are kept in intervals where they cannot be immediately used for later remanufacturing if an interval with returns exceeding the constraint is followed by another where the opposite holds. The maximum length of these intervals is given by the Maximal Holding Time τ_u which is already known from the basic model. If during a return consumption period demand falls below the remanufacturing capacity it might be preferable to additionally build up a serviceables inventory through remanufacturing with maximum rate. This stock is used up in a period where demand rises above the constraint. A maximal length criterion holds which also takes into account the period where returns have been held in the recoverables stock and therefore, the maximal length of an interval with positive serviceables stock decreases in time.

If the return rate is larger than the demand rate (d < u), primarily a serviceables inventory is used in situations where an interval with demand being lower than the remanufacturing capacity is followed by another where demand exceeds the constraint. As before, a maximal length of the interval can be given.

3.4 Proofs

Proof (Proof of Proposition 3.1). From (2.23) and (2.24) it follows that $k_1 = k_2 = 0$. Thus, (2.21) and (2.22) imply

$$\dot{\lambda}_s = \alpha \lambda_s + h_s > 0 \text{ and } \dot{\lambda}_u = \alpha \lambda_u + h_u > 0.$$
 (3.38)

The proofs for r = 0 and w = 0 are the same as in Proposition 2.2. Thus,

$$\lambda_s < \lambda_u + c_r \text{ and } \lambda_u > -c_w$$
 (3.39)

 $0 requires <math>\mu_1 = 0$ in (2.18) and $\mu_4 = 0$ in (3.8) yielding $\lambda_s = c_p$ in (3.7) which contradicts (3.38).

p = 0 requires $\lambda_s < c_p$ and $p = \bar{p}$ necessitates $\lambda_s > c_p$ from (3.6). Together with (3.39), in the latter case it follows $\lambda_u > c_p - c_r$.

Proof (Proof of Proposition 3.2). From (2.24) it follows that $k_2 = 0$. Thus, (2.22) implies

$$\dot{\lambda}_u = \alpha \lambda_u + h_u > 0. \tag{3.40}$$

The proof for w = 0 is the same as in Case 1 (see proof of Proposition 2.2). Thus,

$$\lambda_u > -c_w \tag{3.41}$$

follows from (2.14).

 $0 requires <math>\mu_1 = \mu_4 = 0$ from (2.18) and (3.8) which yields $\lambda_s = c_p$ in (3.7). If simultaneously r > 0 than from (2.19) it follows $\mu_2 = 0$, leading to $\lambda_u = c_p - c_r$ in (2.16). It follows that $\dot{\lambda}_u = 0$ which contradicts (3.40).

In any interval where 0 and <math>r = 0 hold, the definition of Case 2 $(\dot{y}_s = 0)$ requires $p = d < \bar{p}$ and $\dot{y}_u = u \ge 0$. From (3.6) and (2.13) it follows $\lambda_u > c_p - c_r$.

r > 0 from (2.13) necessitates $\lambda_s = \lambda_u + c_r$ and $\dot{\lambda}_s = \dot{\lambda}_u$. p = 0 requires r = din order to fill the demand and $\lambda_s < c_p$ from (3.6) from which $\lambda_u < c_p - c_r$ follows. $p = \bar{p}$ requires (using the definition of the case) $d > \bar{p}$ and thus $r = d - \bar{p}$ as well as $\lambda_s > c_p$ from (3.6) from which $\lambda_u > c_p - c_r$ follows.

Proof (Proof of Proposition 3.3). This proof proceeds in the same way as the proof to Proposition 2.6. (i) time points inside intervals where $y_s = 0$ or $y_u = 0$ holds are examined and (ii) we deal with entry and exit points of such intervals.

(i) First we look for circumstances under which the matrix

$$\begin{pmatrix} 1 & 0 & 0 & p & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & r & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & w & 0 & 0 & 0 \\ -1 & 0 & 0 & 0 & 0 & \bar{p} - p & 0 & 0 \\ 1 & 1 & 0 & 0 & 0 & 0 & y_s & 0 \\ 0 & -1 & -1 & 0 & 0 & 0 & 0 & y_u \end{pmatrix} \begin{bmatrix} I \\ III \\ III \\ V \\ V \\ VI \end{bmatrix}$$
(3.42)

has not full rank of six. Reconsidering $\bar{p} > 0$, d > 0, and u > 0, this is not the case in the following five situations:

- (*i-i*) If p = 0, r = 0 and $y_s = 0$ then I + II = V. This requires d = 0 and can therefore be excluded.
- (*i*-*ii*) If p = p̄, r = 0 and y_s = 0 then −*IV* + *II* = *V*. This requires d = p̄ and a jump in λ_s may occur
 (a) inside Case 2 intervals when switching from Subcase 2(3) to 2(2). From the definition of both cases it follows d̄ < p̄. This completes the proof of Proposition 3.3(i).
 (b) inside Case 4 intervals. But here r = min{u, d} > 0 and therefore the

(b) inside Case 4 intervals. But here $r = \min\{u, d\} > 0$ and therefore the co-state is continuous.

(c) at entry and exit times of Case 4 intervals when switching from/to Case 2. Since only downward jumps are allowed for λ_s and because d > 0, this would allow for a transition from Case 2(3) to Case 4(2). Additionally, a jump in λ_u is required which can only occur under conditions put forth in part (*ii*) of this proof.

- (*i-iii*) If r = 0, w = 0 and $y_u = 0$ then II + III = -VI. This requires u = 0 and can therefore be excluded.
- (*i*-*iv*) If p = 0, w = 0, y_s = y_u = 0 then I − III = V + VI. This situation occurs when demand equals returns (d = u) and is already known from the basic model. See proof of Proposition 2.6.
 (a) Inside Case 4 intervals a jump in both λ_s and λ_u occurs when switching from Case 4(2) to Case 4(1). The definition of both cases requires d < u. This completes the proof of Proposition 3.3(ii).
 (b) At entry and exit times of Case 4 intervals in the case of a transition from/to Case 2, a jump in λ_s may occur when switching (b1) from Case 4(2) to Case 2(1), (b2) from Case 2(1) to Case 4(1) as well as (b3) from Case 2(3) to Case 4(2). As before, all transitions (b1-3) also require a jump in λ_u. Therefore, (b3) would require d − p̄ = u < d (necessary for a non-tangential entry into the y_u = 0 interval) which contradicts d = u because of p̄ > 0.
- (i-v) If p = p̄, w = 0, y_s = y_u = 0 then -IV III = V + VI. This requires d = u + p̄ > u and may appear
 (a) inside Case 4 intervals. Since this inequality shows the presence of Subcase 4(2), the co-states are fixed and therefore continuous.
 (b) at entry and exit times of Case 4 intervals when switching from/to Case 2, λ_s may jump when switching from Case 2(3) to Case 4(2). As before, this transition also requires for a jump in λ_u.

(ii) Let θ_s^1 be the entry time of an interval where $y_s = 0$ holds, i.e. the begin of a Case 2 or Case 4 interval. Then, λ_s is continuous at this time point if y_s enters this interval in a non-tangential way, i.e. $\dot{y}_s = p + r - d$ jumps. This happens if the sum of the controls p and r jumps at θ_s^1 . An entry point θ_s^1 can only be present at a switch (a) from Case 1 to Case 2 or (b) from Case 3 to Case 4 (other transitions are not possible when applying the optimal decisions in Case 1 and Case 3 intervals). In the first case (a), p + r does not jump if $d = \bar{p}$ holds at the end of Subcase 1(2) interval, where $\lambda_u(\theta_s^1) > c_p - c_r$. Since $y_u > 0$ holds for both intervals, λ_u must be continuous, thus only transitions to Cases 2(2) or (3) are possible. Since $\lambda_s < \lambda_u + c_r$ holds inside Subcase 1(2), $\lambda_s = \lambda_u + c_r$ inside Subcase 2(3), and λ_u is continuous during this transition $(y_u > 0), \lambda_s$ must be continuous as well when switching from Subcase 1(2) to 2(3). In the second case (b), p + r = 0 holds at the end of a Case 3 interval, thus the Case 4 interval is entered non-tangentially, because by definition of this case p + r = d > 0 must hold. Thus the only not continuous entry into an interval where $y_s = 0$ holds (i.e. where λ_s jumps) is a transition from Case 1(2) to 2(2) if $d = \bar{p}$. In order to leave Case 1(2), $d > \bar{p}$ is required at the end of such an interval. This leads to $\dot{d} < \dot{\bar{p}}$ at switching time. This completes the first part of the proof of Proposition 3.3(iii).

An exit time θ_s^2 is present at a transition from Case 2 to Case 1 and from Case 4 to Case 1 or Case 3. A switch from Case 4 to Case 1(1) or Case 3 is not possible because there no serviceables stock can be accumulated by applying optimal case decisions. Since upward jumps in the co-states are not allowed, a transition from Case 4 to Case 1(2) must take place continuously. The same holds for transitions from Case 2 (1) to Case 1(2). Proceeding similarly as for the entry time shows that only the transition from Case 2(3) to Case 1(2) allows for a jump in λ_s , if $d = \bar{p}$. In order to enter Case 1(2), $d > \bar{p}$ is required at the begin of such an interval. Again, this leads to $\dot{d} < \dot{p}$ at switching time. This completes the second part of the proof of Proposition 3.3(iii).

Let θ_u^1 be the entry time and θ_u^2 the exit time of an interval, where $y_u = 0$ holds (Case 3 or Case 4). Then, λ_u is continuous at these time points if y_u enters or leaves this interval in a non-tangential way, i.e. $\dot{y}_u = u - r - w$ jumps. That occurs if r + w jumps at (c) θ_u^1 or (d) θ_u^2 , respectively. (c) Case 3 can not be reached, neither from a Case 1 nor a Case 2 interval by following optimal policies. An entry point of Case 4 is only present after a Case 2 interval. Thus, λ_u will always be continuous except for transitions from Case 2(1) to Case 4(1) where d = u and $d < \dot{u}$, and for transitions from Case 2(3) to Case 4(2) where $d = u + \bar{p}$ and $d < \dot{u} + \bar{p}$. In both cases, also λ_s may jump, because the system is inside a boundary situation w.r.t. the serviceables inventory and situations as put forth in (i-iv) and (i-v), respectively, apply. This completes the proof of the first part of (iv) as well as part (v) of Proposition 3.3. In case (d), all transitions starting at Case 3 or Case 4 intervals can be excluded or must proceed continuously in λ_u with the exception of a transition from Case 4(2) to Case 2(1) requiring u = d. From (*i*-*iv*), also λ_s jumps under such circumstance. This completes the second part of the proof of Proposition 3.3(iv).

Proof (Proof of Proposition 3.7). From Proposition 3.1 we get a minimal value for λ_s of $\lambda_s^{\min} = c_p$. For Case 1(2) we can give an upper bound for $\lambda_s^{\max} = \lambda_u + c_r$. Given $\lambda_u(\theta_{e,1(2)}^i)$, the recoverables co-state is

$$\lambda_u(t) = \left(\lambda_u(\theta_{e,1(2)}^i) + \frac{h_u}{\alpha}\right) e^{\alpha(t-\theta_{e,1(2)}^i)} - \frac{h_u}{\alpha}.$$

Together with continuity of λ_s and λ_u inside the Case 1(2) interval and the solution of co-state development (3.9) yields

$$(c_p + \frac{h_s}{\alpha}) e^{\alpha \left(\theta_{x,1(2)}^i - \theta_{e,1(2)}^i\right)} - \frac{h_s}{\alpha} \le \lambda_s(\theta_{x,1(2)}^i) \text{ and} \lambda_s(\theta_{x,1(2)}^i) \le \left(\lambda_u(\theta_{e,1(2)}^i) + \frac{h_u}{\alpha}\right) e^{\alpha \left(\theta_{x,1(2)}^i - \theta_{e,1(2)}^i\right)} - \frac{h_u}{\alpha} + c_r$$

Solving the combined inequality for $\theta^i_{x,1(2)} - \theta^i_{e,1(2)}$ finally yields

90 3 On the Effects of Capacity Constraints in Product Recovery

$$\theta_{x,1(2)}^{i} - \theta_{e,1(2)}^{i} \leq \frac{1}{\alpha} \ln \left(\frac{\alpha c_r + h_s - h_u}{\alpha \left(c_p - \lambda_u(\theta_{e,1(2)}^{i}) \right) + h_s - h_u} \right)$$

Proof (Proof of Proposition 3.9). From (2.23) and (2.24) it follows that $k_1 = k_2 = 0$. Thus, (2.21) and (2.22) imply

$$\dot{\lambda}_s = \alpha \lambda_s + h_s > 0 \text{ and } \dot{\lambda}_u = \alpha \lambda_u + h_u > 0.$$
 (3.43)

The proofs for p = 0 and w = 0 are the same as in Proposition 2.2. Thus,

$$\lambda_s < c_p \text{ and } \lambda_u > -c_w.$$
 (3.44)

 $0 < r < \bar{r}$ requires $\mu_2 = 0$ in (2.19) and $\mu_4 = 0$ in (3.29) yielding $\lambda_s - \lambda_u = c_r$ in (3.28). This requires $\dot{\lambda}_s = \dot{\lambda}_u$ which contradicts (3.43) because $h_s > h_u$. r = 0 requires $\lambda_s - \lambda_u < c_r$ and $r = \bar{r}$ necessitates $\lambda_s - \lambda_u > c_r$ from (3.27). Together with (3.44), in the latter case it follows $\lambda_u < c_p - c_r$ and $\lambda_s > c_r - c_w$.

Proof (Proof of Proposition 3.10). From (2.24) it follows that $k_2 = 0$. Thus, (2.22) implies

$$\dot{\lambda}_u = \alpha \lambda_u + h_u > 0. \tag{3.45}$$

The proof for w = 0 is the same as in Case 1 (see proof of Proposition 2.2). Thus,

$$\lambda_u > -c_w \tag{3.46}$$

from (2.14).

 $0 < r < \bar{r}$ requires $\mu_2 = \mu_4 = 0$ from (2.19) and (3.29) which yields $\lambda_s - \lambda_u = c_r$ in (3.28). If simultaneously p > 0, then from (2.18) it follows $\mu_1 = 0$, leading to $\lambda_s = c_p$ in (2.15) as well as $\lambda_u = c_p - c_r$ in (3.28). It follows $\lambda_u = 0$ which contradicts (3.45).

In any interval where $0 < r < \bar{r}$ and p = 0 hold, the definition of Case 2 $(\dot{y}_s = 0)$ requires $r = d < \bar{r}$ and $\dot{y}_u = u - d$. From (2.12) and (3.27) it follows $\lambda_s = \lambda_u + c_r$, $\lambda_u < c_p - c_r$, and $\dot{\lambda}_s = \dot{\lambda}_u$.

p > 0 from (2.12) necessitates $\lambda_s = c_p$ and $\dot{\lambda}_s = 0$. r = 0 requires p = d in order to fill the demand and $\lambda_s - \lambda_u < c_r$ from (3.27) which also requires $\lambda_u > c_p - c_r$. p > 0 and $r = \bar{r}$ requires (using the definition of the case) $d > \bar{r}$ and thus $p = d - \bar{r}$ as well as $\lambda_u < c_p - c_r$ from (3.27).

Proof (Proof of Proposition 3.11). From (2.23) it follows that $k_1 = 0$. (2.21) implies

$$\dot{\lambda}_s = \alpha \lambda_s + h_s > 0. \tag{3.47}$$

p > 0 requires $\mu_1 = 0$ from (2.18) which yields $\lambda_s = c_p$. It follows that $\dot{\lambda}_s = 0$ which contradicts (3.47). Thus, $p^* = 0$ and

$$\lambda_s < c_p. \tag{3.48}$$

r = 0, w = 0 requires $\dot{y}_u = u$ which is not feasible as long as $y_u = 0$.

 $0 < r < \bar{r}$ requires $\mu_2 = \mu_4 = 0$ in (2.19) and (3.29) which yields $\lambda_s - \lambda_u = c_r$ in (3.28). It follows that $\dot{\lambda}_s = \dot{\lambda}_u$. Applying to (3.47) and (2.22) and solving for k_2 yields $k_2 = \alpha(-c_r) + h_u - h_s$ and $k_2 \ge 0$ in (2.24) contradicts the assumptions that $h_s > h_u$ and $c_r > 0$.

r = 0 and w > 0 necessitates w = u in order to stay in Case 3 ($\dot{y}_u = 0$ has to hold). It further requires $\mu_3 = \mu_4 = 0$ in (2.20) and (3.29), yielding $\lambda_u = -c_w$ in (2.17) and finally $\lambda_s < c_r - c_w$ in (3.28).

 $r = \bar{r}$ and w > 0 requires $w = u - \bar{r}$ as well as $u > \bar{r}$. As before, $\lambda_u = -c_w$ holds and together with $\mu_2 = 0$ from (2.19) this yields $\lambda_s > c_r - c_w$ in (3.28).

Proof (Proof of Proposition 3.12).

From the definition of Case 4 it follows $\dot{y}_s = \dot{y}_u = 0$ which implies p + r = dand r + w = u. When neglecting boundary situations, i.e. where demand or return rate equal zero, or where demand or return rate exactly match the remanufacturing constraint, this already excludes the alternatives (p = 0, r = 0), (r = 0, w = 0), $(p = 0, r = \bar{r})$, and $(r = \bar{r}, w = 0)$.

For $p > 0, 0 \le r < \overline{r}, w > 0$ we find $\lambda_s = c_p, \lambda_u = -c_w$ and $\mu_2 = c_r - c_p - c_w$ which has to be non-negative (2.19). This contradicts the assumption of a positive recovery advantage (2.6).

For $p = 0, 0 < r < \overline{r}, w > 0$ we find r = d to ensure $\dot{y}_s = 0$ and then w = u - d to ensure $\dot{y}_u = 0$. This decision requires $d < \overline{r}$ and d < u. The co-states are given by $\lambda_s = c_r - c_w$ and $\lambda_u = -c_w$.

For $p > 0, 0 < r < \bar{r}, w = 0$ we find r = u to ensure $\dot{y}_u = 0$ and then p = d - u to ensure $\dot{y}_s = 0$. This configuration is only feasible if d > u and $u < \bar{r}$. The co-states are given by $\lambda_s = c_p$ and $\lambda_u = c_p - c_r$.

 $p > 0, r = \overline{r}, w > 0$ requires $p = d - \overline{r}$ and $w = u - \overline{r}$ leading to $d > \overline{r}$ as well as $u > \overline{r}$. The co-states are given by $\lambda_s = c_p$ and $\lambda_u = -c_w$.

Proof (Proof of Proposition 3.13). This proof proceeds in the same way as the proof to Proposition 2.6. (i) time points inside intervals where $y_s = 0$ or $y_u = 0$ holds are examined and (ii) we deal with entry and exit points of such intervals. Thereby, boundary Cases 1(1), 3(1) and 2(2) are neglected.

(i) First we look for circumstances under which the matrix

$$\begin{pmatrix} 1 & 0 & 0 & p & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & r & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & w & 0 & 0 & 0 \\ 0 & -1 & 0 & 0 & 0 & 0 & \bar{r} - r & 0 & 0 \\ 1 & 1 & 0 & 0 & 0 & 0 & y_s & 0 \\ 0 & -1 & -1 & 0 & 0 & 0 & 0 & y_u \end{pmatrix} \begin{bmatrix} I \\ III \\ III \\ IV \\ V \\ VI \end{bmatrix}$$
(3.49)

has not full rank of six. Reconsidering $\bar{r} > 0$, d > 0, u > 0, as well as assumption (3.25), this is not the case in the following five situations:

• (*i-i*) If p = 0, r = 0 and $y_s = 0$ then I + II = V. This requires d = 0 and can therefore be excluded.

- (*i*-*ii*) If p = 0, r = r̄ and y_s = 0 then I IV = V. This requires d = r̄ and a jump in λ_s may occur
 (a) inside Case 2 intervals when switching from Subcase 2(3) to 2(1).
 (b) inside Case 4 intervals when switching from Subcase 4(3) to 4(1).
 (c) at entry and exit times of Case 4 intervals when switching from/to Case 2. Since only downward jumps are allowed for λ_s, this would allow for transitions from Case 2(3) to Case 4(1) as well as from Case 4(3) to Case 2(1). From the definition of Case 2(3) and 4(1), the first transitions would require d = u = r̄ which contradicts (3.25). From the definition of the cases or in order to clear the recoverables inventory, d̄ < r̄ is required in all relevant situations (a-c). This completes the first part of the proof of Proposition 3.13(i).
- (i-iii) If r = 0, w = 0 and $y_u = 0$ then II + III = -VI. This requires u = 0 and can therefore be excluded.
- (i-iv) If p = 0, w = 0, $y_s = y_u = 0$ then I III = V + VI. This situation occurs when demand intersects returns from above $(d = u, \dot{d} < \dot{u})$ and is already known from the basic model. See proof of Proposition 2.6.

(a) Inside Case 4 intervals a jump in both λ_s and λ_u occurs when switching from Case 4(2) to Case 4(1).

(b) At entry and exit times of Case 4 intervals in the case of a transition from/to Case 2, a jump in λ_s may occur when switching (b1) from Case 4(2) to Case 2(1) and (b2) from Case 2(1) to Case 4(1). All transitions (b1-2) also require for a jump in λ_u . In anticipation of part (*ii*) of this proof, from optimal decisions within the respective cases it can be seen that intervals where $y_u = 0$ holds are entered or exited in a non-tangential way and thus, also λ_u jumps. This completes the proof of Proposition 3.13(ii).

• (*i-v*) If $r = \bar{r}$, w = 0, $y_u = 0$ then -IV + III = -VI. This requires $u = \bar{r}$ and may appear inside Case 4 intervals when switching from Case 4(2) to Case 4(3). Here also $\dot{u} > \dot{\bar{r}}$ holds. This completes the first part of the proof of Proposition 3.13(iii).

(*ii*) Let θ_s^1 be the entry time and θ_s^2 the exit time of an interval where $y_s = 0$ holds, i.e. the beginning of a Case 2 or Case 4 interval. Then, λ_s is continuous at these time points if y_s enters or leaves this interval in a non-tangential way, i.e. $\dot{y}_s = p + r - d$ jumps. This happens if the sum of the controls p and r jumps at θ_s^1 .

(*ii-i*) An entry point θ_s^1 can be present at a switch (a) from Case 1(2) to Case 2 or (b) to Case 4 and (c) from Case 3 to Case 4 (other transitions are not possible when applying the optimal decisions in Case 1 and Case 3 intervals).

(a) In the first case, the transition is non-tangential (p + r does not jump)if $d = \bar{r}$ holds at the end of Subcase 1(2). Since $\lambda_s > \lambda_u + c_r$ holds inside Subcase 1(2) and $\lambda_s = \lambda_u + c_r$ inside Subcase 2(1), λ_s jumps at transition time. In order to leave Case 1(2), $d > \bar{r}$ must hold. Together with the definition of Subcase 2(1), $\dot{d} < \dot{\bar{r}}$ must hold at θ_s^1 . Since $\lambda_s < c_p$ holds inside Subcase 1(2) and $\lambda_s = c_p$ in Subcase 2(3), a transition from Subcase 1(2) to Subcase 2(3) would be continuous.

- (b) In situation (b), $d > \bar{r} > u$ must hold in order to clear both inventories simultaneously. The only transition requiring a jump in λ_s would be to Case 4(1). But from the definition of a Case 4 interval $d = \bar{r} = u$ must hold which is excluded by assumption (3.25).
- (c) In the third case, $d > \bar{r}$ must hold to empty the serviceables inventory. The only transition requiring a jump in λ_s is from Case 3(2) to Case 4(1). This necessitates $d = \bar{r}$ as well as $\dot{d} < \dot{\bar{r}}$ at θ_s^1 .

(ii-ii) An exit time θ_s^2 is present at a transition from (a) Case 2 to Case 1 and from Case 4 (b) to Case 1 or (c) Case 3.

- (a) A jump in λ_s only is required at a switch from Case 2(3) to Case 1(2) and it requires $d = \bar{r}$. Since $d < \bar{r}$ is required to build up serviceables stock during a Case 1(2) interval, $\dot{d} < \dot{\bar{r}}$ is required.
- (b) A switch from Case 4 to Case 1(2) necessitates $d < \bar{r} < u$ at the beginning of the Case 1(2) interval. This already excludes a transition starting in Case 4(2) since this would require $d = \bar{r} = u$. A transition starting in Case 4(3) needs $d = \bar{r}$ as well as $\dot{d} < \dot{\bar{r}}$.
- (c) $d < \bar{r}$ at the beginning of the Case 3(2) interval is required for a switch from Case 4 to Case 3(2). As before only a transition starting in Case 4(3) is possible, and $d = \bar{r}$ as well as $\dot{d} < \dot{\bar{r}}$ must hold at θ_s^2 .

This completes the last part of the proof of Proposition 3.13(i).

Let θ_u^1 be the entry time and θ_u^2 the exit time of an interval, where $y_u = 0$ holds (Case 3 or Case 4). Then, λ_u is continuous at these time points if y_u enters or leaves this interval in a non-tangential way, i.e. $\dot{y}_u = u - r - w$ jumps. That occurs if r + w jumps at (c) θ_u^1 or (d) θ_u^2 , respectively.

(*ii-iii*) An entry time θ_u^1 is present when switching from Case 1(2) (a) to Case 3(2) or (b) to Case 4, and (c) from Case 2 to Case 4.

- (a) A non-tangential transition from Case 1(2) to Case 3(2) requires $u = \bar{r}$. In order to leave Case 1(2), $u < \bar{r}$ must hold. Together with the definition of Subcase 3(2), $\dot{u} > \dot{\bar{r}}$ must hold at θ_u^1 .
- (b) When switching from Case 1(2) to Case 4, $d > \bar{r} > u$ must hold in order to clear both inventories simultaneously. The only transition requiring a jump in λ_u terminates in Case 4(3), requiring $u = \bar{r}$. From the definition of a Case 4 interval $\dot{u} > \dot{\bar{r}}$ must hold.
- (c) Besides the transition already discussed in (i-iv)(b) requiring a simultaneous jump in λ_s and λ_u , only a switch from Case 2(3) to Case 4(3) is possible. Depleting the recoverables inventory requires $u < \bar{r}$. Together with the definition of a Case 4(3) interval, besides $u = \bar{r}$ as well as $\dot{u} > \dot{\bar{r}}$ hold.

(*ii-iv*) An exit time θ_u^2 is present when switching (a) from Case 3(2) to Case 1(2) and from Case 4 (b) to Case 1, or (c) to Case 2.

- (a) A transition from Case 3(2) to Case 1(2) proceeds continuously.
- (b) A discontinuous switch w.r.t. λ_u from Case 4 to Case 1(2) would only be possible starting with a Case 4(2) interval. $d < \bar{r} < u$ must hold in order to build up both inventories simultaneously. From the definition of a Case 4(2) interval $d = \bar{r} = u$ must hold which is excluded by assumption (3.25).
- (c) As before, a transition of situation (c) has been discussed in (i-iv)(b). Aside from this, a switch from Case 4(2) to Case 2(3) is possible. A start of collecting returns in the recoverables inventory requires $u > \bar{r}$ at the beginning of a Case 2(3) interval. Together with the definition of a Case 4(2) interval, $u = \bar{r}$ and $\dot{u} > \dot{\bar{r}}$ hold at θ_u^2 .

This completes the proof of Proposition 3.13(iii).

Proof (Proof of Proposition 3.14).

In all cases where $y_u > 0$ it must hold that $-c_w < \lambda_u < c_p - c_r$ and $\dot{\lambda}_u = \alpha \lambda_u + h_u$. Proceeding in the same way as done in the proof of Proposition 2.7 finally yields $\theta_{x,c} - \theta_{e,c} \leq \tau_u$.

Proof (Proof of Proposition 3.17). From Proposition 3.9 we get a minimal value for λ_s of $\lambda_s^{\min} = \lambda_u(\theta_{e,s}^i) + c_r$. The upper bound is given by $\lambda_s^{\max} = c_p$. Together with continuity of λ_s inside the interval with positive serviceables stock and the solution of co-state development (3.9) yields

$$\left(\lambda_u(\theta_{e,s}^i) + c_r + \frac{h_s}{\alpha}\right) e^{\alpha\left(\theta_{x,s}^i - \theta_{e,s}^i\right)} - \frac{h_s}{\alpha} \le \lambda_s(\theta_{x,s}^i) \le c_p$$

Solving the combined inequality for $\theta_{x,s}^i - \theta_{e,s}^i$ finally yields

$$\theta_{x,s}^{i} - \theta_{e,s}^{i} \leq \frac{1}{\alpha} \ln \left(\frac{\alpha c_{p} + h_{s}}{\alpha \left(\lambda_{u}(\theta_{e,s}^{i}) + c_{r} \right) \right) + h_{s}} \right).$$

Proof (Proof of Proposition 3.18). From Proposition 3.12 we get a minimal value for λ_s of $\lambda_s^{\min} = c_r - c_w$. The upper bound is given by $\lambda_s^{\max} = c_p$. Together with continuity of λ_s inside the interval with positive serviceables stock and the solution of co-state development (3.9) yields

$$\left(c_r - c_w + \frac{h_s}{\alpha}\right) e^{\alpha(\theta_{x,s} - \theta_{e,s})} - \frac{h_s}{\alpha} \le \lambda_s(\theta_{x,s}^i) \le c_\mu$$

Solving the combined inequality for $\theta_{x,s} - \theta_{e,s}$ finally yields

$$\theta_{x,s} - \theta_{e,s} \le \frac{1}{\alpha} \ln \left(\frac{\alpha c_p + h_s}{\alpha \left(c_r - c_w \right) \right) + h_s} \right).$$

Knowledge Acquisition and Product Recovery

4.1 Motivation

So far, our analysis based on the assumption that all processes maintained an initial performance level and constant cost rates. However, ongoing competition and the search for profit maximization provide incentives for productivity improvements. This chapter relates to an empirical phenomenon found quite often in practical applications dating back to the 1920's and intends to explore effects of acquiring knowledge in product recovery on strategic recoverables inventory management. The learning curve, introduced by Wright (1936), shows a (potential) relationship between cumulative output and labor hours per unit produced. It's empirical evidence has been proven in a large number of studies, for comprehensive literature surveys see Yelle (1979) or Dutton and Thomas (1984). Traditionally, learning is seen as autonomous acquisition of tacit knowledge which is more prominent in labor intensive processes, but it also has been proven to occur in highly mechanized industries.

Because of the shortening of manufacturing times or more generally, reduction of input quantities, learning leads to an increase of the capacity of a manufacturing system and a decrease of direct production costs. This relationship has led to two main representations of the experience curve, as the relationship between cumulative output and unit costs is called. Let c^0 be the unit costs of producing the first item, c the unit costs of the X-th one, and b > 0 a learning parameter. According to Zangwill and Kantor (1998), the most widespread used functional forms of learning curves are

(a) the so-called *Power Law* form

$$c(X) = c^0 \cdot X^{-b} \tag{4.1}$$

(b) and the Mixed Exponential Learning form

$$c(X) = c^0 \cdot e^{-bX}.\tag{4.2}$$

The first type (a) represents the classical type first introduced by Wright (1936). This form predicts a reduction of unit production costs by a constant percentage each time the cumulative output doubles. The second function (b) is well-known from psychological learning theory, but also used for economic analyses (see Kantor and Zangwill (1991)). Both functional forms have in common that unit costs persistently decrease at a decreasing rate, i.e. further efficiency improvements become increasingly difficult to achieve. It should be noted, however, that since these functions only represent results of a system's inherent learning process, efforts have been taken to explain the underlying causes, resulting in refinements of the functional forms. Adler and Clark (1991) for instance investigate the impact of forced engineering changes and training on productivity improvement. Further factors that influence the learning rate have been elaborated by Argote and Epple (1990). These factors include organizational forgetting and employee turnover, both having a negative effect, and different ways to transfer knowledge between products or different departments of a company, which in turn increase the learning rate.

Quantitative approaches integrating learning effects into production planning decisions have been surveyed by Gulledge and Khoshnevis (1987). They include the following issues

• break-even analysis

Pegels (1976) aims to predict at which level of production volume total costs equal total revenue or at which point marginal costs equal marginal revenue.

• aggregate production planning

Ebert (1976) extends the well known HMMS model (Holt et al. (1960)) by introducing a dependence between productivity of workforce and cumulative production. Liao (1979) and Reeves and Sweigart (1981) consider this issue on a more detailed level aiming to determine the optimal *productmix*. In this work, a number of different resource requirements depend on the number of manufactured items of a product.

• economic lot sizing

The influence of learning on economic manufacturing quantity has been reviewed by Smunt and Morton (1985) and, more recently, by Jaber and Bonney (2001). In this context a number of different learning and 'forgetting' effects are investigated. For instance learning in the set-up of batch production leads to smaller lot sizes and a reduction of cycle stocks.

• dynamic pricing

Li and Rajagopalan (1998) analyze a dynamic profit maximization model, where a monopolistic firm sets price and production quantities by assuming a generic learning function and additionally considering knowledge depreciation.

Cost reductions due to learning are not restricted to production alone. It can also be presumed to occur for remanufacturing processes, if there are repeated operations performed on a large number of similar items. In such a case, and since remanufacturing operations usually include labor intensive tasks (Guide et al. (2000)), there exists a considerable potential for cost-reductions from acquiring e.g. tacit disassembly know how or due to specialized tools developed during the remanufacturing process. It must also be stated, that if there is a large diversity of remanufactured products with only a few information on how to deal with them correctly, sufficient experience might be more difficult to obtain. Under such circumstances remanufacturing usually would be dominated by lower levels of product recovery. This especially holds for independent recyclers whereas OEM have a competitive advantage due to an easier access to required information. See Toffel (2004) for a more detailed discussion.

In addition to the traditional volume learning effects, there is a second level of using the knowledge that was acquired during recovery operations. As indicated by Toffel (2004), a transfer of product recovery knowledge back to the production stage may lead to improvements in product design and manufacturing processes such that overall profitability increases.

The single-use camera case (see Chapter 1) provides a good example of large scale product recovery. Figure 4.1 exemplifies the increase in yearly volume indicating that considerable productivity improvements have been realized. With respect to the above-mentioned second level learning, there is an ongoing process of design changes to increase the reusability level. This is facilitated by the fact that the products are designed for 'reverse compatibility', which means that a new product generation can by design use some parts recovered from a previous generation. This helps to serve customer demand while still benefiting from a high product recovery level. For a more detailed discussion of this issue see Goldstein (1994).



Fig. 4.1. Single-use cameras recycled by Kodak between 1990 and 2003 (in Millions). Sources: Annual Reports 1999-2003, The Kodak Corporation.
Quantitative approaches devoted to effects of knowledge accumulation in product recovery can rarely be found. Exceptions are offered in the context of Total Quality Management. For instance, Lapré et al. (2000) present a model where learning leads to a reduction of unwanted by-products or waste inside the production process. The scarcity of these approaches is intuitive, because similar effects are expected when dealing with the pure remanufacturing process as encountered in production planning. When dealing with an *integrated* product recovery system new results are likely to occur. This chapter starts to fill this gap by analyzing the impact of acquiring remanufacturing knowledge on optimal decisions in such an environment. Possible research questions are how strategic stock-keeping changes and what would be the effect on optimal remanufacturing and production policies. For instance, as a strategic implication of productivity improvements, remanufacturing can be profitable in the long run, even if there is no immediate cost advantage over the production of new items, because subsequent unit remanufacturing costs are lowered. Here, benefits are derived from using an optimal control framework because, as explained in Chapter 2, the indirect effect of current decisions on future expenses is also valued.

The remainder of this chapter is organized as follows. In Section 4.2 an optimal control model is introduced which is further analyzed in Section 4.3. Special attention is drawn on the impact of zero and nonzero interest rates and by further distinguishing between specific cost/cash flow conditions, the main qualitative additions to the results of the basic formulation known from Chapter 2 are identified in Section 4.4. Numerical examples are used in Section 4.5 to illustrate the findings and final conclusions are presented in Section 4.6.

4.2 A Model with Remanufacturing Knowledge Acquisition

In this section a model is presented where current decisions on remanufacturing have an additional impact on future cash flow, because they change a knowledge stock which in turn lowers unit remanufacturing costs. The model uses the basic formulation introduced in Chapter 2 which has been adapted in order to account for knowledge acquisition. The following changes and assumptions are made.

Since only learning effects in the remanufacturing process are considered, it is assumed that the manufacturing process uses a 'mature' or old technology with negligible learning rate. Further, a re-transfer of remanufacturing knowledge back to the production stage (second level learning) is disregarded here, because this would destroy homogeneity of produced and later returned used products. Rate effects and knowledge decay are also not considered. A possible method for including knowledge depreciation into a dynamic framework has been presented by Li and Rajagopalan (1998). Learning in the remanufacturing shop is based on cumulative remanufacturing volume R which is derived, given an initial stock of knowledge R_0 , by using

$$\dot{R}(t) = r(t) \ge 0 \text{ and } R(0) = R_0 > 0.$$
 (4.3)

Constant out-of-pocket remanufacturing unit costs are replaced by a generic exogenous and deterministic function $c_r(R)$ measuring the learning curve effect realized so far. For this function we assume

$$c'_{r}(R) < 0, \ c''_{r}(R) > 0$$
, as well as $\lim_{R \to \infty} c_{r}(R) \ge 0$, (4.4)

i.e. unit remanufacturing costs decrease at a decreasing rate. Initial remanufacturing costs $c_r(R_0)$ can be so high that assumption (2.6) is violated and thus, a situation with a negative initial remanufacturing cost advantage $(c_p + c_w - c_r(R_0) < 0)$ in general is allowed.

All processes are supposed to be unrestricted. A serviceables inventory is therefore not considered here $(y_s(t) = 0 \forall t)$. Production quantity p(t) immediately follows, when setting a corresponding remanufacturing rate r(t), from the necessity to immediately satisfy all demand. Therefore, we have

$$p(t) = d(t) - r(t).$$
 (4.5)

Non-negativity of the production rate requires the return rate not to exceed the demand rate (r(t) < d(t)).

Besides an unchanged recoverables inventory transition equation

$$\dot{y}_u(t) = u(t) - r(t) - w(t) \tag{4.6}$$

initial and final recoverables inventory levels are set to zero $(y_u(0) = y_u(T) = 0)$. As usual, it should not be advantageous to hold unneeded returned products

$$h_u > \alpha c_w. \tag{4.7}$$

For analytical convenience but without loss of generality we assume strictly positive demand and return rates during the whole planning horizon

$$d(t) > 0 \text{ and } u(t) > 0.$$
 (4.8)

Thus, a constraint qualification required for the results presented in the next section is always satisfied.

Now, an optimal control problem with two states $(R \text{ and } y_u)$ and two control variables (r and w) has to be solved such that the total discounted cash outflow during a finite planning horizon is minimized. The problem is subject to the state equations, a pure state constraint, initial and (partly) terminal conditions for the state variables as well as non-negativity constraints for control variables and an upper limit for the remanufacturing rate

$$\min NPV = \int_0^T e^{-\alpha t} \left(c_p(d(t) - r(t)) + c_r(R(t))r(t) + c_w w(t) + h_u y_u(t) \right) dt$$

s.t. (4.3), (4.6), (4.9)
$$y_u(t) \ge 0 \ y_u(0) = 0, \ y_u(T) = 0,$$

$$d(t) - r(t) \ge 0, r(t) \ge 0, w(t) \ge 0.$$

4.3 Optimality Conditions and General Results

Necessary Conditions

There is no fundamental difference in optimal control theory when dealing with dynamic instead of static parameters and thus, the general solution procedure known from Chapter 2 can also be applied to the extended problem. Since the remanufacturing costs depend on a state, the objective loses the property of linearity in both controls and states. Necessary conditions as presented below are therefore no longer sufficient for optimality.

After introducing a new co-state variable λ_R , which corresponds to the stock of accumulated experience R, the Hamiltonian reads as

$$H(.) = \lambda_0 (-c_p (d-r) - c_r (R)r - c_w w - h_u y_u) + \lambda_u (u - r - w) + \lambda_R \eta (4.10)$$

Using Kuhn-Tucker multipliers μ_1 , μ_2 , μ_3 , and k_2 the Lagrangian is given by

$$L(..) = H(.) + \mu_1(d-r) + \mu_2 r + \mu_3 w + k_2 y_u.$$
(4.11)

In the appendix we show that $\lambda_0 = 1$. The adjoint λ_R rates the impact of the current remanufacturing decision on future costs and can therefore be interpreted as the shadow 'price' or *value of acquiring knowledge* and thereby changing the stock of experience R. It incorporates the discounted value of all improvements in future remanufacturing costs.

Now, let (y_u^*, R^*) represent the optimal trajectory of the state variables and (r^*, w^*) be a piecewise continuous trajectory of optimal control policies to problem (4.9). Then, there exists a continuous function of time λ_R as well as piecewise continuous functions of time $\lambda_u, \mu_i, i = 1, 2, 3, k_2$ and a set of points $\theta_u \in \Theta_u$ where the co-state λ_u jumps with corresponding height parameters $\eta_u(\theta_u)$. Except for points of discontinuity in the controls and junction points, the following necessary conditions (4.12)–(4.24) must hold.

The Hamiltonian is maximized, if the following bang-bang equations (4.12) and (4.13) hold

$$r^* = \begin{cases} 0 \quad \lambda_u > c_p - (c_r(R^*) - \lambda_R) \\ singular \quad \lambda_u = c_p - (c_r(R^*) - \lambda_R) \\ d \quad \lambda_u < c_p - (c_r(R^*) - \lambda_R) \end{cases}$$
(4.12)

$$w^* = \begin{cases} 0 \quad \lambda_u > -c_w \\ singular \quad \lambda_u = -c_w. \end{cases}$$
(4.13)

The remanufacturing rate is zero and demand is filled from production, if the value of a recoverables item exceeds the difference of current production expenses and the *net expenses of remanufacturing* (current direct costs less the future cost reductions valued by λ_R). r^* equals demand if the opposite holds and it can take any value between 0 and the demand rate, if there is indifference (equality of both sides). Similarly, a returned item is kept only, if its value exceeds a salvage revenue $(-c_w)$.

In order to maximize the Lagrangian, (4.14) and (4.15) must hold. Nonnegativity of controls as well as complementary slackness conditions apply as stated in (4.16)-(4.18).

$$\frac{\partial L}{\partial r} = c_p - c_r(R^*) - \lambda_u + \lambda_R - \mu_1 + \mu_2 = 0 \tag{4.14}$$

$$\frac{\partial L}{\partial w} = -c_w - \lambda_u + \mu_3 = 0 \tag{4.15}$$

$$\mu_1 \ge 0 \ \ \mu_1 \cdot (d - r^*) = 0 \tag{4.16}$$

$$\mu_2 \ge 0 \quad \mu_2 \cdot r^* = 0 \tag{4.17}$$

$$\mu_3 \ge 0 \ \ \mu_3 \cdot w^* = 0 \tag{4.18}$$

Optimal co-state transitions are given in (4.19) and (4.20), and complementary slackness condition for the recoverables inventory is represented by (4.21).

$$\dot{\lambda}_u = \alpha \lambda_u - \frac{\partial L}{\partial y_u} = \alpha \lambda_u + h_u - k_2 \tag{4.19}$$

$$\dot{\lambda}_R = \alpha \lambda_R - \frac{\partial L}{\partial R} = \alpha \lambda_R + c'_r(R^*) \cdot r^*$$
(4.20)

$$k_2 \ge 0 \quad k_2 \cdot y_u^* = 0. \tag{4.21}$$

New condition (4.20) can be interpreted as follows. The value of knowledge (i) increases with the interest to be paid on it $(\alpha \lambda_R)$, i.e. a later acquisition of experience through remanufacturing would have been cheaper in terms of the discounted value of required remanufacturing expenses, and it (ii) decreases with the rate at which remanufacturing unit costs decrease $(\dot{c}_r^* = c'_r(R^*) \cdot r^*)$. This is because a current cost reduction lowers the remaining reduction potential.

Transversality conditions for the co-state variables are given in (4.22) and (4.23)

$$\lambda_u(T)$$
 is free, (4.22)

$$\lambda_R(T) = 0. \tag{4.23}$$

The value of stored recoverables at the end of the planning period can not be predetermined because there the inventory is forced to zero. Instead of this, $\lambda_u(T)$ depends on the relation of demand and return functions. At the same time, remanufacturing knowledge becomes useless. Therefore, the corresponding co-state takes on a value of zero.

As in the basic model, there are points $\theta_u \in \Theta_U$ with $y_u^* = 0$ where downward jumps in λ_u and in the Hamiltonian occur.

$$\lambda_u(\theta_u^-) = \lambda_u(\theta_u^+) + \eta_u(\theta_u)$$
where $\eta_u(\theta_u) \ge 0$ and $\eta_u(\theta_u) \cdot y_u^*(\theta_u) = 0.$

$$(4.24)$$

Downward jumps in λ_R are not permitted.

Properties of an Optimal Solution

In this section particular features are derived that must hold for any optimal solution. We start with general results regarding the co-states, which are followed by a derivation of optimal policies in different cases in the state space. Lastly, conditions for case transitions are deduced. These findings are later used in Section 4.4 to examine optimal policies in more specific situations with respect to cash flow parameters.

General Properties of the Co-states $\lambda_{\rm R}$ and $\lambda_{\rm u}$

Reconsidering co-state movement (4.20), bounds for the value of acquiring knowledge λ_R as well as for its rate of change can be derived. Except for the special case where discounting can be neglected ($\alpha = 0$), the direction of the co-state development is not predetermined in (4.20), but it depends on the relation of the two terms on the right hand side of (4.20). Since c'_r is strictly negative, the second term is negative. If additionally the first term is less than zero, requiring the co-state to be negative, then the co-state is not able to rise again which contradicts transversality condition (4.23). Thus, the value of acquiring knowledge is limited to non-negative values, i.e.

$$0 \le \lambda_R(t). \tag{4.25}$$

The largest possible decrease of λ_R is given by the rate of change of remanufacturing costs (obtained when assuming a zero discount rate) and the maximal rise is determined by the opportunity costs of a later learning. Thus, in an optimal solution the following must hold

$$\dot{c}_r(t) \le \dot{\lambda}_R(t) \le \alpha \lambda_R(t), \tag{4.26}$$

i.e. the shadow price of acquiring knowledge must not decrease faster than current remanufacturing costs and it does not increase more rapidly than the interest to be paid on it.

The maximal possible decrease of λ_R in (4.26) determines an upper bound

$$\lambda_R(t) \le c_r \left(R^*(t) \right) - c_r(R^*(T)) \,. \tag{4.27}$$

Otherwise, transversality condition (4.23) again would be infeasible. Inequality (4.27) implies that the value of knowledge acquisition at time t lies below the remaining remanufacturing cost reduction. As can easily be seen when inserting $\alpha = 0$ into (4.20), equality must hold in (4.27) if $\alpha = 0$. The following lemma summarizes the results regarding λ_R (presented without proof).

Lemma 4.1. The value of acquiring knowledge λ_R must lie between zero and the remaining remanufacturing cost reduction. If discounting can be neglected $(\alpha = 0)$, then λ_R exactly equals the remaining remanufacturing cost reduction.

Another important result concerns the upper bound of the value of returns (λ_u) . In the basic model, this bound was given by the difference of manufacturing and remanufacturing costs, $\lambda_u^{\max} = c_p - c_r$. In our case, a similar but time dependent boundary applies if remanufacturing takes place, i.e. $r^* > 0$ requires $\lambda_u \leq c_p - (c_r(R^*) - \lambda_R)$ from (4.12). The first derivative of the right hand side of this inequality w.r.t. time yields

$$\frac{d}{dt}\left(c_p - \left(c_r(R^*(t)) - \lambda_R(t)\right)\right) = \alpha \lambda_R(t).$$
(4.28)

That means that the difference between production expenses and net expenses of remanufacturing increases with non-negative rate $\alpha \lambda_R$. Consequently, if $\alpha = 0$ it is time independent and equals the projected difference of production and remanufacturing expenses at the end of the planning period. This leads us to Lemma 4.2 (presented without proof).

Lemma 4.2. The maximal level which the value of returns is allowed to achieve if remanufacturing takes place $(r^* > 0)$ rises with rate $\alpha \lambda_R$. In case of zero discounting $(\alpha = 0)$, this level is constant and given by $\lambda_u^{\max} = c_p - c_r(R^*(T))$.

Cases in State Space and Optimal Policies

Since only a single stocking point (the recoverables inventory) is considered, there are only two cases, one with a positive $(y_u > 0)$ and another with a zero inventory level $(y_u = 0)$. In order to remain consistent with Chapter 2, these cases are named Case 2 and 4, respectively. The basic model can be seen as a special case with a zero learning rate, all policies (cases and subcases) that apply are also relevant in the present situation. Further policies are available and conditions for the already known subcases change as shown in the following two propositions.

Proposition 4.1 (Optimal decisions in Case 2 intervals).

If recoverables inventory is positive $(y_u^* > 0)$, the optimal policy is not to dispose of returns $(w^* = 0)$. The decision on whether to remanufacture or to produce depends on a relationship based on co-states and current remanufacturing costs. Two subcases can be distinguished:

Subcase $2(1) \Leftrightarrow \lambda_u < c_p - (c_r(R^*) - \lambda_R)$ Demand is satisfied from remanufacturing returns $(r^* = d)$ and no items are produced $(p^* = 0)$. **Subcase** $2(2) \Leftrightarrow \lambda_u > c_p - (c_r(R^*) - \lambda_R)$

Remanufacturing does not take place $(r^* = 0)$ and demand is filled from production alone $(p^* = d)$.

Under positive recoverables inventory conditions, two different policies might apply. Both have in common that disposal is not optimal, because otherwise items could have been disposed of earlier (from stock or directly when arriving) which would have saved holding costs. Thus $\lambda_u > -c_w$. The question on how to fill demand, either completely from remanufacturing out of stock or by producing new items, depends on whether or not the value of returns exceeds the difference of production expenses c_p and net remanufacturing costs (including indirect learning effects) $c_r(R^*) - \lambda_R$.

The already known Subcase 2(1) was used in the basic model for building an anticipation stock. Subcase 2(2), however, might be used in order to postpone remanufacturing decisions, if holding an item and remanufacturing it later would be cheaper than immediately processing it.

Proposition 4.2 (Optimal decisions in Case 4 intervals).

If recoverables inventory is empty $(y_u^* = 0)$, optimal decisions depend on the net recovery cost advantage and on how demand relates to the return rate. Here three subcases can be distinguished:

Subcase 4(1) $\Leftrightarrow d \leq u, \lambda_u = -c_w, and c_p + c_w > c_r(R^*) - \lambda_R$

Demand is satisfied completely by remanufacturing returns $(r^* = d)$ and excess returns are disposed of $(w^* = u - d)$. No items are produced $(p^* = 0)$.

Subcase 4(2) \Leftrightarrow u < d, $\lambda_u = c_p - c_r(R^*) + \lambda_R$, and $c_p + c_w > c_r(R^*) - \lambda_R$ All returns are remanufactured ($r^* = u$) and the missing items are produced ($p^* = d - u$). No items are disposed of ($w^* = 0$).

Subcase 4(3) $\Leftrightarrow \lambda_u = -c_w, c_p + c_w < c_r(R^*) - \lambda_R$ Remanufacturing does not take place $(r^* = 0)$ and demand is filled from

production alone $(p^* = d)$. All returns are disposed of $(w^* = u)$.

In the case of zero inventories, besides those two policies already known from the basic model, a third subcase applies. While the first two aim to maximize current remanufacturing volume, Subcase 4(3) is characterized by a zero remanufacturing rate and disposal of all returns. It is preferred to the other two if net remanufacturing costs exceed the costs of producing a new item and disposing of the old one, i.e. in situations where remanufacturing is not preferable. Depending on optimal decisions λ_u either takes on its lower bound $-c_w$ if disposal takes place (Subcases 4(1) and 4(3)) or its upper bound $\lambda_u = c_p - c_r(R^*) + \lambda_R$ in case it does not (Subcase 4(2)).

Main results of both cases including state and co-state developments are summarized in Table 4.1.

	r^*	w^*	\dot{y}_u	λ_u	$\dot{\lambda}_u$	λ_R	$\dot{\lambda}_R$	k_2						
	(\dot{R})													
Case 2: $y_u > 0$														
(1)	d	0					$\alpha \lambda_R + c'_r(R^*)d$	0						
				< A	$(>0)\\\alpha\lambda_u + h_u$		$lpha\lambda_R$							
(2)	0	0	u	$-c_w <$	$\alpha \lambda_u + h_u$		$\alpha\lambda_R$	0						
				A <	(> 0)									
Case 4: $y_u = 0$														
(1) $(d \le u)$	d	u-d	0	$-c_w$	0	B <	$\alpha\lambda_R + c'_r(R^*)d$	$h_u - \alpha c_w$						
				< A			_							
(2) $(u < d)$	u	0	0	A	$\alpha\lambda_R$	B <	$\alpha\lambda_R + c'_r(R^*)u$	$\alpha(\lambda_u - \lambda_R) + h_u$						
	0			$-c_w <$	0									
(3)	0	u	0	$-c_w$	0	< B	$\alpha\lambda_R$	$h_u - \alpha c_w$						
				A <										
Abbreviat														
$A = c_p - (c_r(R^*) - \lambda_R) \dots$ cost advantage of remanufacturing including learning														
$B = c_r(R^*) - c_p - c_w$ current (direct) cost dis advantage of remanufacturing														

 Table 4.1. Main results of optimal cases in a product recovery system with learning.

Optimal Transitions Between Cases and Subcases

As in the previous chapters, we classify transitions with respect to their dependence on an actual decision to build up stock, and differentiate between *forced* and *automatic* case transitions. A new type is named a *learning forced* transition which happens, if the optimal policy switches between non-performance and start of the remanufacturing process. As usual, $A \rightarrow B$ reads as a transition from a Case A to a Case B interval, and *continuous* and *discontinuous* transitions are distinguished.

An exploration of the continuity properties of the adjoints leads to Proposition 4.3.

Proposition 4.3 (Continuity of the co-states). λ_R is always continuous. λ_u is continuous, i.e. jump parameter η_u vanishes everywhere, except at time points $\theta \in \Theta$ where $y_u(\theta) = 0$ and $u(\theta) = d(\theta)$ holds.

Corollaries 4.1-4.3 provide results regarding possible case transitions. The proofs are not completely straightforward and therefore explicitly given in the proof section. Since not all transitions are possible under all cash flow parameter conditions, we distinguish between low (**L**), moderate (**M**), and high (**H**) values of $c_r(R^*(t))$ being defined as follows

- Low: $c_r(R^*(t)) \leq c_p + c_w$ (and consequently, $\alpha(c_r(R^*(t)) c_p) < h_u)$, There currently exists a positive direct recovery cost advantage.
- Moderate: $c_r(R^*(t)) > c_p + c_w$ and $\alpha(c_r(R^*(t)) c_p) \le h_u$, Remanufacturing does not immediately pay off. Out of pocket holding costs are higher than savings from deferring to remanufacture an item.

• High: $h_u < \alpha(c_r(R^*(t)) - c_p)$ (including $c_r(R^*(t)) > c_p + c_w)$, Remanufacturing is so expensive, that holding an item is cheaper than the interests that are saved when postponing remanufacturing of that item.

The following discussion concentrates on transitions that are not known from the basic model. Each transition incorporating one of the new cases (Subcase 2(2) and 4(3)) will be briefly discussed and it is indicated which cost condition is required.

Corollary 4.1. Within Case 2 Subcase 2(2) is followed by a Subcase 2(1) interval. This continuous and automatic transition requires $\lambda_u = c_p - (c_r(R^*) - \lambda_R)$ and is only possible under high remanufacturing cost conditions (**H**).

Subcase 2(2) accumulates returns. In order to deplete the stock, a switch to Subcase 2(1) is required. This happens at a time at which the value of returns equals the difference of production and net remanufacturing costs. The required cost scenario (**H**) indicates that Subcase 2(2) is used to postpone the start of remanufacturing in Subcase 2(1) while still able to recover value from collected returns.

Corollary 4.2. Within Case 4 the following transitions are possible: $4(2) \rightarrow 4(1)$. This automatic and discontinuous transition requires d = uand $\dot{d} < \dot{u}$.

- $4(3) \rightarrow 4(1)$. This learning forced and continuous transition requires d < u, $\lambda_R = c_r(R^*) - c_p - c_w$, and moderate or high remanufacturing costs (\mathbf{M}, \mathbf{H}) .
- $4(3) \rightarrow 4(2)$. This learning forced and continuous transition requires d > u, $\lambda_R = c_r(R^*) - c_p - c_w$, and moderate remanufacturing costs (**M**).

As the value of acquiring knowledge reaches a critical value, i.e. the current recovery cost disadvantage, a switch from Subcase 4(3) to one of the other two subcases takes place. Transition $4(3) \rightarrow 4(2)$ additionally requires a moderate cost situation. Otherwise, stock-keeping of returns and postponing the start of remanufacturing would be preferable. This is feasible since there is excess demand available after a transition time which could be filled from stored returns.

Corollary 4.3. Between a Case 2 and a Case 4 interval the following transitions are possible:

- 2(1) \rightarrow 4(1). This automatic and discontinuous transition requires d = u and $\dot{d} < \dot{u}$.
- $2(1) \rightarrow 4(2)$. This automatic and continuous transition requires d > u.
- $4(1) \rightarrow 2(1)$. This forced and continuous transition requires d < u.
- $4(2) \rightarrow 2(1)$. This forced and discontinuous transition requires d = u and $\dot{d} < \dot{u}$.
- $4(3) \rightarrow 2(2)$. This learning forced and continuous transition requires high remanufacturing costs (**H**).

A transition from Case 4(3) to 2(2) takes place as disposal of returns ceases and accumulation for later remanufacturing during a Subcase 2(1) interval starts. Since Subcase 2(2) is left terminating in a Subcase 2(1) interval, the same remanufacturing cost condition must hold.

All possible transitions and relevant initial cost conditions are depicted in Figure 4.2. The grey shaded area highlights transitions that are known from the basic model.



Fig. 4.2. Optimal case transitions in a product recovery system with learning.

Corollaries 4.1-4.3 showed that there are no optimal transitions from the already known (2(1),4(1),4(2)) to new subcases (2(2), 4(3)). The latter might therefore only occur during an initial period and the whole planning period can be divided into two parts, the first where no remanufacturing takes place and another where it does. Let $\theta_l \in [0, T]$ be the time which divides the two parts, i.e. after which remanufacturing is used. Corollary 4.4 summarizes the results regarding case transitions (presented without proof).

Corollary 4.4. After remanufacturing a first returned item at time θ_l this process is continued until the end of the planning horizon, i.e. $r^*(t) \ge 0 \forall t \ge \theta_l$. Cases 4(3) and 2(2) are only present during an initial period and in both cases it holds that $R^* = R_0$.

Therefore, for any transition involving Subcases 2(2) and 4(3) it holds $R^*(t) = R_0$ and the $(\mathbf{L}/\mathbf{M}/\mathbf{H})$ distinction introduced before bases on *initial* unit remanufacturing costs $c_r(R_0)$. Corollaries 4.1-4.3 have to be adapted accordingly.

As another interesting implication of the split planning period the development of the value of acquiring knowledge λ_R in general proceeds as follows

$$\dot{\lambda}_R(t) = \begin{cases} \alpha \lambda_R & t < \theta_l \\ \alpha \lambda_R + c'_r(R^*) \cdot r^* & t \ge \theta_l \end{cases}.$$
(4.29)

Using boundary condition (4.23) the solution of this differential equation is

$$\lambda_R(t) = \begin{cases} e^{\alpha(t-\theta_l)} \left(-\int\limits_{\theta_l}^T e^{\alpha(\theta_l-s)} \left(c'_r(R^*(s)) \cdot r^* \right) ds \right) & t < \theta_l \\ -\int\limits_t^T e^{\alpha(t-s)} \left(c'_r(R^*(s)) \cdot r^* \right) ds & t \ge \theta_l \end{cases}$$
(4.30)

Equation (4.30) can be interpreted as follows. At any point later than θ_l , the value of knowledge acquisition is given by the discounted value (in terms of that time point) of all later cost reductions $\dot{c}_r = c'_r(R^*) \cdot r^*$. For all time points $t < \theta_l$, $\lambda_R(\theta_l)$ is only adapted to account for different time value, i.e. it is discounted from θ_l down to t.

4.4 Optimal Policies in Specific Situations

This section deals with qualitative additions of remanufacturing knowledge accumulation on the strategic stock-keeping known from the basic model. Since the effects differ when having a zero or non-zero interest rate, in the next two subsections optimal policies for each of these two cases are developed.

4.4.1 Optimal Policy with a Zero Interest Rate

The case of negligible discounting is characterized by an equal valuation of all payments independent of their timing. A postponement of remanufacturing decisions does therefore not make sense and the solution takes on a simple structure as provided in Proposition 4.4.

Proposition 4.4. If the interest rate is zero, then either remanufacturing takes place right from start of the planning period or it does not take place at all, i.e. $\theta_l \in \{0, T\}$.

Thus, the optimal solution either is characterized by a sequence of intervals of Cases 2(1), 4(1) and 4(2) (*Type 1*), or alternatively Case 4(3) is present throughout the planning period (*Type 2*). Both types are now examined in detail.

A Type 1 solution exhibits the same structure as the basic model without initial inventories. It requires the value of acquiring knowledge (1) to exceed a possible direct recovery cost disadvantage $(\lambda_R(t) > c_r(R^*(t)) - c_p - c_w \forall t)$ and (2) to continuously decrease with the same rate at which remanufacturing costs fall ($\dot{\lambda}_R = \dot{c}_r$). The value of returns depend on the relation of demand and return functions and on whether it is optimal to build up an anticipation stock of recoverables. Optimal Case 2(1) intervals require attributes like a Location Property (Corollary 2.6) and Inventory Conditions (Proposition 2.8), which both remain unchanged. A Maximal Length Property similar to Proposition 2.7 can be derived by using results provided by Lemma 4.2. Subsequently, let $\theta_{e,i}$ and $\theta_{x,i}$ denote the entry and exit time of a Case i interval. **Proposition 4.5 (Maximal Length Property where** $\alpha = 0$). If the interest rate is zero, then the maximal length of a Case 2(1) interval $I = (\theta_{e,2(1)}, \theta_{x,2(1)})$ is given by

$$(\theta_{x,2(1)} - \theta_{e,2(1)}) \le \tau_u := \max\left\{0, \frac{c_p + c_w - c_r(R^*(T))}{h_u}\right\}$$

The Maximal Holding Time τ_u is constant over time, and it immediately anticipates all later-acquired experience right from the beginning. It now balances a trade-off between incurred holding costs and the *net* recovery cost advantage.

A Type 2 solution is characterized by disposal of all returns which is only optimal if $\lambda_u(t) = -c_w \ \forall t$. Since there is no learning, the value of acquiring knowledge is zero and it never exceeds the recovery cost disadvantage, i.e. $0 = \lambda_R(t) < c_r(R^*(t)) - c_p - c_w \ \forall t$.

It remains to be seen which type of solution applies. In the case of low initial remanufacturing costs (**L**) where $c_r(R_0) \leq c_p + c_w$, Case 4(3) is invalid and remanufacturing takes place. But if we have a situation with moderate initial remanufacturing costs where $c_r(R_0) > c_p + c_w$ (**M**, while **H** conditions are excluded by definition when interest rate is zero) it is questionable whether the investments spent for 'riding down the experience curve' (i.e. into the initial recovery cost disadvantage) can later be recaptured. Since independent of their timing all payments are valued equally, this question can be answered by using a break-even type of analysis. Let \tilde{R} be a return quantity at which total remanufacturing costs equal total costs of disposal and production of new items, i.e.

$$\tilde{R} : \int_{R_0}^{R_0 + \tilde{R}} c_r(x) \, dx = (c_p + c_w) \tilde{R}.$$
(4.31)

Then, the following proposition is used to decide upon the solution type.

Proposition 4.6. If the interest rate is zero, then remanufacturing takes place if the total remanufacturing quantity surpasses a break-even total remanufacturing quantity, i.e. $R^*(T) \ge \tilde{R}$.

Since the total remanufacturing quantity $R^*(T)$ plays a role both in determining τ_u and in the question whether to remanufacture or not, the choice of the planning horizon T becomes a critical decisive factor. A longer planning period would lead to an increased Maximal Holding Time and thus to more stock-keeping. But also remanufacturing would more likely pay off because total remanufacturing rises. This result complies with the strategic focus of the learning curve approach.

The main problem when constructing the optimal solution, is to find the right total remanufacturing quantity $R^*(T)$ which on one side depends on the

Maximal Holding Time τ_u and on the other side is required to determine just it. In order to overcome these difficulties we propose the following solution algorithm.

Algorithm 4.1

Step 1

Start with an initial total remanufacturing quantity $\hat{R}(T)$. Determine a preliminary solution by using Algorithm 2.1 (see Section 2.3.6) assuming a Maximal Holding Time of $\tau_u = \max\left\{0, \frac{c_p + c_w - c_r(\hat{R}(T))}{h_u}\right\}$. Evaluate for this solution the total remanufacturing quantity $R^0(T)$.

Step 2

let i := 0. repeat

let i := i + 1

Determine a preliminary solution under the assumption of a Maximal Holding Time $\tau_u = \max\left\{0, \frac{c_p + c_w - c_r(R^{i-1}(T))}{h_u}\right\}$. Evaluate for this solution the total remanufacturing quantity $R^i(T)$. until $|R^i(T) - R^{i-1}(T)| < \varepsilon$.

Step 3

The solution determined in Step 2 is (approximately) optimal if $R^i(T) \ge \tilde{R}$. Otherwise remanufacturing does not take place.

Algorithm 4.1 starts with an initial guess of the total remanufacturing quantity (Step 1) and it iteratively improves this value in Step 2 until a sufficient precision (measured by a parameter ε) is reached. In Step 3, it is checked whether remanufacturing is performed at all. A possible way to initialize $\hat{R}(T)$ would be to use break-even quantity \tilde{R} or the maximal potential total experience given by the sum of initial experience and accumulated return rate, i.e. $R_0 + \int_0^T u(t)dt$. The algorithm converges, because if $R^{i-1}(T)$ was chosen too large, then terminal remanufacturing costs are under- and Maximal Holding Time is overestimated. Thus, still too many items are remanufactured and therefore $R^i(T)$ is also located in between $R^*(T)$ and $R^{i-1}(T)$. A similar argument holds for too low values of $R^{i-1}(T)$.

4.4.2 Optimal Policy with a Positive Interest Rate

When discounting matters, three main modifications to the previous case occur. First of all, even if remanufacturing does take place it is not necessarily useful to start it at the beginning of the planning horizon. Therefore, the question must be answered when to start this process. A trade-off is struck between early remanufacturing that leads to higher but later direct cost savings, and a lower discounted value of the initial expenses if it is started later. Secondly, under (**H**) initial remanufacturing cost conditions, a strategic inventory might be used to postpone remanufacturing. Finally, the Maximal Holding Time is no longer constant. The following proposition uses results derived in Lemma 4.2.

Proposition 4.7 (Maximal Length Property where $\alpha > 0$).

If the interest rate is positive, then the maximal length of a Case 2 interval $I = (\theta_{e,2}, \theta_{x,2})$ is time dependent, and it increases over time. It is given as a function of the exit time of the respective Case 2(1) interval

$$(\theta_{x,2} - \theta_{e,2}) \le \tau_u(\theta_{x,2}) := \frac{1}{\alpha} \ln \left(\frac{\alpha \left(c_p - \left(c_r(R^*(\theta_{x,2})) - \lambda_R(\theta_{x,2}) \right) \right) + h_u}{-\alpha c_w + h_u} \right).$$

The Maximal Length Property corresponds to that known from the basic model (Proposition 2.7), but here remanufacturing expenses (c_r) are replaced by net remanufacturing costs $(c_r(R^*(\theta_{x,2})) - \lambda_R(\theta_{x,2}))$. In the case of positive discounting, not all later acquired experience is anticipated because the respective cost savings are valued less than current expenses. Since this difference in time value decreases as time advances, the Maximal Holding Time rises with time, i.e. a later Case 2 interval is allowed to be longer than an earlier one.

Next, we consider the different initial remanufacturing cost situations.

Low Initial Remanufacturing Costs: $c_r(R_0) \le c_p + c_w$

Under (**L**) conditions there is an immediate advantage of remanufacturing over producing new items and disposing of the old ones. The optimal policy in such a situation therefore is to start remanufacturing as early as possible, i.e. $\theta_l = 0$ and to follow a *Type 1* policy. An anticipation stock of returns is held under circumstances as described before, but in contrast to the zero discounting case, the Maximal Holding Time τ_u utilizes Proposition 4.7 and must therefore be determined individually for each Case 2(1) interval. This is accomplished by applying a simple procedure as sketched below.

Algorithm 4.2

Start with the solution of Algorithm 4.1. Let n be the number of Case 2 intervals and for all i = 1 to n let $\tau_u^i = (\theta_{x,2}^i - \theta_{e,2}^i)$. repeat for all i = 1 to n. if $\tau_u(\theta_{x,2}^i) < \tau_u^i$ then decrease τ_u^i if $\tau_u(\theta_{x,2}^i) > \tau_u^i$ then increase τ_u^i end for Determine a new solution by using Algorithm 2.1 under the assumption of a sequence of individual Maximal Holding Times $\{\tau_u^i\}$. until max $\{\tau_u(\theta_{x,2}^i) - \tau_u^i\} < \varepsilon$.

Algorithm 4.2 iteratively improves an initial guess for the sequence of Maximal Holding Times $\{\tau_u^i\}$. Note that the algorithm neglects the possibility of a Case 2 interval that has been joined during interval construction (see Algorithm 2.1 in Section 2.3.6). This means that when decreasing the Maximal Holding Time during Step 2 it might occur that this interval is split into two succeeding independent intervals, and for each one a different Maximal Holding Time must be determined.

Moderate Initial Remanufacturing Costs: $c_r(R_0)>c_p+c_w$ and $\alpha(c_r(R_0)-c_p)\leq h_u$

If the initial remanufacturing expenses exceed the sum of direct cost of producing a new item and disposing of the old one (**M**) it is possible to start remanufacturing later than at time zero, which leads to a third solution type (*Type* 3) being attributed by a period where Case 4(3) is present followed by another sequence including Cases 2(1)/4(1)/4(2). Depending on the current demand/return situation both periods are connected by a transition $4(3) \rightarrow 4(1)$ or $4(3) \rightarrow 4(2)$ at time θ_l . For these transitions, a break-even like condition for the net recovery cost advantage at the time where remanufacturing starts is provided by Proposition 4.8.

Proposition 4.8. At the optimal start time of remanufacturing $\theta_l^* > 0$ it must hold that

$$\lambda_R(\theta_l^*) = c_r(R_0) - c_p - c_w. \tag{4.32}$$

Thus, at θ_l^* the value of acquiring knowledge must exactly outweigh the initial recovery cost disadvantage. If it would be lower, then the net recovery cost advantage would be negative and production would be preferable. In the opposite case, remanufacturing could have been started earlier since there also was a positive cost advantage.

Similarly to the zero interest rate case, having a point satisfying Proposition 4.8 does not necessarily mean that investments in knowledge acquisition pay off. But in contrast to that case it is not possible to formulate a simple condition as presented in Proposition 4.6. But when considering the differences in the time values it needs to be said that the total remanufacturing quantity must be higher than break-even quantity \tilde{R} derived before in order to amortize initial high costs.

Since only first order necessary optimality conditions were considered, local and global optimality as well as existence and uniqueness of a candidate satisfying condition (4.32) is not assured. A possible solution method would therefore be to numerically search for all points for which break-even condition (4.32) holds and to compare the respective objective value of each candidate with the non-performance of remanufacturing ($\theta_l = T$) and with start of remanufacturing at time zero ($\theta_l = 0$).

High Initial Remanufacturing Costs: $h_u < \alpha (c_r(R_0)-c_p)$ (and $c_r(R_0)>c_p+c_w)$

Under high remanufacturing cost conditions (**H**), a strategic inventory might be used in order to further postpone the start of remanufacturing. This occurs as a sequence $4(3) \rightarrow 2(2) \rightarrow 2(1) \rightarrow 4(1)/4(2)$, i.e. at time $\theta_{e,2(2)}$ disposal of returns stops which then are accumulated during a Subcase 2(2) interval. After switching to Subcase 2(1) at time θ_l , the returns are used up. This requires that at the end of the respective interval demand exceeds the return rate in order to deplete the recoverables stock.

The Case 2 interval is built around a time point θ_l where

$$\lambda_u(\theta_l) = c_p - c_r(R^*(\theta_l)) + \lambda_R(\theta_l) \tag{4.33}$$

which can be interpreted as another Location Property. Aside a Maximal Length Property as provided with Proposition 4.7, Inventory Conditions must hold which assure a positive stock-level during the whole Case 2 interval and a zero level at exit time. These considerations lead to Proposition 4.9 (presented without proof).

Proposition 4.9 (Inventory Conditions).

Let $I = (\theta_{e,2(2)}, \theta_l) \cup [\theta_l, \theta_{x,2(1)})$ be the open time interval of a sequence $2(2) \rightarrow 2(1)$ where $y_u > 0$ and $y_u(\theta_{e,2}) = y_u(\theta_{x,2}) = 0$. Then,

(i) cumulative demand equals cumulative returns over the whole interval

$$\int_{\theta_{e,2(2)}}^{\theta_l} u(t) dt + \int_{\theta_l}^{\theta_{x,2(1)}} (u(t) - d(t)) dt = 0, \qquad (4.34)$$

(ii) at any point $\theta \in [\theta_l, \theta_{x,2(1)})$, cumulative returns must be larger than cumulative demand, especially

114 4 Knowledge Acquisition and Product Recovery

$$\int_{\theta_{e,2(2)}}^{\theta_l} u(t) \, dt + \int_{\theta_l}^{\theta} (u(t) - d(t)) \, dt > 0.$$
(4.35)

This result has been adapted from Proposition 2.8 by additionally taking into account the two different policies inside the interval where recoverables are kept.

4.5 Numerical Examples

In this section four numerical examples based on two different demand/return scenarios are used to exemplify the main results of the model with learning. All examples use a learning function according to the Power Law, i.e.

$$c_r(R) = c_r^0 R^{-b} (4.36)$$

holds with an 80% progress ratio (b=0.32) and $R_0 = 1$. Common cash flow parameters are $c_p = 2$, $c_w = 1$.

Example 4.1 The first example aims to show the effects of learning when having a zero interest rate. It uses a scenario which remains basically the same as Example 2.1 in Chapter 2 except for remanufacturing cost rate. Demand and return functions are given as follows

$$d(t) = 1 + 0.5\sin(t)$$
 and $u(t) = 0.7d(t - \pi)$. (4.37)

The planning horizon has been extended to $T = 6\pi$ in order to increase the number of possible Case 2 intervals. This cyclical scenario is depicted in Figure 4.3.



Fig. 4.3. Demands and returns in Examples 4.1 and 4.2.

Further parameters are $c_r^0 = 3$ and $h_u = 1$. Thus, there is no initial recovery cost disadvantage leading to a break-even total remanufacturing quantity $\tilde{R} = 0$. The solution is naturally of Type 1 ($\theta_l = 0$).

Algorithm 4.1 was used to solve this example yielding a total optimal remanufacturing quantity $R^*(T) = 11.1371$, $c_r(R^*(T)) = 1.3872$ and thus, a Maximal Holding Time $\tau_u = 1.6128$. The optimal solution is depicted in Figure 4.4 showing three collection intervals where the first two have maximal length and the last one is shorter because the planning horizon is reached. Optimal Case 2(1) intervals are shown in Table 4.2.



Fig. 4.4. Optimal solution of Example 4.1.

Table 4.2. Optimal Case 2(1) intervals in Examples 4.1 and 4.2.

	$\theta_{e,2}^1$	$\theta^1_{x,2}$	$\theta_{e,2}^2$	$\theta_{x,2}^2$	$ heta_{e,2}^3$	$\theta^3_{x,2}$
Example 4.1	5.0715	6.6842	11.3547	12.9674	18.1107	Т
Example 4.2	5.0752	6.6814	11.3141	12.9990	18.1107	Т

The optimal development of the value of returns λ_u can be found in Figure 4.5 and Figure 4.6 exhibits the decreasing slope of the value of acquiring knowledge λ_R .

The sensitivity of the Maximal Holding Time on the length of the planning horizon can be seen when modifying it, $T = 4\pi$ would lead to $\tau_u = 1.4126$ whereas $T = 8\pi$ yields $\tau_u = 1.7394$.

Example 4.2 For the second example the discount rate was modified to $\alpha = 0.2$. This leads to modified optimal Case 2(1) intervals which are determined by using Algorithm 4.2 and also shown in Table 4.2. The first two intervals



Fig. 4.5. Optimal development of co-state λ_u in Example 4.1.



Fig. 4.6. Optimal development of co-state λ_R in Example 4.1.

each having full length show an increasing holding time of $\tau_u(\theta_{x,2(1)}^1) = 1.6062$ and $\tau_u(\theta_{x,2(1)}^2) = 1.6849$. As before, the length of the third collection interval is limited by the end of the planning period, but at this point a maximal length of $\tau_u(\theta_{x,2(1)}^3) = 1.7277$ be allowed. Figure 4.7 shows the optimal solution.

The optimal recoverables co-state development is depicted in Figure 4.8. The upper dotted line shows how the maximal allowed value of returns rises with time and therefore, the Maximal Holding Time increases. The evolution of the value of acquiring knowledge can be seen from Figure 4.9.



Fig. 4.7. Optimal solution of Example 4.2.



Fig. 4.8. Optimal development of co-state λ_u in Example 4.2.



Fig. 4.9. Optimal development of co-state λ_R in Example 4.2.

Example 4.3 The third and fourth examples are used to exemplify the problem of choosing the time at which remanufacturing starts (θ_l) . In both examples a scenario is used where the planning horizon $T = 3\pi$. Demand and return functions are

$$d = 1 + \sin(t/2)$$
 and $u = 0.75 d(t - 1.5\pi)$. (4.38)

While the demand function approximates a product life cycle, the return function is set in a way that it shows two peaks. This could for instance be motivated by assuming that each peak corresponds to one out of two classes of returns: (i) commercial returns, e.g. defective items that are assumed to occur more often in the beginning of the product life cycle and (ii) used products. Both types are assumed to be remanufactured using the same process and at the same costs. This scenario is depicted in Figure 4.10.



Fig. 4.10. Demands and returns in Examples 4.3 and 4.4.

Cash flow parameters are as in Example 4.2 except for initial remanufacturing costs, which now are given by a moderate level of $c_r^0 = 3.8$. Therefore, a situation with a positive discount rate and parameter condition (**M**) is considered.

The optimal time where remanufacturing starts is determined using Proposition 4.8. Since there is no reason for keeping stock in this example, $\lambda_R(\theta_l)$ follows from inserting the optimal remanufacturing decisions in Cases 4(1)/(2) into (4.30)

$$\lambda_R(\theta_l) = -\int_{\theta_l}^T e^{\alpha(\theta_l - s)} \left(c'_r \left(R_0 + \int_{\theta_l}^s \min\{d(t), u(t)\} dt \right) \min\{d(s), u(s)\} \right) ds$$
(4.39)

Inserting (4.39) into (4.32) and solving for θ_l yields 1.7176 and 5.9718. Comparing the respective objective values with the non-performance of remanufacturing and with starting at time zero yields $\theta_l = 1.7176$ as the optimal time to start remanufacturing. The optimal policy is therefore characterized by disposal of commercial returns up to θ_l because later cost savings that are due to the acquired knowledge do not suffice to compensate the higher time value of remanufacturing expenses when starting early.

Figure 4.11 shows the optimal development of λ_R .



Fig. 4.11. Optimal development of co-state λ_R in Example 4.3.

Example 4.4 For the fourth example out-of-pocket holding costs are reduced to $h_u = 0.3$. Therefore (**H**) conditions apply and it is beneficial to have a positive recoverables stock under conditions that characterize the optimal start time of remanufacturing in the previous example. Maximal Length Property (Proposition 4.7), Inventory Conditions (Proposition 4.9) and the Location Property like condition (4.33) are used to determine a sequence $4(3) \rightarrow 2(2) \rightarrow 2(1) \rightarrow 4(2)$. The optimal solution is represented by $\theta_{e,2(2)} = 1.2304, \theta_l = 3.9514$, and $\theta_{x,2(1)} = 4.3646$, and is illustrated in Figure 4.12. Optimal co-state developments are depicted in Figures 4.13 and 4.14.

Compared with Example 4.3 a smaller holding cost rate (or relatively higher initial remanufacturing costs) leads to the possibility to use a strategic recoverables stock in order to postpone the start time of remanufacturing θ_l . High initial remanufacturing expenses are pooled at and after that time when they are valued less than at the time when returns would have been remanufactured without stock-keeping. Moreover, a higher total number of returns is recovered because return collection starts earlier than remanufacturing in the case without stock-keeping.



Fig. 4.12. Optimal solution of Example 4.4.



Fig. 4.13. Optimal development of co-state λ_u in Example 4.4.



Fig. 4.14. Optimal development of co-state λ_R in Example 4.4.

4.6 Conclusions and Managerial Insights

In this chapter the effects of introducing a 'learning' remanufacturing process into the optimal control framework from Chapter 2 are investigated. The anticipation of later knowledge acquisition led to the possibility to remanufacture used products even if there exists no immediate cost advantage. More specifically when neglecting discounting, the Maximal Holding Time only depends on remanufacturing costs at the end of the planning horizon after all cost reductions due to learning have occurred. The decision on whether to remanufacture starting at time zero or not at all is made based on a simple break-even like condition for the total remanufacturing quantity during the planning period. In case of positive interest rate this process might start later than the begin of the planning period. The case with discounting is further characterized by two main additions to strategic stock-keeping. First, the maximal length of collection intervals increases with time and secondly, we discover another motivation for stock-keeping. Strategic inventory can be used to postpone the beginning of remanufacturing under conditions where initial remanufacturing costs are high enough that more interest is saved when delaying it compared to incurred holding costs.

Results are sensitive to planning horizon changes. This especially holds for small planning periods with considerable potential for cost-improvements after the end of the planning period. Since the learning curve concept is a strategic one, the planning horizon should be chosen sufficiently large. But used together with the product life cycle concept, this analysis can be a helpful tool for deciding whether to engage in remanufacturing at all.

Besides learning curve effects, there is a number of other reasons why cost parameters change over time. For instance, this might occur when varying prices for expensive raw materials lead to significant fluctuations of direct production costs. In addition to a reason for stock-keeping of these materials in anticipation of a price increase such a situation also leads to price dependent holding times for recoverables. Another example might be that due to the growing regulation in many industrialized countries intended to reduce the outcome of waste, using the disposal option becomes more and more expensive, influencing both stock-keeping and the decision when to start the remanufacturing process.

4.7 Proofs

Proof (Proof of $\lambda_0 = 1$).

This proof proceeds in a similar manner as its equivalent in Chapter 2. Let us assume $\lambda_0 = 0$. Then, the Lagrangian reduces to

$$L(..) = \lambda_u (u - r - w) + \lambda_R r + \mu_1 (d - r) + \mu_2 r + \mu_3 w + k_2 y_u, \quad (4.40)$$

and necessary conditions (4.14), (4.15), (4.19), and (4.20) change to

122 4 Knowledge Acquisition and Product Recovery

$$\frac{\partial L}{\partial r} = -\lambda_u + \lambda_R - \mu_1 + \mu_2 = 0 \tag{4.41}$$

$$\frac{\partial L}{\partial w} = -\lambda_u + \mu_3 = 0 \tag{4.42}$$

$$\dot{\lambda}_u = \alpha \lambda_u - \frac{\partial L}{\partial y_u} = \alpha \lambda_u - k_2 \tag{4.43}$$

$$\dot{\lambda}_R = \alpha \lambda_R - \frac{\partial L}{\partial R} = \alpha \lambda_R. \tag{4.44}$$

From (4.44) together with transversality condition (4.23) it follows that $\lambda_R = 0$ and $\dot{\lambda}_R = 0$. Non-triviality condition $(\lambda_0, \lambda_u, \lambda_R) \neq 0$ thus requires $\lambda_u \neq 0 \quad \forall t$. Non-negativity of μ_3 requires $\lambda_u > 0$ and $\mu_3 > 0$ in (4.42). Thus, w = 0 in (4.18). In order to achieve equality with zero in (4.41), $\mu_2 > 0$ must hold, which in turn requires r = 0. Thus $\dot{y}_u = u > 0$ which contradicts final inventory condition $(y_u(T) = 0)$.

Proof (Proof of Proposition 4.1). $y_u > 0$ requires $k_2 = 0$ from (4.21); (4.19) thus reduces to

$$\lambda_u = \alpha \lambda_u + h_u > 0. \tag{4.45}$$

w > 0 necessitates $\mu_3 = 0$ in (4.18) yielding $\lambda_u = -c_w$ in (4.15). It follows $\dot{\lambda}_u = 0$, which contradicts (4.45). Thus, $\lambda_u > -c_w$ and $w^* = 0$.

Three cases are possible for remanufacturing rate r:

- **2(1)** r = d (p = 0) requires $\mu_2 = 0$ in (4.17) yielding $\lambda_u < c_p c_r(R) + \lambda_R$ in (4.14). States and co-states develop as follows: $\dot{y}_u = u - d$, $\dot{R} = d$, and $\dot{\lambda}_R = \alpha \lambda_R + c'_r(R) \cdot d$.
- $0 < r < d \ (p = d r > 0)$ requires $\mu_1 = \mu_2 = 0$ in (4.16) and (4.17). Inserting into (4.14) yields

$$\lambda_u = c_p - c_r(R) + \lambda_R. \tag{4.46}$$

Differentiating both sides with respect to time leads to $\lambda_u = \alpha \lambda_R$. Together with (4.45) it follows $\lambda_R = \lambda_u + \frac{h_u}{\alpha}$. Re-inserting into (4.46) leads to

$$c_r(R) = c_p + \frac{h_U}{\alpha}.\tag{4.47}$$

Since the RHS of (4.47) is a constant, R must not change which contradicts the definition of this case $(\dot{R} = r > 0)$.

• **2(2)** r = 0 (p = d) requires $\mu_1 = 0$ in (4.16) yielding $\lambda_u > c_p - c_r(R) + \lambda_R$ in (4.14). States and co-states develop as follows: $\dot{y}_u = u > 0$, $\dot{R} = 0$, and $\dot{\lambda}_R = \alpha \lambda_R$.

Proof (Proof of Proposition 4.2).

Case 2 conditions $(y_u = 0)$ require p = d - r and w = u - r. Therefore, four situations with respect to the remanufacturing rate r are to be distinguished.

- $0 < r < \min\{d, u\}$ (p > 0, w > 0) requires $\mu_1 = \mu_2 = \mu_3 = 0$ in (4.16)– (4.18), leading to $\lambda_u = -c_w$ in (4.15). Therefore, $0 = c_p - c_r(R) + c_w + \lambda_R$ in (4.14). Differentiating RHS with respect to time yields $0 = \alpha \lambda_R$ which necessitates $\lambda_R = 0$ as well as $\dot{\lambda}_R = 0$. Inserting both values into (4.20) leads to $0 = c'_r(R) \cdot r$ which contradicts r > 0.
- 4(1) r = d < u (p = 0, w > 0) requires $\mu_2 = \mu_3 = 0$ in (4.17) and (4.18). This yields $\lambda_u = -c_w$ in (4.15), $\dot{\lambda}_u = 0$, as well as $k_2 = h_u - \alpha c_w > 0$ in (4.19). Inserting μ_2 and λ_u into (4.14) leads to $\lambda_R > c_r(R) - c_p - c_w$. Remaining optimal state and co-state movements are given by $\dot{\lambda}_R = \alpha \lambda_R + c'_r(R) \cdot d$ and $\dot{R} = d$.
- 4(2) $0 < r = u < d \ (p > 0, w = 0)$ requires $\mu_1 = \mu_2 = 0$ in (4.16) and (4.17). Inserting into (4.14) leads to

$$\lambda_u = c_p - c_r(R) + \lambda_R > -c_w. \tag{4.48}$$

Differentiating with respect to time leads to $\dot{\lambda}_u = \alpha \lambda_R$. Together with (4.19) it follows $k_2 = \alpha(\lambda_u - \lambda_R) + h_u$. Solving inequality (4.48) for λ_R yields $\lambda_R > c_r(R) - c_p - c_w$. State and co-states develop as follows: $\dot{\lambda}_R = \alpha \lambda_R + c'_r(R) \cdot u$ and $\dot{R} = u$.

• 4(3) r = 0 (p > 0, w > 0) requires $\mu_1 = \mu_3 = 0$ in (4.16) and (4.18). Proceeding as in the first subcase yields $\lambda_u = -c_w$ in (4.15), $\dot{\lambda}_u = 0$, $k_2 = h_u - \alpha c_w > 0$, as well as $\lambda_R < c_r(R) - c_p - c_w$. Optimal state and co-state movements are $\dot{\lambda}_R = \alpha \lambda_R$ and $\dot{R} = 0$.

Proof (Proof of Proposition 4.3).

Continuity of λ_R has already been established when rendering necessary optimality conditions (see Section 4.3). The proof of continuity of λ_u proceeds in two steps. In the first step (*i*) points inside intervals where $y_u = 0$ holds are examined. Afterwards, step (*ii*) deals with entry and exit points of such an interval.

(i) A constraint qualification guarantees the continuity of adjoint variable λ_u inside intervals where $y_u = 0$. Thus, continuity is given, if the matrix (with line numbers given on the right hand side)

$$\begin{pmatrix} -1 & 0 & d-r & 0 & 0 & 0 \\ 1 & 0 & 0 & r & 0 & 0 \\ 0 & 1 & 0 & 0 & w & 0 \\ -1 & -1 & 0 & 0 & 0 & y_u \end{pmatrix} \begin{matrix} I \\ III \\ IV \end{matrix}$$

has full rank of four. Analysis of the above matrix yields that the constraint qualification is not satisfied in three situations.

• If p = d - r = 0 and r = 0 then I = -II. This can only happen in Case 4 $(y_u = 0)$ in situations where d = 0 which is excluded by assumption (4.8) (d > 0).

- If r = 0, w = 0 and $y_u = 0$ then -II III = IV. This can only happen in Case 4 if u = 0 which again contradicts assumption (4.8) (u > 0).
- If p = d r = 0, w = 0, and $y_u = 0$ then I III = IV. This situation occurs in Case 4 ($y_u = 0$) when demand equals returns and the policy switches from p > 0, r > 0 to r > 0, w > 0 or vice versa. Only in this case we find a discontinuity λ_u . The height of this jump is $\eta_u = c_p + c_w - c_r(R) + \lambda_R$.

(ii) Let θ_u^1 be the entry time and θ_u^2 be the exit time of an interval, where $y_u = 0$ holds. Then, λ_u is continuous at this point if y_u enters this interval in a non-tangential way, i.e. $\dot{y}_u = u - r - w$ jumps. This requires a jump in r + w which might occur if d = u. Then, a tangential transition between cases 2(1) and 4(1) or 4(2) happens. The height of this jump is $\eta_u < c_p + c_w - c_r(R) + \lambda_R$. Any other tangential transition (between 2(1) and 4(3)) would require d = u = 0 which is excluded by assumption (4.8).

Proof (Proof of Corollary 4.1).

Since returns are accumulated but not used in Subcase 2(2) the only transition starting in such an interval is one that terminates into Subcase 2(1), where recoverables inventory can be depleted. Thus, any sequence of transitions between Subcases of Case 2 must terminate in a Subcase 2(1) interval. Both types of transition must take place continuously, and at transition time it must hold that

$$-c_w < \lambda_u = c_p - (c_r(R^*) - \lambda_R) \tag{4.49}$$

as well as $\dot{\lambda}_u = \alpha \lambda_u + h_u > \frac{d}{dt} (c_p - (c_r(R^*) - \lambda_R)) = \alpha \lambda_R$ for 2(1) \rightarrow 2(2) or $\dot{\lambda}_u = \alpha \lambda_u + h_u < \frac{d}{dt} (c_p - c_r(R^*) + \lambda_R) = \alpha \lambda_R$ for 2(2) \rightarrow 2(1), respectively. Inserting the value for λ_u as determined in (4.49) finally leads us to the following cost condition at transition time

$$h_u \begin{cases} \leq \alpha \left(c_p - c_r(R^*) \right) & \text{for } 2(1) \to 2(2) \\ > \alpha \left(c_p - c_r(R^*) \right) & \text{for } 2(2) \to 2(1) \end{cases}$$
(4.50)

It can easily be seen that since c_r is a decreasing function, a transition $2(1) \rightarrow 2(2)$ prevents from a later transition in the opposite direction, and that a transition therefore is not possible.

Transition $2(2) \rightarrow 2(1)$ takes place requiring condition 4.49 as well as high remanufacturing costs (**H**).

Proof (Proof of Corollary 4.2).

 $4(1) \rightarrow 4(2)$ would require an upward jump in λ_u except for a situation where $-c_w = \lambda_u = c_p - c_r(R^*) + \lambda_R$ implying $\lambda_R = c_r(R^*) - c_p - c_w$ which contradicts the definition of both cases in Proposition 4.2.

 $4(1) \rightarrow 4(3)$ and $4(2) \rightarrow 4(3)$ necessitate $\lambda_R = c_r(R^*) - c_p - c_w$ and

 $\dot{\lambda}_R < \dot{c}_r(R^*) \Leftrightarrow \frac{d}{dt} (\lambda_R - \dot{c}_r(R^*)) < 0$ at transition time. This contradicts (4.28).

 $4(2) \rightarrow 4(1)$ proceeds discontinuous and thus, it requires d = u.

 $4(3) \rightarrow 4(1)$ and $4(3) \rightarrow 4(2)$ are continuous and need $-c_w = \lambda_u = c_p - c_r(R^*) + \lambda_R \Leftrightarrow c_r(R^*) > c_r(R^*) - \lambda_R = c_p + c_w$. Moderate or high remanufacturing costs are necessitated (**M**,**H**). In the second case, at and after entering Subcase 4(2) it must hold that $k_2 = \alpha(\lambda_u - \lambda_R) + h_u \ge 0$. Thus, $h_u \ge \alpha(c_r(R^*) - c_p)$ only allowing for moderate remanufacturing costs (**M**).

Proof (Proof of Corollary 4.3).

Case 2 can only be left starting from a Subcase 2(1) interval. Transitions to Subcases 4(1) and 4(2) proceed under the same conditions as in the basic model without learning. $2(1) \rightarrow 4(3)$ would require a downward jump in λ_u . For a non-tangential transition, d = u = 0 are required, which contradicts assumption (4.8).

Transitions starting in Subcases 4(1) and 4(2) to 2(1) proceed under the same conditions as in the basic model without learning. 4(1) \rightarrow 2(2) requires $-c_w = \lambda_u = c_p - c_r(R^*) + \lambda_R \Leftrightarrow \lambda_R - c_r(R^*) = -c_w - c_p$. Since in Subcase 4(1) it holds that $\lambda_R - c_r(R^*) > -c_w - c_p$ and RHS of this inequality increases with rate $\alpha \lambda_R > 0$ (see (4.28)), the transition can be excluded.

 $4(2) \rightarrow 2(2)$ necessitates $-c_w < \lambda_u = c_p - c_r(R^*) + \lambda_R \Leftrightarrow \lambda_u - \lambda_R = c_p - c_r(R^*)$. At the end of the Subcase 4(2) interval it must hold that $k_2 = \alpha(\lambda_u - \lambda_R) + h_u \geq 0$. Inserting previously determined value for the co-state difference $\lambda_u - \lambda_R$ yields $h_u \geq \alpha(c_r(R^*) - c_p)$. In analogy to the proof of Corollary 4.1 this would exclude a later transition $2(2) \rightarrow 2(1)$ and thus contradict the zero final inventory level condition.

 $4(3) \rightarrow 2(2)$ does not require any special condition, but since Subcase 2(2) can only be left under high remanufacturing cost (**H**), it must also hold here. For a transition $4(3) \rightarrow 2(1)$, switching time is (a) determined by demand/return developments and at the same time (b) the value of learning must exactly equal the initial cost disadvantage of remanufacturing which also depends on demand and return development. Thus, the transition is a rare event which is not of practical interest and can be neglected in our discussion. In order to keep generality, it is further treated as a special case of a $4(3) \rightarrow 2(2) \rightarrow 2(1)$ with zero Case 2(2) length.

Proof (Proof of Proposition 4.4).

A zero interest rate excludes the case of high initial remanufacturing costs (\mathbf{H}) . All transitions requiring such circumstances are impossible. Therefore, Subcase 2(2) will not be present in an optimal solution.

Subcase 4(3) can not be left, because from Lemma 4.1 together with the definition of the case in Proposition 4.2 inside this case it holds that $\lambda_R = c_r(R_0) - c_r(R^*(T)) < c_r(R_0) - c_p - c_w$, and therefore $c_r(R^*(T)) > c_p + c_w$. Throughout the planning period it is not possible to obtain a positive recovery cost advantage and remanufacturing does not occur.

Proof (Proof of Proposition 4.5).

From (4.13), a minimal value for λ_u is given $(\lambda_u^{\min} = -c_w)$. For a Case 2(1) interval the upper bound is provided by Lemma 4.2 as follows $\lambda_u^{\max} = c_p - c_r(R^*(T))$. In the undiscounted case the development of λ_u is linear within a Case 2 interval which together with its continuity yields

$$-c_w \le \lambda_u(t) = \lambda_u(\theta_{e,2}) + h_u \cdot (t - \theta_{e,2}) \le c_p - c_r(R^*(T))$$

$$(4.51)$$

Let $t = \theta_{x,2}$ and $\lambda_u(\theta_{e,2}) = -c_w$. Solving (4.51) for $\theta_{x,2} - \theta_{e,2}$ yields

$$-c_w + h_u \cdot (\theta_{x,2} - \theta_{e,2}) \le c_p - c_r(R^*(T))$$
$$\Leftrightarrow (\theta_{x,2} - \theta_{e,2}) \le \frac{c_p + c_w - c_r(R^*(T))}{h_u}$$

Proof (Proof of Proposition 4.7).

The proof proceeds in the same way as the proof of Proposition 4.5 with $\lambda_u^{\min} = -c_w$, but now we have $\lambda_u^{\max}(t) = c_p - c_r(R^*(t) + \lambda_R(t))$. Solving differential equation (4.19) assuming $\lambda_u(\theta_{e,2}) = \lambda_u^{\min}$ leads to

$$\lambda_u(\theta_{x,2}) = \left(-c_w + \frac{h_u}{\alpha}\right) e^{\alpha(\theta_{x,2} - \theta_{e,2})} - \frac{h_u}{\alpha} \le c_p - c_r(R^*(\theta_{x,2})) + \lambda_R(\theta_{x,2}),$$
(4.52)

i.e. it must be smaller than $\lambda_u^{\max}(\theta_{x,2})$. Solving for $(\theta_{x,2} - \theta_{e,2})$ finally yields

$$(\theta_{x,2} - \theta_{e,2}) \le \frac{1}{\alpha} \ln \left(\frac{\alpha \left(c_p - c_r (R^*(\theta_{x,2})) + \lambda_R(\theta_{x,2}) \right) + h_u}{-\alpha c_w + h_u} \right).$$
(4.53)

From (4.28) we know $\dot{\lambda}_u^{\max}(t) = \alpha \lambda_R(t) \ge 0$. Thus the numerator in the logarithm expression in (4.53) increases and with it the whole term.

Proof (Proof of Proposition 4.8).Proposition 4.8 directly follows from Corollaries 4.2 and 4.4.

Technology Selection in the Context of Reverse Logistics

5.1 Motivation

When developing new products and setting up production facilities, firms often have the choice between different technologies in order to manufacture the product. Besides the quality and service aspects, this decision has a major impact on direct (variable) production costs and necessary capital expenditures in building and maintaining new facilities or modifying existing ones. In the context of reverse logistics, an additional issue has to be considered: tightened recycling and reuse legislation and environmental awareness of customers forces firms to take back their products from customers after use. At this point, the selected production technology also affects the ways on how to deal with returned/used products. That raises the question of whether to design and produce a product for single use only, or in a way that allows for reuse after some recovery process (e.g. rework, upgrading or remanufacturing, see de Brito and Dekker (2004) for an overview on available options). This can yield additional profit, as some of the added value will not be lost under certain recovery options, as it would be the case with material-recycling or disposing of the returned item. On the other side, there can be higher expenses for setting up production facilities, as well as higher direct production unit costs, that are caused by the necessity to add properties to the product in order to make it usable. This is illustrated by the following real-life example.

Case 5.1. CopyMagic (See Thierry et al. (1995) and Thierry (1997)) As a multinational copier manufacturing company, CopyMagic sells its products in all segments of the copier market, mainly by using leasing contracts. This creates a continuous flow of returned used products namely off-lease copiers. Depending on the returned product one or more of the following product recovery options is used: repair, cannibalization, remanufacturing and recycling. In particular the last two options require a special product design which is different from a 'classical' single use product which can only be disposed of after use. Design for recycling requires a reduction in the number

$\mathbf{5}$

of used materials, a replacement of non-recyclable with recyclable materials and easy separability of materials. Design requirements for remanufacturing go even further. Here it must be ensured that a product or its components are (in principle) capable to be used for more than one life time for the product to be sold 'as-good-as-new'. This causes higher production expenses and together with outlays for product recovery the question is which recovery option is preferable.

Although there exists a wealth of literature on operative issues in reverse logistics (for literature surveys see Fleischmann et al. (1997), Guide et al. (2000), or Fleischmann (2001b)), aspects of financial justification of product recovery, which are highly influential to investment decisions, have widely been neglected so far. Because they require both, a dynamic consideration as well as the application of discounted cash flow (DCF) techniques, which substantially complicates the matter, models usually deal with stationary situations and focus on average cost/profit. Debo et al. (2005) for instance consider the problem of technology selection in connection with market segmentation. Thereby, they assume a situation where remanufactured products are valued less than newly produced items, but both compete on the same market where customers have a heterogeneous willingness-to-pay. The chosen technology is characterized by a level of remanufacturability influencing variable production costs, but investment expenditures are not considered. The objective is to select a remanufacturability level which maximizes average profit under equilibrium conditions.

Durable products tend to remain with the customer for a considerable amount of time compared to the time period where they are sold. Demand may be subject to dynamic processes like the product life cycle and thus, a static (equilibrium) analysis based on average costs is often not appropriate. A dynamic DCF framework is required which also takes into account the time value of money and, especially, of investments. In the previous chapters it was assumed that both, production and remanufacturing facilities already existed, i.e. the choice upon remanufacturability and the selection of an appropriate technology has already been made. Since introducing a technology for remanufacturing may require additional investments, it remains to be seen if these investments will pay off or not. Furthermore, when considering more than one technology with different investments and variable unit costs, the optimal product recovery technology has to be determined. Consequently, a technology in our sense is described by four key characteristics - investment expenditure for setting up the production and remanufacturing facilities and variable costs for producing new items as well as remanufacturing returns.

Regarding the remanufacturing activities, it is seldom preferable to start it immediately when production starts, as this often requires a considerable capital commitment while not yet yielding a large number of returns. For instance in the case of engine remanufacturing in car industry (see, e.g., Seitz and Peattie (2004)) the usual life time of an engine must be considered while specialized equipment is required that sometimes must meet even higher standards than the one used for producing new engines. New production technologies like the use of aluminum instead of steel for engine components necessitate high investment in the remanufacturing process as well. Therefore, a decision has to be made when to introduce the process. This decision has to balance a trade-off which is given by the fact that postponing the investment reduces its time value whereas an earlier process introduction increases the potential benefit from remanufacturing.

Another question is whether to hold returned items in strategic inventory for later use or just to keep items that are necessary to satisfy current demand and therefore dispose excess returns. Practitioners often apply simplistic rules (e.g. collecting all returns and disposing of none, see Kiesmüller et al. (2004)). Both issues are related to the problem of timing a capacity expansion known from production/inventory theory (see, e.g., Slack and Lewis (2003)), where a (serviceables) inventory is used to postpone capacity expansion compared to a strategy without stockkeeping where capacity expansion must lead demand in order to avoid backorders. It is questionable whether recoverables inventory would influence the remanufacturing investment time in the same way.

This chapter is organized as follows. In Section 5.2, we propose a dynamic environment for strategic decision making in the context of reverse logistics based on simple assumptions regarding the product life cycle and an availability cycle for returns. Three investment projects representing different environmental policies are introduced in Section 5.3, and for each one the optimal policy parameters are determined. Main results of a numerical study, in which effects of strategic inventory are examined together with a comparison of simple heuristic rules with optimal strategies are presented in Section 5.4. Section 5.5 discusses the effects of introducing a remanufacturing constraint and the last section provides conclusions and further research possibilities.

5.2 A Dynamic Modeling Environment for Strategic Decision Making

In this section, we propose a simple generic environment for investment decisions for product recovery which integrates used product returns into a product life cycle development. First, we present the required assumptions on such a dynamic situation which are used in the optimization procedures of the next section. Later, we show that these structural properties hold for a specific modeling environment.

A Generic Dynamic Environment

In the following, we consider a demand/return scenario as illustrated in Figure 5.1 which complies with the following assumptions:

A.1 Demand d(t) is assumed to be a deterministic continuously differentiable function of time showing the typically unimodal shape of a product life cycle with its maximum located at $t_d^{max} > 0$. At the end demand must vanish, i.e. $\lim_{t\to\infty} d(t) = 0$.

A.2 Returns u(t) are not available prior to time point $\Delta > 0$ ($u(t) = 0 \forall t < \Delta$) and otherwise given by an unimodal function of time with $u(t) > 0 \forall t \ge \Delta$ representing the availability cycle of returns. Further, the return function has a maximum at t_u^{max} and is continuously differentiable for $t > \Delta$. As demand, returns finally vanish, i.e. $\lim_{t\to\infty} u(t) = 0$.

A.3 There exists at most a single intersection of demand and return functions $t_I \ge \max\{t_d^{max}, \Delta\}$ for which it holds u(t) < d(t) if $t < t_I$ and u(t) > d(t) if $t > t_I$. We thus presume that returns do not exceed demand during the growth phase of the product life cycle. If demand always exceeds the return rate there is no intersection, and t_I is set to infinity.

It is intuitively clear that at t_I the product life cycle has already entered its decline phase, i.e. demand is decreasing and intersects the return rate from above. Using Δ and t_I , one can distinguish between three different regions. In Region I, which ends at Δ , there are practically no returns accessible. Demand has to be filled completely by producing new items. Region II shows



Fig. 5.1. A generic environment complying with assumptions A.1-A.3.

less returns than demand (excess demand) and Region III is characterized by returns exceeding demand (excess returns) and decreasing demand. If the remanufacturing option is available, product returns can be used in both regions (II and III) to satisfy part or all of the demand. Moreover, in the last region it makes no sense to keep more than currently needed returns, because returns remain larger than demand until the end of the product's life cycle. Therefore, excess returns are to be disposed of.

A Specific Model

The demand process can be modeled by using one of the many different newproduct diffusion approaches which are quite common in marketing theory (for an overview see Mahajan et al. (1993)), but have also been introduced to operations management, for instance in combined forecasting and inventory models (see Kurawarwala and Matsuo (1996)). The most prominent example being applicable to durable goods has been developed by Bass (1969). According to Mahajan et al. (1993), this model and its revised forms have been proven to have a good predictive capabilities and have been successfully applied in retail service, industrial technology, pharmaceutical and consumerdurable markets. The Bass model uses three parameters, M > 0 is the number of potential adopters, representing the cumulative demand as time reaches infinity. P > 0 denotes the so-called coefficient of innovation and $Q \ge 0$ is the coefficient of immitation. All parameters are assumed to be known in advance. In the continuous variant of the model, demand at time t is given by

$$d(t) = \frac{MP(P+Q)^2 \cdot e^{-(P+Q)t}}{(P+Qe^{-(P+Q)t})^2}$$
(5.1)

This function comprises the demand of two kinds of adopters, namely innovators and immitators. These two groups differ in the reason why they are buying the product. An innovator's decision to buy the product is independent of the decision taken by others and it is purely induced externally by marketing techniques. Thus, a constant fraction P of the remaining potential customers' demand at time t is demanded by this group. This relation leads to a typically positive initial demand rate d(0) = MP > 0. In contrast to the first kind of adopters, immitators are more likely to buy the product the higher the proportion of potential customers who have already bought the product, i.e. they are said to be internally influenced. Demand rate d(t) reaches its maximum at the Point of Inflection of the typically S-shaped cumulative adoptions curve located at $t_d^{max} = -\frac{1}{P+Q} \ln (P/Q)$ (for P < Q, otherwise it is zero).

Forecasting of product returns becomes complicated because there is uncertainty with respect to time, quantity and quality of returns. Nevertheless, there exists a dependence of the return flow with historic demand data. De Brito and van der Laan (2002) give an overview on recent publications on the forecasting of product returns that exploit this connection by additionally using past return flow properties. Uncertainty with respect to time and quantity of returns can be reduced considerably when using leasing contracts that oblige the customer to give back the product after a fixed period $\Delta > 0$. But it is likely that not all products can be sold in this manner and some returns may not be in a condition to be recovered, only a fraction $F \in (0,1]$ of previous demands becomes available for remanufacturing. In the simplest case, both parameters can be assumed to be deterministic and known in advance. Thus, the return rate u(t) is given by

$$u(t) = \begin{cases} 0 & \text{for } t < \Delta \\ F \cdot d(t - \Delta) & \text{otherwise} \end{cases}$$
(5.2)

The return rate reaches its maximum at $t_u^{max} = t_d^{max} + \Delta$. A typical demand/return situation with parameters P = 0.028, Q = 0.25, and M = 100000, F = 0.4, and $\Delta = 5$ is depicted in Figure 5.2, both showing growth, maturity and decline phases of the respective life cycle. We now consider properties of the planning situation, exemplified by using demand and return functions as defined above. Proposition 5.1 discusses the relation between them.

Proposition 5.1. Under demand and return conditions (5.1) and (5.2), there exists at most one intersection point t_I of demand and returns. There is no intersection point, if



Fig. 5.2. Demands and returns in a specific model.

(i) $F < e^{-(P+Q)\Delta}$, or

 $(ii)F \cdot P \cdot M > d(\Delta).$

Proposition 5.1 states that there exist two settings where product life cycle and availability cycle of returns do not intersect. If (i) either the time lag Δ and/or the return fraction F are sufficiently small, returns will never exceed demand. In this case, t_I is defined to be infinity. If (ii) the time lag Δ is large and/or return fraction F is high enough, then returns at Δ , i.e. $u(\Delta) =$ $Fd(0) = F \cdot P \cdot M$, immediately exceed demand rate $d(\Delta)$. Here, t_I is defined to equal Δ .

5.3 Three Investment Projects

In this section we present three basic investment projects differing with respect to design and technology decisions. After introducing the respective optimal dynamic policies that describe how production, remanufacturing, and disposal decisions evolve over time, individual policy parameters are derived for each investment project. Finally, rules on how to determine the optimal parameters are deduced from minimizing the Net Present Value (NPV) of the respective series of payments within the planning horizon which for analytical convenience is set to infinity.

The following types of operational cash outflows are considered: investment expenditures for production and remanufacturing processes, constant production, remanufacturing and disposal per unit payments as well as outof-pocket inventory holding costs. Revenues, as well as payments connected with the take back of used products are not taken into account, since both demand as well as returns are assumed to be given and thus not subject of our considerations. Investment expenses include all discounted outlays for acquiring, maintaining, extending, and (possibly) salvage revenues for selling facilities with sufficient capacity. Therefore, restrictions on operative processes are supposed to never become binding.

Regarding the environmental policy of the firm and the existence of a strategic inventory that keeps returns for a later use, the following capital investment projects are considered:

(a) Design for single use

Products are designed in such a way that they can not be remanufactured. Thus, all returns have to be disposed of at costs c_w , which can be positive, if actual payments are necessary, or negative, if there is a positive salvage value. The corresponding investment into the production process at t = 0 is $K_p^s > 0$. Direct unit production costs are given by $c_p^s > 0$.

(b) Design for reuse

Products are designed such, that all returned products can serve as perfect substitutes to newly produced items after remanufacturing. This also implies
that products might be remanufactured several times, i.e. we assume unlimited (or at least sufficient) durability. See Geyer and Van Wassenhove (2005) for a discussion on the impacts of limited remanufacturability. Investment expenditures for setting up production at t = 0 amount to $K_p^r > K_p^s$, producing each item would cost $c_p^r > c_p^s$. Both values exceed the above expenses for single use design because of the additional requirements which are needed for a later remanufacturing. Returning used products can be remanufactured at unit costs $c_r > 0$ after introducing a remanufacturing process at time $t_r \ge \Delta$. This leads to a cash outlay of $K_r > 0$. In order to assure that this investment project constitutes a viable option a benefit must be realized if, instead of simultaneous production of a new item and disposal of the returned one, the latter is merely remanufactured, i.e. there is a positive direct recovery cost advantage

$$c_p^r + c_w - c_r > 0. (5.3)$$

Otherwise, no remanufacturing would take place. A possibility to store returns is not considered; all returns arriving before t_r must be disposed of.

(c) Design for reuse with strategic inventory

In addition to (b), it is now possible to keep returns for later use in inventory, e.g. in order to store items at a time where remanufacturing is not yet possible. The inventory level at t is denoted by $y_u(t)$. Out-of-pocket holding costs are assumed to be proportional to the time and quantity of used products on stock. The respective holding cost parameter is denoted by $h_u > 0$. A meaningful solution is assured if it is not advantageous to hold unneeded returned products as opposed to disposing of them, i.e.

$$h_u > \alpha c_w. \tag{5.4}$$

If this were not the case, disposal would not take place, because delaying the disposal of an item saves interests on the expense which would be higher than out-of-pocket costs incurred by holding the item.

5.3.1 Valuation of Investment Project (a) - Design for Single Use

When assuming a single use product, the optimal dynamic policy is obviously to dispose of all returns immediately upon receipt, i.e. $w^*(t) = u(t)$. Product requirements are satisfied by producing new items $(p^*(t) = d(t))$ as depicted in Figure 5.3 for the demand/return scenario known from Figure 5.2. This leads to the following expression for the Net Present Value

$$NPV_{a} = K_{p}^{s} + \int_{0}^{\infty} e^{-\alpha t} \left[c_{p}^{s} d(t) + c_{w} u(t) \right] dt.$$
 (5.5)

 NPV_a can be considered as a benchmark against which the financial benefit of the other investment projects have to be compared.



Fig. 5.3. Optimal decisions in investment project (a)

5.3.2 Investment Project (b) - Design for Reuse

Dynamic policy. The remanufacturing option is available after a capital expenditure of K_r at time t_r , subdividing the planning horizon into two parts. Before t_r , only production can be used to satisfy demand. Since returned items cannot be stored, the same policy applies as for investment project (a), i.e.

$$p^*(t) = d(t), \ r^*(t) = 0, \ w^*(t) = u(t) \quad \forall t < t_r.$$

The optimal policy for $t \ge t_r$ is to remanufacture as many units as possible, to produce excess demand (if necessary), and to dispose of remaining returns, yielding

$$p^{*}(t) = \max\{d(t) - u(t), 0\}, \ r^{*}(t) = \min\{d(t), u(t)\}, \ w^{*}(t) = \max\{u(t) - d(t), 0\} \\ \forall t \ge t_{r}.$$

The dynamic policy is shown in Figure 5.4.

Optimization of policy parameter. The Net Present Value of investment project (b) NPV_b depends on the investment time t_r , and minimizing it leads to the following non-linear optimization problem

$$\min_{t_r} NPV_b = K_p^r + \int_0^{t_r} e^{-\alpha t} \left[c_p^r d(t) + c_w u(t) \right] dt + e^{-\alpha t_r} K_r$$

$$+ \int_{t_r}^{\infty} e^{-\alpha t} \left[c_p^r \max\{d(t) - u(t), 0\} + c_r \min\{d(t), u(t)\} + c_w \max\{u(t) - d(t), 0\} \right] dt$$
(5.6)



Fig. 5.4. Optimal decisions in investment project (b)

Although the properties of this function depend to a certain extent on the underlying demand and return functions, it can be shown that based on the assumptions A.1-A.3 from Section 5.2, $NPV_b(t_r)$ is strictly decreasing and convex for $t_r < \Delta$, and there exists a point t after which it finally becomes a strictly decreasing and convex function. Depending on the parameters, there is either a single local minimum between Δ and t_I which is followed by a local maximum or the function is decreasing during the whole planning period $[0, \infty)$. The first derivative of $NPV_b(t_r)$ is a continuous function except for $t_r = \Delta$, given a jump discontinuity of the return function u(t) at Δ . Typical shapes of the objective and its first derivative for the demand/return scenario used in Figure 5.2 are depicted in Figure 5.5 for relatively high (i) and low (i) expenditures for the remanufacturing facility.

When choosing the investment time t_r , a trade-off has to be made between the lower discounted value of investment expenses K_r if the introduction of the remanufacturing process is postponed and a larger realized recovery cost advantage if it is placed earlier. Reconsidering the dynamic environment introduced in Section 5.2 and exemplified in Figure 5.4, it is easy to see that in Region I, where no returns are yet available, changing t_r only affects the discounted value of expenses required for the recovery investment. Therefore, postponing it would always be preferable. Analogously for Region III, as time goes to infinity, the demand rate approaches zero, and with it the potential



Fig. 5.5. Typical shape of the objective function and its first derivative in a demand/return scenario with (i) high and (ii) low investment expenditures.

current benefit of remanufacturing diminishes and again, a delay (this time until infinity) of the recovery investment is favorable. This consideration leads to local properties of the optimal investment time t_r^* (given it is finite and thus, located in Region II) as expressed by Proposition 5.2 which is derived by exploiting the first and second order derivatives of the objective (5.6).

Proposition 5.2. If it exists, a finite investment time t_r^* must be located within the half open interval $[\Delta, \min\{t_u^{max}, t_I\})$, and one of the following situations must apply

(i)
$$u(t_r^*)(c_p^r + c_w - c_r) = \alpha K_r$$
 and $\dot{u}(t_r^*) > 0$ if $t_r^* > \Delta$,
(ii) $u(t_r^*)(c_p^r + c_w - c_r) \ge \alpha K_r$ if $t_r^* = \Delta$

Proposition 5.2 states that at t_r^* , the current cost advantage of remanufacturing $u(t_r^*)(c_p^r + c_w - c_r)$ must at least earn interests on the investment, i.e. αK_r . If (i) equality holds, the remanufacturing rate must increase at t_r^* to start recouping the investment expenses. This is not possible at or later than either the time where returns reach their maximum t_u^{max} or the intersection time of demand and return rate t_I . Because in both cases the remanufacturing rate no longer increases. Thus, investment time t_r^* must lie in an interval given by $[\Delta, \min\{t_u^{max}, t_I\})$. As a special case *(ii)*, at time Δ there may be a jump in the return rate (and consequently in the potential remanufacturing rate) from $u(\Delta^-) = 0$ up to a point $u(\Delta^+) > 0$ where a higher recovery cost advantage may be realized than interests on investment expenses require. This leads to a different condition in Proposition 5.2. Both situations are sketched in Figure 5.5.

Resorting the first order condition in Proposition 5.2 leads to a critical value for the return rate u^{crit}

$$u^{crit} = \frac{\alpha K_r}{c_p^r + c_w - c_r} \tag{5.7}$$

which can be interpreted as a dynamic break-even point. If the investment into a remanufacturing facility takes place at all, then it is placed at a point in time where an increasing return rate surpasses u^{crit} as seen in Figure 5.6.



Fig. 5.6. Return rate surpassing u^{crit} at optimal final investment time t_r^* .

Since the remanufacturing rate is limited to the available returns, a finite investment time t_r^* does not exist if the return rate never exceeds the critical return rate u^{crit} in Region II. In such a case, the initial interest rate on the investment exceeds the maximum possible current cost advantage of remanufacturing (to be earned at $\min\{t_u^{max}, t_I\}$) as stated in the following corollary.

Corollary 5.1. There exists no finite optimal investment time t_r^* , if

$$u(\min\{t_u^{max}, t_I\}) \le u^{crit} \iff u(\min\{t_u^{max}, t_I\})(c_p^r + c_w - c_r) \le \alpha K_r.$$
(5.8)

So far, only local conditions ensuring the possibility to *start* recouping the investment expenditures required for the remanufacturing facility have been considered. Of course, these investments have to be paid off *completely*. If the expenses are rather high, even in the optimal case, the cumulative cost advantage may not be sufficient for amortization. In this case, the optimal investment time is infinity. By comparing the values of the objective function of the finite candidate satisfying Proposition 5.2 with its limit as time approaching infinity, the following sufficient (global) condition for optimality of a finite investment time t_r^* must hold.

Proposition 5.3. For the optimal investment time t_r^* it must hold that the total realized advantage of remanufacturing discounted to t_r^* at least equals the expenses needed for setting up the remanufacturing facility

$$K_r \le \int_{t_r^*}^{\infty} e^{-\alpha(t-t_r^*)} \left[c_p^r + c_w - c_r \right] \min\{d(t), u(t)\} dt \,.$$
(5.9)

Comparison with Investment project (a). There exists no simple rule for determining the best of the two investment projects, but by comparing (optimal) Net Present Values it can be stated that investment project (b) is preferable to (a) if the total discounted net advantage of remanufacturing $A_r^b \geq 0$

$$A_r^b = \int_{t_r^*}^{\infty} e^{-\alpha t} (c_p^r + c_w - c_r) \min\{d(t), u(t)\} dt - e^{-\alpha t_r^*} K_r$$
(5.10)

exceeds the increase of the total discounted expenditures for the production process $D_p^b > 0$

$$D_{p}^{b} = \left(K_{p}^{r} - K_{p}^{s}\right) + \left(c_{p}^{r} - c_{p}^{s}\right) \int_{0}^{\infty} e^{-\alpha t} d(t) dt.$$
(5.11)

It is easy to see that if there is no finite optimal investment time for investment project (b), then $A_r^b = 0$ holds and (a) should be chosen. If A_r^b is positive, then preferability of the design for reuse depends on the increase of initial investments in the production process compared with a single use production as well as on the increase of direct production costs.

5.3.3 Investment project (c) - Design for Reuse with Strategic Inventory

Dynamic policy. As an extension to investment project (b), returns can be stored in a recoverables inventory. As such, the problem becomes truly dynamic, because the decision to store a returned item influences future possibilities of remanufacturing. In analogy to (b) and given a value for t_r and for the system's state at this time $y_u(t_r)$, the planning horizon can be subdivided into two parts. Prior to t_r , the question arises when to start collecting returns in order to achieve the desired stock. Obviously, it is not useful to dispose of returns during the collection period, because otherwise one could have started gathering later and thus, saved holding costs. Therefore, to each value $y_u(t_r)$ a corresponding time point $t_e \leq t_r$ can be given where disposal stops and all returns are put to stock, being defined by the following equation

$$\int_{t_e}^{t_r} u(s) ds = y_u(t_r).$$
(5.12)

Since stock-keeping can start earliest at Δ , i.e. $t_e \geq \Delta$, the maximum possible quantity on stock at t_r is given by

$$y_u(t_r) \le \int_{\Delta}^{t_r} u(s) ds.$$
(5.13)

Thus, optimal decisions in the first part are given by

$$p^{*}(t) = d(t), r^{*}(t) = 0, w^{*}(t) = u(t) \quad \forall \ t < t_{e},$$

$$p^{*}(t) = d(t), r^{*}(t) = 0, w^{*}(t) = 0 \qquad \forall \ t_{e} \le t < t_{r}.$$

The optimal solution of the second part is derived by using results of the basic model presented in Chapter 2. First, the recoverables inventory is depleted by filling excess demand d(t) - u(t) from remanufacturing stored returns. This is completed at a time $t_x \ge t_r$, given by

$$\int_{t_r}^{t_x} \left(d(s) - u(s) \right) ds = y_u(t_r).$$
(5.14)

Completion time t_x must not be larger than t_I because afterwards returns always exceed the demand rate and carrying an inventory is no longer necessary. Beside (5.13), this gives another condition for $y_u(t_r)$

$$y_u(t_r) \le \int_{t_r}^{t_I} (d(s) - u(s)) \, ds.$$
 (5.15)

After t_x , the same policy is used as in investment project (b) because it is not useful to build up stock again, yielding the following optimal decisions in the second part

$$p^{*}(t) = 0, r^{*}(t) = d(t), w^{*}(t) = 0 \qquad \forall \ t_{r} \le t < t_{x},$$

$$p^{*}(t) = \max\{d(t) - u(t), 0\}, r^{*}(t) = \min\{d(t), u(t)\}, w^{*}(t) = \max\{u(t) - d(t), 0\}$$

$$\forall t \ge t_{x}.$$

The dynamic policy in investment project (c) is depicted in Figure 5.7. Of course, this policy requires sufficient capacity and a high flexibility both in the production as well as in remanufacturing process. This is for instance the case if workers that normally are employed to produce new items, can easily be switched to remanufacture used products. In the presence of capacity constraints, a more complex model would be necessary for which only numerical results can be obtained. See Section 5.5.



Fig. 5.7. Optimal decisions in investment project (c)

Optimization of policy parameters. In contrast to investment project (b), the optimal solution to policy class (c) not only consists of investment time t_r^* , but additionally the corresponding recoverables stock $y_u^*(t_r^*)$ has to be determined. Equivalently, and probably even more interesting than the actual stock value, the starting time of collecting returns t_e^* will be determined. In the following, we restrict ourselves to find the best finite solution, i.e. $\Delta \leq t_e < t_r < t_I$. Since the option of investing never, i.e. $t_e = t_r = \infty$, still belongs to the set of solution candidates, it has to be considered in order to find the global optimum. Note that we disregard demand and return constraints as given by Proposition 5.1 (*ii*) (see Section 5.2) in the following, because under such circumstances (returns exceed demands immediately after Δ) it is obviously not optimal to hold recoverables for later use.

Restricting ourselves to finite solution candidates, we get the following optimization problem (5.16)-(5.21).

$$\min_{t_{e}, t_{r}} NPV_{c} = K_{p}^{r} + \int_{0}^{t_{e}} e^{-\alpha t} [c_{p}^{r}d(t) + c_{w}u(t)] dt$$

$$+ \int_{t_{e}}^{t_{r}} e^{-\alpha t} [c_{p}^{r}d(t) + h_{u}y_{u}(t)] dt + e^{-\alpha t_{r}}K_{r} + \int_{t_{r}}^{t_{x}} e^{-\alpha t} [c_{r}d(t) + h_{u}y_{u}(t)] dt$$

$$+ \int_{t_{x}}^{\infty} e^{-\alpha t} [c_{p}^{r}\max\{d(t) - u(t), 0\} + c_{r}\min\{d(t), u(t)\} + c_{w}\max\{u(t) - d(t), 0\}] dt$$
(5.16)

with

142 5 Technology Selection in the Context of Reverse Logistics

$$y_u(t;t_e,t_r,t_x) = \begin{cases} \int_{t_e}^{t} u(s)ds & \text{for } t \in [t_e,t_r] \\ \int_{t_e}^{t_r} u(s)ds - \int_{t_r}^{t} (d(s) - u(s))ds & \text{for } t \in (t_r,t_x] \\ 0 & \text{otherwise} \end{cases}$$
(5.17)

and t_x being implicitly defined by a function $f(t_e, t_r, t_x)$

$$t_x: f(t_e, t_r, t_x) = \int_{t_e}^{t_r} u(s)ds - \int_{t_r}^{t_x} (d(s) - u(s))ds = 0$$
(5.18)

subject to the restrictions

$$\Delta \le t_e, \tag{5.19}$$

$$t_e \le t_r,\tag{5.20}$$

$$\int_{t_x}^{t_I} \left(d(s) - u(s) \right) ds \ge 0.$$
(5.21)

The objective function (5.16) incorporates all payments connected with the optimal policies in each of the above distinguished regions. Function (5.17) is used to determine the inventory level and (5.18) gives an implicit definition of point t_x where the inventory is depleted. Constraint (5.19) ensures continuity of the objective by limiting the admissible set and (5.20) is needed in order to assure a meaningful solution. Restriction (5.21) is equivalent to $t_x \leq t_I$ but technically it is easier to handle. This inequality represents remaining excess demand between t_x and t_I , which must be non-negative.

Due to the quite general assumptions on demand and return functions, objective (5.16) is neither a (quasi-)convex function in each nor in both decision parameters. Moreover, the admissible region is not convex because of our general assumptions on demand and return developments in (5.21). Hence, conditions derived below by using standard methods of non-linear programming are only necessary for optimality. As a consequence, a solution candidate can represent a local minimum, maximum, or a saddle point. Further, there may be several solution candidates for a single problem instance, in order to find the optimal solution the respective objective values need to be compared.

In the following, four different types of solution candidates are distinguished. For ease of representation, a candidate is given by a triplet (t_e, t_r, t_x) , bearing in mind that t_x is a function of the other two points.

Proposition 5.4 (Solution Candidates). If (t_e^*, t_r^*, t_x^*) is an optimal solution to problem (5.16)-(5.21), $t_e^* < t_r^* (< t_x^*)$ must hold, and one of the following four cases applies

(i) $\Delta < t_e^*, t_x^* < t_I$ (interior solution) (ii) $\Delta = t_e^*, t_x^* = t_I$ (complete use of interval $[\Delta, t_I]$) (iii) $\Delta = t_e^*, t_x^* < t_I$ (availability of returns is binding restriction) (iv) $\Delta < t_e^*, t_x^* = t_I$ (availability of excess demand is binding restriction)

Figure 5.8 shows all cases, for which candidates for the optimal solution can be determined. Now, by exploiting first order necessary conditions, properties



Fig. 5.8. Cases representing candidates for the optimal solution.

of these cases are discussed. Firstly, a general condition regarding the optimal holding time $t_x^* - t_e^*$ can be given.

Proposition 5.5 (Maximal Holding Time). If (t_e^*, t_r^*, t_x^*) is an optimal solution to problem (5.16)-(5.21), it must hold that $t_x^* - t_e^*$ does not exceed a maximal holding time τ_u , i.e.

$$t_x^* - t_e^* \le \frac{1}{\alpha} \ln\left(\frac{\alpha(c_p^r - c_r) + h_u}{-\alpha c_w + h_u}\right) =: \tau_u.$$
(5.22)

Maximal holding time τ_u as defined in Proposition 5.5 comprises the same marginal criterion known from the basic model (see Section 2.3) which balances the cost advantage of storing an otherwise disposed item between t_e^* and t_x^* in order to replace production by remanufacturing at t_x^* and the required holding costs.

The following Propositions 5.6-5.9 present results for each of the cases.

Proposition 5.6 (Case (i) - interior solution). A triplet (t_e, t_r, t_x) with $\Delta < t_e < t_r < t_x < t_I$ is a solution candidate to problem (5.16)-(5.21) of Case (i), if it satisfies the following equations

$$t_x - t_e = \tau_u, \tag{5.23}$$

$$-e^{-\alpha t_r}[c_p^r - c_r]d(t_r) = h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_x}[c_p^r - c_r]d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.24)

Equations (5.23) and (5.24) follow from setting the first derivatives of the objective (5.16) to zero and can be interpreted as follows. Since $t_x - t_e$ equals the maximal holding time τ_u , the decision maker must be indifferent between (1) disposing of a (marginal) return unit arriving at t_e , and producing a new one to meet demand at t_x or (2) holding this item until t_x when it

is remanufactured to serve demand. Next, at t_r one needs to be indifferent between starting the remanufacturing process and thereby realizing the direct cost advantage of remanufacturing immediately or to postpone it which saves interests on the investment expenses. Then, a (marginal) demand $d(t_r)$ is served from producing new items and the thus saved (marginal) return is kept until t_x which results in holding costs and lowers the discounted value of the direct remanufacturing cost advantage.

Using (5.23) together with the definition of the case requires

$$t_I - \Delta > \tau_u, \tag{5.25}$$

which has to be assured first in order to find a Case (i) solution candidate. Then, simultaneously solving (5.18), (5.23) and (5.24) for (t_e, t_r, t_x) yields the candidate, given it exists.

Proposition 5.7 (Case (ii) - complete use of interval $[\Delta, t_I]$). A triplet (t_e, t_r, t_x) with $\Delta = t_e < t_r < t_x = t_I$ is a solution candidate to problem (5.16)-(5.21) of Case (ii), if the following conditions are satisfied

$$e^{-\alpha\Delta}c_w d(t_r) \ge h_u \int_{\Delta}^{t_r} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_r} [c_p^r - c_r] d(t_r) + \alpha e^{-\alpha t_r} K_r, \quad (5.26)$$

$$-e^{-\alpha t_r}[c_p^r - c_r]d(t_r) \ge h_u \int_{t_r}^{t_I} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_I}[c_p^r - c_r]d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.27)

Inequality (5.26) implies that it would be preferable to put additional returns in stock at t_e for use at t_r by simultaneously lowering t_e and t_r , even at the cost of an earlier investment. But this is not possible because $\Delta = t_e$. Likewise, using (5.27), the value of the objective could be lowered by postponing investment time t_r . This is also forbidden because we would need to increase $t_x = t_I$ which again is not possible.

A Case (ii) candidate may only exist, if

$$t_I - \Delta \le \tau_u. \tag{5.28}$$

Thus, it is not possible to have a planning situation where we could obtain solution candidates in both Case (i) and Case (ii) simultaneously. If the Case (ii) pre-requirement is fulfilled, from (5.18) one gets a value for t_r which is verified if (5.26) and (5.27) hold.

Proposition 5.8 (Case (*iii***) - availability of returns is binding restriction).** A triplet (t_e, t_r, t_x) with $\Delta = t_e < t_r < t_x < t_I$ is a solution candidate to problem (5.16)-(5.21) of Case (*iii*), if the following conditions are satisfied

$$t_x - \Delta \le \tau_u, \tag{5.29}$$

$$-e^{-\alpha t_r}[c_p^r - c_r]d(t_r) = h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_x}[c_p^r - c_r]d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.30)

From (5.29) we know that maximal holding time is not yet reached. But, in contrast to Case (*ii*), from (5.30) we are indifferent regarding the postponement of t_r . Placing it earlier is also not possible, because t_e is fixed to Δ . A Case (*iii*) candidate is to be found by simultaneously solving equations (5.18) and (5.30) for t_r and t_x by assuming $t_e = \Delta$. The result is a solution candidate if inequality (5.29) is satisfied.

Proposition 5.9 (Case (*iv***) - availability of excess demand is binding restriction).** A triplet (t_e, t_r, t_x) with $\Delta < t_e < t_r < t_x = t_I$ is a solution candidate to problem (5.16)-(5.21) of Case (*iv*), if the following conditions are satisfied

$$t_I - t_e \le \tau_u,\tag{5.31}$$

$$e^{-\alpha t_e} c_w d(t_r) = h_u \int_{t_e}^{t_r} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_r} [c_p^r - c_r] d(t_r) + \alpha e^{-\alpha t_r} K_r.$$
(5.32)

As before, but with t_x fixed to t_I , (5.31) implies to decrease t_e which is not possible without changing t_r . Choosing t_r requires indifference between disposing a (marginal) returned item at t_e or using it to lower t_r which in turn causes an increase in associated holding and interest expenses due to sooner investment but it also replaces production by remanufacturing at t_r . The determination of a Case (*iv*) candidate requires to simultaneously solve equations (5.18) and (5.32) for t_e and t_r assuming $t_x = t_I$. The result is a solution candidate if inequality (5.31) is satisfied.

Algorithm 5.1

Step 1

if $t_I - \Delta > \tau_u$

then Simultaneously solve (5.23), (5.24), and (5.18) for (t_e, t_r, t_x) . add result to set of candidates.

else Set $t_e = \Delta$, $t_x = t_I$ and solve (5.18) for t_r .

add result to set of candidates, if (5.26) and (5.27) are satisfied.

set $t_e = \Delta$, and simultaneously solve (5.30) and (5.18) for t_r and t_x and add result to set of candidates, if (5.29) is satisfied.

set $t_x = t_I$, and simultaneously solve (5.32) and (5.18) for t_e and t_r and add result to set of candidates, if (5.31) is satisfied.

add $(t_e, t_r, t_x) = (\infty, \infty, \infty)$ to set of candidates.

Step 2

For all candidates, evaluate the objective value (5.16). Smallest value gives the (global) optimum.

Comparison with Investment project (b). Investment project (c) generally leads to a lower Net Present Value than (b) because it is a generalization of (b) which uses a strategic inventory to maximize the benefit from replacing production by remanufacturing. Another interesting question is how the possibility to hold returns for later use affects investment time t_r . Unfortunately, there is no general answer. A scenario that allows for postponing the investment time is that it no longer has a direct effect on the remanufacturability of returns since these also can be put in stock and remanufactured later. Other aspects make it possible to start remanufacturing earlier, e.g. a higher direct cost advantage of remanufacturing can be realized at t_r because demand is sourced completely from remanufacturing returns. In order to gain more insight into this question we conducted a numerical investigation presented in the next section.

5.4 A Numerical Investigation

The purpose of this study is threefold. First, a pre-test should show that all types of (finite) solution candidates of investment project (c) as presented in Proposition 5.4 are relevant for determining the optimal solution. Further, an assessment of the potential benefit derived from permitting stock-keeping had to be performed and finally, the influence of a strategic recoverables inventory on the investment time t_r was assessed. A second test was used to clarify, under which conditions simple heuristic rules relevant for a practical application perform sufficiently well. Since the effects of changes in the interacting parameters are manifold, we decided to perform the study based on a large number of randomly generated examples.

In this pre-test study we used a demand function according to the Bass model (see Section 5.2) with parameters P = 0.01, Q = 0.3, $M = 100\,000$. Since we did not have real-live data, the parameters for each one of 30 000 instances were generated from uniform probability distributions over each of the following ranges, partly including extreme values: Range(F) = [0; 1], $Range(\Delta) = [0; 12]$, $Range(\alpha) = [0.05; 0.15]$, $Range(c_p^r) = [4; 5]$, $Range(c_r) =$ [1; 3], $Range(c_w) = [-1; 1]$, $Range(K_r) = [20\,000; 100\,000]$ and $Range(h_u) =$ [0; 1]. Since the cash outlays for setting up the production facility were not relevant for our comparison, we set K_n^r equal to zero.

In total, 10658 examples showed finite optimal solutions for investment project (c) according to Cases (i)-(iv). Of these, 4685 (44.0%) belonged to Case (i), 90 (0.8%) to Case (ii), 5872 (55.1%) to Case (iii), and only 11 (0.1%) examples were Case (iv) solutions. Although all cases are relevant for determining the optimal solution these numbers show that in more than half of all considered instances it was optimal to immediately start storing returns. Less than half of the instances showed an interior solution.

Next, by comparing the optimal objective values for investment projects (b) and (c), we found that the benefit from keeping stock averaged to about 2% but the maximum difference was more than 11%, found in a scenario with the following parameters: F = 0.63, $\Delta = 0.60$, $\alpha = 0.14$, $c_p^r = 4.23$, $c_r = 1.07$, $c_w = 0.93$, $K_r = 91\,628$, $h_u = 0.16$. Thus, taking into account that aside of operational expenses also investment expenditures where savings amount to a remarkable amount of money.

Regarding the investment time t_r , the results indicate that it is usually (i.e. in 81.7% of all considered examples) postponed due to the strategic inventory, except for cases where the optimal investment time is infinite in investment project (b) but finite in (c) because of the additional benefit from storing returns. This happened for 15.7% of all examples. But 280 instances (2.6%) including all Cases (i)-(iv) exhibited the opposite behavior. Particularly noteworthy, all examples where the availability of excess demand was a binding restriction (Case (iv)) exhibited an earlier investment time when allowing for stockkeeping. This result was confirmed by another 2 000 instances, which were generated in order to increase the number of Case (iv) solutions, where we changed the ranges for the following parameters: Range(F) = [0.8; 1], $Range(\Delta) = [0; 2]$, $Range(\alpha) = [0.1; 0.15]$, $Range(K_r) = [60\,000; 100\,000]$, $Range(h_u) = [0.1; 0.3]$. Thereby, 161 (11.5%) out of 1396 finite optimal solutions were of Case (iv).

In a second test we compared the performance of four simple heuristic rules, which are described in the following. While the first two neglect the possibility of keeping stock and just try to select an appropriate investment time, H3 and H4 use more or less sophisticated methods to control storing of returns for later use by following the most common investment project (c) cases identified before. Since the change of the optimal investment time when allowing for stock-keeping averaged to just about 1.1 in the pre-test, all heuristics except of the first use as investment time t_r the value which is optimal for investment project (b). The heuristics are now explained in more detail.

H1 The remanufacturing facility is set up as soon as the first returns arrive, i.e. at time $t_r = \Delta$. Thus, there is no need for stock-keeping. This heuristic neglects the decreasing time value of investment expenses due to discounting and is expected to perform well if investment expenditures for the remanufacturing facility are low or discount rate is small.

H2 The optimal solution of investment project (b) is used as a second heuristic. It should lead to good results in circumstances where the Maximal Holding Time is relatively small, e.g. where out-of-pocket holding costs are large.

H3 The third heuristic combines H2 with a simple rule with respect to stock keeping. Returns are kept starting at time Δ and used up after t_r . If there are any items left on stock at t_I , these are disposed of. This heuristic approximately corresponds to a Case (*iii*) solution candidate and might perform well if out-of-pocket holding costs are low because it disregards a possible limitation of stock-keeping in time.

H4 In contrast to H3, this last heuristic only keeps stock that can be used up before t_I and stored no longer than the Maximal Holding Time. Thus, building up the anticipation inventory may start later than Δ leading to a Case (i) candidate like solution. Since this heuristic considers both, holding time and the time value of investment expenses, its performance is expected to be superior to that of the other heuristics. The heuristic rules are tested for a collection of classes of randomly generated instances. The variety is based on a single demand and a number of return scenarios, as well as on levels of key parameters, namely direct recovery cost advantage, discount rate, out-of-pocket holding cost rate, and investment expenditures for setting up the remanufacturing facility. For each of these parameters two ranges representing comparably high and low values have been defined. More precisely, we used the following experimental design.

- Demand function is fixed as in the pre-test $(P = 0.01, Q = 0.3, M = 100\,000)$.
- Four return scenarios are used with a low/high return fraction F and small/large duration of use Δ as depicted in Table 5.1. Since after t_I not

Scenario	F	Delta	t_I	Total returns	Usable returns
Ι	0.4	3	28.4	40000	39982
II	0.4	6	17.7	40000	33059
III	0.7	3	15.1	70000	62562
IV	0.7	6	15.3	70000	46063

Table 5.1. Four considered return scenarios

all returns can be used, the last column in Table 5.1 shows the maximal usable number of returns which better expresses the potential benefit from remanufacturing.

- As in the pre-test, initial investment expenditures into the remanufacturing facilities were set to zero $(K_p^r = 0)$. The difference $c_p^r c_r$ is normalized to 1. Objective values are calculated by using $c_p^r = 1$ and $c_r = 0$.
- Since the other parameters $(c_p^r \text{ and } c_r)$ are fixed, the recovery cost advantage only depends on the disposal cost rate c_w . Two intervals are considered, one with a relative low disposal costs, i.e. $c_w \in (-1,0)$, and another one with a comparably high cost rate $c_w \in (0,1)$. In the first case, the direct recovery cost advantage ranges between 0 and 1, and in the second it is in an interval between 1 and 2.
- In order to find a possible impact of discounting, α is assumed to belong to one out of two intervals, being either low ($\alpha \in (0.05, 0.1)$) or high ($\alpha \in (0.1, 0.15)$).
- Holding cost rate h_u was assumed to be taken either out of an interval with a relative low level, i.e. $h_u \in (0, 0.25)$, or from another with comparably high level $h_u \in (0.25, 0.5)$. Hereby it is ensured that only values are used which satisfy assumption (5.4).
- For the investment expenditures K_r we chose the following two ranges: $K_r \in (0, 40\,000)$ and $K_r \in (40\,000, 80\,000)$. The upper border is motivated by the fact that because of discounting, in case of a highest possible recovery cost advantage a and lowest possible discount rate, this number

represents maximal investment expenditures that can be earned from remanufacturing in Scenario III where most usable returns are present.

In total, there have been 64 combinations of scenario and parameter intervals which correspond to a certain setting (4x2x2x2x2 factorial design). Since we fixed some of the parameters, we could suffice with only 200 examples for each setting (12 800 in total) yielding enough material for statistical tests. For each example the relative errors of the heuristic (H1-H4) objective values were calculated. In order to do a fair comparison, only those examples were considered, under which remanufacturing actually would be useful, i.e. where a finite investment project (c) solution is optimal. With other words, it is assumed that the decision maker is able to decide whether remanufacturing actually makes sense or not, and the only concern is ascertaining at which time to invest and whether and when to start collecting returns. By appropriately grouping the examples, a sensitivity analysis of the average performance of the heuristics with respect to return scenarios as well as the examined key parameters was performed.

This analysis was complemented by statistical tests which ensured the comparison of average performance of the heuristics one against each other, where the results originated from the same experiments (matched pairs), but the change of the heuristics performance due to different settings (independent group means), two different types of tests had to be performed. In the first case, a paired t-test was carried out which, because of the large sample sizes, was approximated by a Normal z Test. For comparing independent group means w.r.t. the same heuristic in different settings, a single-sided version of the approximative two groups Normal z Test was performed. Because of the large sample size, the significance was tested on a 99% level. For a detailed treatment of the tests performed see Section 5.7. All comparisons of average errors are significant except where stated otherwise. In spite of this procedure, since it is not possible to generate a general setting which integrates all possible demand/return situations and cash flow parameter combinations, all following statements should rather be seen to express tendencies, which should be verified before applying to an actual situation.

The main results of the study are presented in Tables 5.2-5.6 showing average and maximal relative errors of the heuristic solutions with respect to the objective. Here also the fraction of finite optimal investment project (c) solutions in the respective subset of all experiments can be found, represented by the sign #. This number expresses a relationship between the setting and an average profitability of remanufacturing. The variability of the performances of the heuristics is presented in Section 5.7.

Overall results and scenario comparison. Considering all examples (see last row in Table 5.2), H4 performs best in terms of the average error as previously expected. It also shows the smallest maximal deviation from the optimal solution. Also not unexpectedly, H1 performs worse than all other heuristics in both criteria. Especially when comparing the performance with H2, a large benefit can be obtained only by postponing the investment time. Introducing a simple rule for stock-keeping (H3 instead of H2) yields in average an additional benefit, but under circumstances described below it can also lead to a substantial performance loss.

1			I	I1	I	H2	H	I3	I	I4
	Scenario	#	avg.	max.	avg.	max.	avg.	max.	avg.	max.
	Ι	27.8	8.9	49.2	3.1	13.6	1.5	13.7	1.2	13.6
	II	25.1	5.3	27.7	2.1	8.1	1.0	6.5	0.8	6.5
	III	44.4	11.7	59.0	4.2	15.9	2.1	33.3	1.4	14.5
	IV	35.5	6.2	33.4	2.8	10.2	1.3	11.9	1.2	8.0
	Overall	33.2	8.4	59.0	3.2	15.9	1.6	33.3	1.2	14.5

Table 5.2. Maximal and average NPV deviations of heuristics from optimal solution(in percent) within the considered return scenarios.

By comparing the results in the different scenarios and reconsidering the corresponding number of usable returns, it can be seen that the higher this number the higher the potential benefit from remanufacturing. In such cases, specifically the number of instances where remanufacturing takes place increases (e.g. in Scenario III it is higher than in Scenario II). The performance of the heuristic approaches decreases if either the return fraction F increases or returns arrive (relatively) early. Since both cases allow for higher investment expenses, an erroneous determination of investment time has a higher impact on the performance.

Low versus high recovery cost advantage (disposal cost rate). Having high disposal costs or a high recovery cost advantage noticeably increases the number of instances where remanufacturing makes sense as shown in Table 5.3. Performance of the considered heuristics decreases except for H3 where the difference lacks significance, because of its generally large variability of relative deviations from the optimal solution. Another reason why this heuristic does not perform much worse is the positive effect of the recovery cost advantage on the Maximal Holding Time τ_u . Thus, the profitability of using a recoverables inventory increases, but also the possible error when neglecting τ_u decreases. This reasoning also explains why H3 performs poorly compared with H2 if c_w is low.

Low versus high discount rate. Although the effect can hardly be termed large (see Table 5.4), a higher discount rate leads to a decreasing profitability of remanufacturing, but it also lowers the precision of H1-H3. This especially holds for H1, which does not consider time value. The potential benefit of keeping stock increases (H2 performs worse), because it would allow for a further postponement of the investment time which has a stronger effect than would be with a lower discount rate. In contrast to H3 which performs worse when increasing the discount rate, there is probably (although it lacks significance)

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	H4	
low 15.3 3.8 38.3 1.2 10.1 1.4 33.3 0.3	ax.	
high 51.1 9.8 59.0 3.8 15.9 1.6 14.5 1.5 1	4.5	

Table 5.3. Maximal and average NPV deviations of heuristics from optimal solution(in percent) with relative high and low recovery cost advantages.

⁶ Averages are not significantly different.

an improvement for H4, which is due to H4 reacting on a modification in discounting both by changing the investment time and by correctly adapting the modified Maximal Holding Time.

Table 5.4. Maximal and average NPV deviations of heuristics from optimal solution (in percent) with relative high and low discount rates.

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			H1		ŀ	1 2	H3		$H4^*$	
	α	#	avg.	max.	avg.	max.	avg.	max.	avg.	max.
high 30.1 11.4 59.0 3.5 15.9 1.8 33.3 1.1 14.5										
	high	30.1	11.4	59.0	3.5	15.9	1.8	33.3	1.1	14.5

* Averages are not significantly different.

Low versus high out-of-pocket holding cost rate. Similarly to a high discount rate, a high out-of-pocket holding cost rate decreases the profitability of remanufacturing, especially of items that have been kept in stock for later use (see Table 5.5). As intuition suggests, it improves the performance of those heuristics which do not keep stock, but surprisingly H3 and H4 also perform better. For H3 this at first glance appears counter-intuitive, but it becomes reasonable because less stock-keeping also results in a smaller deviation from the approximated investment time from the optimum and an improvement of the investment time estimation overcompensating higher holding costs.

Table 5.5. Maximal and average NPV deviations of heuristics from optimal solution(in percent) with relative high and low out-of-pocket holding cost rate.

		I	H1	ŀ	ł2	ŀ	Ŧ3	ŀ	H4
h_u	#	avg.	max.	avg.	max.	avg.	max.	avg.	max.
									14.5
hig	$h_{31.6}$	6.8	44.2	2.3	9.0	1.3	33.3	0.6	7.7

Low versus high investment expenditures for remanufacturing facility. High investment expenditures clearly lead to a strong decrease of the number of instances where remanufacturing makes sense (see Table 5.6). All heuristics perform worse, as they do not correctly reflect the potential of changing the investment time due to stock-keeping, which has a larger effect if investment expenditures increase. This can especially be seen from the average error H4 exhibited.

Table 5.6. Maximal and average NPV deviations of heuristics from optimal solution (in percent) with relative high and low investment expenditures.

Γ			H1		ŀ	H2	I	1 3	H4	
K	ζ_r	#	avg.	max.	avg.	max.	avg.	max.	avg.	max.
										13.6
h	igh	7.4	20.7	59.0	5.4	15.4	2.6	14.5	2.4	14.5

Summary of results. Summarizing the results it can be seen that 'dumb' investment time rule H1 should not be used, because there exists a considerable amount of savings to be realized by applying H2. It is also clear that the use of an anticipation stock not only becomes reasonable because of the additional remanufacturing but it also can be used to change (mostly postponing, as seen in the pre-test) the investment time t_r . This yields the biggest effect with low out-of-pocket holding costs or a high recovery cost advantage. The question on whether to apply one of the heuristics (H2-H4) depends (a) on the situation and (b) on the error which the decision maker is willing to accept. Especially, the knowledge of data and computational requirements which H4 necessitates are comparable to those needed for finding the optimal solution.

5.5 Effects of a Limited Remanufacturing Capacity

So far, capacity aspects did not play a role in our discussion. The optimal solution, however, shows a large variability in both production and remanufacturing rates, leading to an optimal dynamic strategy which requires demand to be completely satisfied from remanufacturing between t_r and t_x and therefore to stop production completely. In this section we deal with the changes caused by a time independent capacity constraint for the remanufacturing process \bar{r} when applying the results derived in Chapter 3.

Investment Project (b). In case of investment project (b), the optimal policy for $t \ge t_r$ changes to $r^*(t) = \min\{d(t), u(t), \bar{r}\}, p^*(t) = d(t) - r^*(t),$ $w^*(t) = u(t) - r^*(t)$ and thus, remanufacturing would be limited if $\bar{r} < u(\min\{t_u^{\max}, t_I\})$. The results with respect to the optimal investment time as stated in Proposition 5.2 remain valid but existence and pay back conditions have to be adapted. A finite investment time does not exist if the critical return rate u^{crit} as defined in Corollary 5.1 exceeds the constraint, i.e. $u^{crit} > \bar{r}$. In this case, the interest rate on the investment always exceeds the maximal feasible cost advantage of remanufacturing. The pay-off condition presented in Proposition 5.3 now must take into account the changed optimal policy and thus, the total discounted net advantage of remanufacturing A_r^b from (5.10) changes to

$$\bar{A}_{r}^{b} = \int_{t_{r}^{*}}^{\infty} e^{-\alpha t} (c_{p}^{r} + c_{w} - c_{r}) \min\{d(t), u(t), \bar{r}\} dt - e^{-\alpha t_{r}^{*}} K_{r}.$$
(5.33)

Since returns are not stored, profitability of remanufacturing reduces because at times when the constraint is binding, part of returns can not be used and have to be disposed of.

Investment Project (c). A (binding) remanufacturing capacity may change the dynamic policy allowing for stock-keeping in two ways (see Figure 5.9). Regarding the already known stock-keeping motive around investment time



Fig. 5.9. Optimal decisions in investment project (c) when having a limited remanufacturing rate.

 t_r (leading to interval $[t_e^1, t_x^1]$), since only part of demand can be satisfied from remanufacturing there is a tendency towards having less returns in inventory at t_r . In an extreme case it might no longer be necessary to have a positive stock at all if the return rate already exceeds the constraint, i.e. $u(t_r) > \bar{r}$. Secondly, by reconsidering the pure effects of a remanufacturing constraint derived in Corollary 3.18, having a positive recoverables stock is beneficial, if there is a time θ in between t_u^{\max} and t_I (where demand is still higher than return rate but return rate is already falling) with $u(\theta) = \bar{r}$. Of course, this requires a situation where $t_u^{\max} < t_I$. Starting at $t_e^2 < \theta$, returns which exceed the remanufacturing constraint can be saved and used up after this time by remanufacturing with maximal remanufacturing rate. This is completed at a time $t_x^2 \leq t_I$. Since depending on the present situation both effects can interact with each other, i.e. returns that are collected before t_r might be used up after θ , a general solution procedure for the constrained problem is considerably more complicated.

5.6 Conclusions and Possible Extensions

In this chapter we used properties of a dynamic situation consisting of a product life cycle and an availability cycle for returns in order to find optimal dynamic policies for three investment projects that differ with respect to the environmental policy. For investment projects that incorporate remanufacturing, the time of the remanufacturing investment proved to be a crucial decision variable because it influences both the time value of the expenses accompanied with it, but the advantage that can be obtained by replacing production of new products by remanufacturing of returns. We have shown that improvements exist if returns can be kept in a strategic inventory. From our experiments it looks that this option should not be implemented as a rule of thumb e.g. by generally keeping all returns. That being said, further research using real life data is required to quantify the performance losses of heuristic approaches.

A number of possibilities exist for further research. It would be interesting to see how robust our results behave when assuming imperfect knowledge on future demand and return developments. A more complex demand/return situation (for instance with a relaunch of the product life cycle) can be solved by using general results for controlling the product recovery system as shown in Chapter 2. But in contrast to the simplified situation introduced above, only numerical results can be derived.

Although capacity aspects have been neglected while performing the analysis they have an impact on profitability of the considered investment projects. In cases, where the remanufacturing process is limited by a capacity constraint the potential advantage of remanufacturing generally decreases but as a side effect, an additional motivation for stock-keeping must be taken into consideration, namely to deal with a situation where the constraint leaves the state of being binding. As an extension, for instance the optimal level of remanufacturing capacity may be obtained. Furthermore, in a strategic model optimal production/remanufacturing capacity expansion and contraction paths can be determined. As in pure production models, building up capacity might be postponed or avoided by using a strategic serviceables inventory. Next, since there are two options to fill the demand, the choice of capacity in the remanufacturing shop influences the required production capacity and vice versa.

We assumed that demand development does not depend on the chosen technology. Marketing aspects like consumer awareness towards environmental conscious products are neglected. Further, competition both on demand as well on the return side are not considered. For instance, the easier products are recoverable, the higher the possibility that other firms will want to participate and to carry out remanufacturing in competition against the OEM. This can be seen for example in the case of refilling toner cartridges for laser printers utilized by Majumder and Groenevelt (2001) to motivate a two-period model which aims to explain how the level of remanufacturability of a product influences competition. An active returns acquisition management as has been described by Guide and Van Wassenhove (2001) can therefore lead to further cost reductions. See Minner and Kiesmüller (2002) for a detailed analysis of return acquisition in a dynamic framework.

5.7 Proofs and Statistical Tests

Proofs

Proof (Proof of Proposition 5.1). An intersection point $t \ge \Delta$ of demand and return functions requires

$$d(t) = \frac{MP(P+Q)^2 \cdot e^{-(P+Q)t}}{(P+Qe^{-(P+Q)t})^2} = \frac{FMP(P+Q)^2 \cdot e^{-(P+Q)(t-\Delta)}}{(P+Qe^{-(P+Q)(t-\Delta)})^2} = r(t).$$
(5.34)

Let $X(t) := e^{-(P+Q)t}$ be a strictly decreasing function of time with co-domain (0,1] and $Y := e^{(P+Q)\Delta} > 1$ be a constant, then (5.34) can (omitting time indices) be rewritten as

$$\frac{MP(P+Q)^2X}{(P+QX)^2} = \frac{FMP(P+Q)^2XY}{(P+QXY)^2}$$

$$\Leftrightarrow P^2 + 2PQXY + Q^2X^2Y^2 = FYP^2 + 2FPQXY + FQ^2X^2Y.$$

Rearranging terms leads to the following quadratic equation of form $ax^2 + bx + c = 0$ in X(t)

$$Q^{2}Y(Y-F) \cdot X^{2} + 2PQY(1-F) \cdot X + P^{2}(1-FY) = 0.$$
 (5.35)

A solution to (5.35) is given by $\left(x = \frac{-b \pm \sqrt{b^2 - 4ac}}{2a}\right)$

$$X = \frac{-2PQY(1-F)\pm\sqrt{(2PQY(1-F))^2 - 4(Q^2Y(Y-F))(P^2(1-FY))}}{2Q^2Y(Y-F)}$$
$$= -\frac{P(1-F)}{Q(Y-F)} \pm \frac{\sqrt{4FP^2Q^2Y(1-Y)^2}}{2Q^2Y(Y-F)}$$
$$= -\frac{P(1-F)}{Q(Y-F)} \pm \frac{P|1-Y|\sqrt{FY}}{QY(Y-F)}$$
(5.36)

Two positive real roots may exist if the first term in (5.36) is positive, which is not true because 1 - F as well as Y - F never take on negative values. Since 1 - Y < 0 and given X exists, it is

$$X = \frac{P/Y(Y-1)\sqrt{FY} - P(1-F)}{Q(Y-F)},$$
(5.37)

and intersection t_I is uniquely given $t_I = \frac{-\ln(X)}{P+Q}$.

From (5.35) one can see, that a positive root X requires 1-FY < 0. Therefore, if

$$F \le 1/Y = e^{-(P+Q)\Delta} \tag{5.38}$$

no intersection between the demand and return function can exist and we can deduce using F < 1, that $r(t) < d(t) \forall t$. Note that this condition is sufficient only.

In order to derive a necessary and sufficient condition for the existence of intersection point t_I , it additionally has to be assured that X (as derived from using (5.37)) is located in the co-domain (0, 1]. Further, for the inverse of X it must hold that the derived time point $t_I = X^{-1}$ exceeds Δ , which is equivalent to $r(\Delta) = Fd(0) < d(\Delta)$.

Proof (Proof of Proposition 5.2).

When inserting u(t) = 0 for $t < \Delta$ and replacing the max/min operators in (5.6) the objective is (omitting time indices) given by

$$NPV_{b} = \begin{cases} K_{p}^{r} + \int_{0}^{\Delta} e^{-\alpha t} c_{p}^{r} d \, dt + e^{-\alpha t_{r}} K_{r} & t_{r} < \Delta \\ + \int_{\Delta}^{t_{I}} e^{-\alpha t} [c_{p}^{r} (d-u) + c_{r} u] dt + \int_{t_{I}}^{\infty} e^{-\alpha t} [c_{r} d + c_{w} (u-d)] dt \\ K_{p}^{r} + \int_{0}^{\Delta} e^{-\alpha t} c_{p}^{r} d \, dt + \int_{\Delta}^{t_{r}} e^{-\alpha t} [c_{p}^{r} d + c_{w} u] dt + e^{-\alpha t_{r}} K_{r} \\ + \int_{t_{r}}^{t_{I}} e^{-\alpha t} [c_{p}^{r} (d-u) + c_{r} u] dt + \int_{t_{I}}^{\infty} e^{-\alpha t} [c_{r} d + c_{w} (u-d)] dt \\ K_{p}^{r} + \int_{0}^{\Delta} e^{-\alpha t} c_{p}^{r} d \, dt + \int_{\Delta}^{t_{r}} e^{-\alpha t} [c_{p}^{r} d + c_{w} u] dt + e^{-\alpha t_{r}} K_{r} \\ + \int_{t_{r}}^{\infty} e^{-\alpha t} c_{p}^{r} d \, dt + \int_{\Delta}^{t_{r}} e^{-\alpha t} [c_{p}^{r} d + c_{w} u] dt + e^{-\alpha t_{r}} K_{r} \\ + \int_{t_{r}}^{\infty} e^{-\alpha t} [c_{r} d + c_{w} (u-d)] dt \end{cases}$$

The first derivative of the objective (5.39) is different for each of the three regions as defined in Section 5.2 and given by

$$\frac{\partial NPV_b}{\partial t_r} = \begin{cases} -\alpha e^{-\alpha t_r} K_r & \text{for } t_r < \Delta \quad (\text{Region 1}) \\ \text{undefined} & \text{for } t_r = \Delta \\ e^{-\alpha t_r} [(c_p^r + c_w - c_r)u(t_r) - \alpha K_r] \text{ for } \Delta < t_r < t_I \text{ (Region 2)} \\ e^{-\alpha t_r} [(c_p^r + c_w - c_r)d(t_r) - \alpha K_r] \text{ for } t_I \le t_r \quad (\text{Region 3}). \end{cases}$$

$$(5.40)$$

Note that except for $t_r = \Delta$, (5.40) is continuous. Candidates for t_r^* are given by points where (5.40) changes its sign from negative to positive. This is not possible in Region 1, where the derivative is negative.

possible in Region 1, where the derivative is negative. The time point $t_{r,1} = \Delta$ is a candidate for t_r^* if $\lim_{t_r \to \Delta + 0} \frac{\partial NPV_b}{\partial t_r} \ge 0$. This leads to

$$u(\Delta)(c_p^r + c_w - c_r) \ge \alpha K_r.$$
(5.41)

Further candidates, $t_{r,2}$ and $t_{r,3}$, are given by setting the first derivative in the two remaining regions to zero, which gives the following conditions

$$u(t_{r,2})(c_p^r + c_w - c_r) = \alpha K_r \text{ for } \Delta < t_{r,2} < t_I \text{ or}$$
 (5.42)

$$d(t_{r,3})(c_p^r + c_w - c_r) = \alpha K_r \text{ for } t_I \le t_{r,3}.$$
(5.43)

The second derivative of NPV_b is given by

$$\frac{\partial^2 NPV_b}{\partial t_r^2} = \begin{cases} \alpha^2 e^{-\alpha t_r} K_r & \text{for } t_r < \Delta \\ \text{undefined} & \text{for } t_r = \Delta \\ e^{-\alpha t_r} [(c_p^r + c_w - c_r)(\dot{u}(t_r) - \alpha u(t_r)) + \alpha^2 K_r] & \text{for } \Delta < t_r < t_I \\ \text{undefined} & \text{for } t_r = t_I \\ e^{-\alpha t_r} [(c_p^r + c_w - c_r)(\dot{d}(t_r) - \alpha d(t_r)) + \alpha^2 K_r] & \text{for } t_I < t_r. \end{cases}$$

$$(5.44)$$

Inserting (5.42) and (5.43) for αK_r in the respective part of (5.44), conditions for a local minimum can be derived. $t_{r,2}$ is a local minimum if

$$\frac{\partial^2 NPV_b}{\partial t_r^2}\Big|_{t_r=t_{r,2}} = e^{-\alpha t_r} (c_p^r + c_w - c_r) \dot{u}(t_{r,2}) > 0 \Leftrightarrow \dot{u}(t_{r,2}) > 0$$
(5.45)

and thus, $t_{r,2} < t_u^{max}$. Analogously, $t_{r,3}$ is a local minimum if $\dot{d}(t_{r,3}) > 0$. Since demand must decrease for any point $t_{r,3} \ge t_I$ (from assumption), candidate $t_{r,3}$ fails the second order necessary conditions.

From our assumptions about the return function (unimodal) it follows that if inequality (5.41) holds, i.e. $t_{r,1} = \Delta$ is a candidate for an optimal solution, there will be no candidate $t_{r,2}$ and vice versa. Hence, there exists at most a single finite solution, located in an half open interval $[\Delta, \min\{t_u^{max}, t_I\})$.

Note, that this proof must be adapted in order to apply to a situation as defined in Proposition 5.1 *(ii)* and as a result, only $t_{r,1} = \Delta = t_I$ is a solution candidate, because immediately afterwards the remanufacturing rate must decrease.

Proof (Proof of Proposition 5.3).

Since NPV_b decreases for sufficient high t_r , i.e. there exists a time $t_r > t_I$ for which it holds $\forall t > t_r$: $d(t) < \frac{\alpha K_r}{c_p + c_w - c_r}$, solution candidate $t_r^{\infty} = \infty$ (invest never) has to be considered.

In order to find the best alternative, the Net Present Value of the payment stream arising by assuming a relevant finite candidate $\tilde{t}_r \in [\Delta, \min\{t_u^{max}, t_I\})$, i.e. $NPV_b(\tilde{t}_r)$, has to be compared with $NPV_b(t_r^{\infty})$, which is given by

158 5 Technology Selection in the Context of Reverse Logistics

$$NPV_{b}(t_{r}^{\infty}) = K_{p}^{r} + \int_{0}^{\infty} e^{-\alpha t} \left[c_{p}^{r} d(t) + c_{w} u(t) \right] dt.$$
(5.46)

This gives an expression of the total discounted advantage of remanufacturing A^b_r

$$A_{r}^{b} = NPV_{b}(t_{r}^{\infty}) - NPV_{b}(\tilde{t}_{r})$$

$$= -e^{-\alpha \tilde{t}_{r}}K_{r} + \int_{0}^{\infty} e^{-\alpha t} \left[c_{p}^{r}d(t) + c_{w}u(t) \right] dt$$

$$-\int_{\tilde{t}_{r}}^{\infty} e^{-\alpha t} \left[c_{p}^{r} \max\{d(t) - u(t), 0\} + c_{r} \min\{d(t), u(t)\} + c_{w} \max\{u(t) - d(t), 0\} \right] dt$$

$$= -e^{-\alpha \tilde{t}_{r}}K_{r} + \int_{\tilde{t}_{r}}^{\infty} e^{-\alpha t} \left[(c_{p}^{r} + c_{w} - c_{r}) \min\{d(t), u(t)\} \right] dt \qquad (5.47)$$

Therefore, $t_r^* = \tilde{t}_r$ if $A_r^b > 0$, i.e.

$$K_r \le \int_{\tilde{t}_r}^{\infty} e^{-\alpha(t-\tilde{t}_r)} \left[c_p^r + c_w - c_r \right] \min\{d(t), u(t)\} dt.$$
(5.48)

Otherwise $t_r^* = \infty$.

Proof (Proof of Propositions 5.4 to 5.9).

Propositions 5.4 to 5.9 are results of an optimization approach which is carried through in the following. Since both the objective function NPV_c and constraint (5.21) are in general not convex, in accordance with Sydsæter and Hammond (1995), p 608 the following solution method is used:

- (1) Determination of the partial derivatives of the objective function.
- (2) Identification of possible solution candidates (Steps 1 and 2 in Sydsæter and Hammond (1995)) using standard methods of Nonlinear Programming. This proofs Proposition 5.4. Exploring a joint property of all valid cases proofs Proposition 5.5 while individual properties confirm results stated in Propositions 5.6-5.9.
- (3) Comparison of values of NPV_c at candidate points against each other (Step 3) and with the Net Present Value of investing never $NPV_b(t_r^{\infty})$ as given in (5.46). Smallest value is the (global) minimal value of NPV_c (Step 5). As this requires actual data, we will omit this part.

(1) Partial derivatives of the objective function

Subsequently, the partials of t_x with respect to t_e and t_r are needed. This is applied by using the implicit differentiation rules

$$\frac{\partial t_x}{\partial t_e} = -\frac{\frac{\partial f}{\partial t_e}}{\frac{\partial f}{\partial t_x}} = -\frac{u(t_e)}{d(t_x) - u(t_x)} \tag{5.49}$$

$$\frac{\partial t_x}{\partial t_r} = -\frac{\frac{\partial f}{\partial t_r}}{\frac{\partial f}{\partial t_x}} = \frac{d(t_r)}{d(t_x) - u(t_x)}$$
(5.50)

The first partial derivative of $NPV_c(t_e, t_r)$ with respect to t_e is given by

$$\frac{\partial NPV_c}{\partial t_e} = e^{-\alpha t_e} \left[c_p^r d(t_e) + c_w u(t_e) \right] - e^{-\alpha t_e} \left[c_p^r d(t_e) + h_u y_u(t_e) \right] \\ + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_e} dt \\ + \left(e^{-\alpha t_x} \left[c_r d(t_x) + h_u y_u(t_x) \right] + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_x} dt \\ - e^{-\alpha t_x} \left[c_p^r (d(t_x) - u(t_x)) + c_r u(t_x) \right] \right) \frac{\partial t_x}{\partial t_e}.$$

Collecting terms and inserting $y_u(t_e) = y_u(t_x) = 0$ yields

$$\frac{\partial NPV_c}{\partial t_e} = e^{-\alpha t_e} c_w u(t_e) + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_e} dt + \left(-e^{-\alpha t_x} \left[c_p^r - c_r \right] \left(d(t_x) - u(t_x) \right) + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_x} dt \right) \frac{\partial t_x}{\partial t_e}.$$

Since the partial of $y_u(t)$ with respect to t_e equals

$$\frac{\partial y_u(t)}{\partial t_e} = \begin{cases} \text{undefined} & \text{for } t = t_e \\ -u(t_e) & \text{for } t \in (t_e, t_x) \\ \text{undefined} & \text{for } t = t_x \\ 0 & \text{otherwise} \end{cases}$$
(5.51)

and because of $\frac{\partial y_u(t)}{\partial t_x} = 0$ it follows that

$$\frac{\partial NPV_c}{\partial t_e} = e^{-\alpha t_e} c_w u(t_e) - \int_{t_e}^{t_x} e^{-\alpha t} h_u u(t_e) dt - e^{-\alpha t_x} [c_p^r - c_r] d(t_x) - u(t_x)) \frac{\partial t_x}{\partial t_e}.$$

Replacing $\frac{\partial t_x}{\partial t_e}$ by $-\frac{u(t_e)}{d(t_x)-u(t_x)}$ using (5.49) gives

$$\frac{\partial NPV_c}{\partial t_e} = \left(e^{-\alpha t_e}c_w - h_u \int_{t_e}^{t_x} e^{-\alpha t}dt + e^{-\alpha t_x} \left[c_p^r - c_r\right]\right) u(t_e).$$
(5.52)

Equation (5.52) can be interpreted as follows. If t_e increases, a marginal return $u(t_e)$ arriving at this time is no longer stored but disposed of, leading to additional unit costs of $e^{-\alpha t_e}c_w u(t_e)$. Since this (marginal) return is not available for later remanufacturing, t_x decreases and thus, costs for producing a (marginal) new product at t_x are caused, given by $e^{-\alpha t_x} (c_p^r - c_r) u(t_e)$. On the other hand, storing less returns reduces inventory holding costs per unit by $h_u \int_{t_e}^{t_x} e^{-\alpha t} dt u(t_e)$.

Evaluating the integral in (5.52) finally yields

$$\frac{\partial NPV_c}{\partial t_e} = \left(e^{-\alpha t_e}\left[c_w - \frac{h_u}{\alpha}\right] + e^{-\alpha t_x}\left[c_p^r - c_r + \frac{h_u}{\alpha}\right]\right)u(t_e).$$
(5.53)

The first partial derivative of $NPV_c(t_e, t_r)$ with respect to t_r is given by

$$\begin{aligned} \frac{\partial NPV_c}{\partial t_r} &= e^{-\alpha t_r} \left[c_p^r d(t_r) + h_u y_u(t_r) \right] - \alpha e^{-\alpha t_r} K_r - e^{-\alpha t_r} \left[c_r d(t_r) + h_u y_u(t_r) \right] \\ &+ \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_r} dt \\ &+ \left(e^{-\alpha t_x} \left[c_r d(t_x) + h_u y_u(t_x) \right] + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_x} dt \\ &- e^{-\alpha t_x} \left[c_p^r (d(t_x) - u(t_x)) + c_r u(t_x) \right] \right) \frac{\partial t_x}{\partial t_r}.\end{aligned}$$

160 5 Technology Selection in the Context of Reverse Logistics

Collecting terms and inserting $y_u(t_x) = 0$ yields

$$\frac{\partial NPV_c}{\partial t_r} = e^{-\alpha t_r} \left[c_p^r - c_r \right] d(t_r) - \alpha e^{-\alpha t_r} K_r + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_r} dt \\ + \left(-e^{-\alpha t_x} \left[c_p^r - c_r \right] (d(t_x) - u(t_x)) + \int_{t_e}^{t_x} e^{-\alpha t} h_u \frac{\partial y_u(t)}{\partial t_x} dt \right) \frac{\partial t_x}{\partial t_r}.$$

Using

$$\frac{\partial y_u(t)}{\partial t_r} = \begin{cases} \text{undefined} & \text{for } t = t_r \\ d(t_r) & \text{for } t \in (t_r, t_x) \\ \text{undefined} & \text{for } t = t_x \\ 0 & \text{otherwise} \end{cases}$$
(5.54)

it follows that

$$\frac{\partial NPV_c}{\partial t_r} = e^{-\alpha t_r} \left[c_p^r - c_r \right] d(t_r) - \alpha e^{-\alpha t_r} K_r + \int_{t_r}^{t_x} e^{-\alpha t} h_u d(t_r) dt - e^{-\alpha t_x} \left[c_p^r - c_r \right] \left(d(t_x) - u(t_x) \right) \frac{\partial t_x}{\partial t_r}.$$

Replacing $\frac{\partial t_x}{\partial t_r}$ by $\frac{d(t_r)}{d(t_x)-u(t_x)}$ from (5.50) yields

$$\frac{\partial NPV_c}{\partial t_r} = \left(\left(e^{-\alpha t_r} - e^{-\alpha t_x} \right) \left[c_p^r - c_r \right] + h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \right) d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.55)

A later investment time t_r decreases the Net Present Value of the investment expenses by $\alpha e^{-\alpha t_r} K_r$. A (marginal) demand $d(t_r)$ is no longer satisfied by remanufacturing returns at t_r , which instead are stored for a later use at t_x . Therefore, a cost reduction at t_x by remanufacturing instead of producing faces an increase in costs at t_r . Additional holding costs are caused by storing the (marginal) return.

Continuing in the same manner as above gives

$$\frac{\partial NPV_c}{\partial t_r} = \left(e^{-\alpha t_r} - e^{-\alpha t_x}\right) \left[c_p^r - c_r + \frac{h_u}{\alpha}\right] d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.56)

(2) Identification of solution candidates

By introducing Lagrange multipliers μ_i , i = 1, 2, 3 which are associated with constraints (5.19)-(5.20) the Lagrangian $\mathcal{L}(t_e, t_r, \mu_1, \mu_2, \mu_3)$ is defined as

$$\mathcal{L}(t_e, t_r, \mu_1, \mu_2, \mu_3) = NPV_c(t_e, t_r) - \mu_1(t_e - \Delta) - \mu_2(t_r - t_e) - \mu_3\left(\int_{t_x}^{t_I} (d(s) - u(s))ds\right).$$
(5.57)

The partials of $\mathcal{L}(t_e, t_r, \mu_1, \mu_2, \mu_3)$ have to equal zero:

$$\frac{\partial \mathcal{L}}{\partial t_e} = \frac{\partial NPV_c}{\partial t_e} - \mu_1 + \mu_2 + \mu_3(d(t_x) - u(t_x))\frac{\partial t_x}{\partial t_e} \\
= \frac{\partial NPV_c}{\partial t_e} - \mu_1 + \mu_2 - \mu_3 u(t_e) = 0$$
(5.58)
$$\frac{\partial \mathcal{L}}{\partial t_r} = \frac{\partial NPV_c}{\partial t_r} - \mu_2 + \mu_3(d(t_x) - u(t_x))\frac{\partial t_x}{\partial t_r} \\
= \frac{\partial NPV_c}{\partial t_r} - \mu_2 + \mu_3 d(t_r) = 0$$
(5.59)

The complementary slackness conditions are given by

$$\frac{\partial \mathcal{L}}{\partial \mu_1} = \Delta - t_e \le 0, \, \mu_1 \ge 0, \, \mu_1(\Delta - t_e) = 0 \tag{5.60}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_2} = t_e - t_r \le 0, \, \mu_2 \ge 0, \, \mu_2(t_e - t_r) = 0 \tag{5.61}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_3} = -\int_{t_x}^{t_I} (d(s) - u(s)) ds \le 0, \, \mu_3 \ge 0, \, \mu_3 \left(\int_{t_x}^{t_I} (d(s) - u(s)) ds \right) = 0$$
(5.62)

Having three constraints, either active or inactive, in total eight cases have to be distinguished. Cases 1-4 are excluded, cases (i)-(iv) make up solution candidates in Proposition 5.4.

Case 1 $t_e = t_r$, $\Delta = t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds = 0 \Leftrightarrow \Delta = t_e = t_r = t_x = t_I$ This case is only feasible if $\Delta = t_I$ and can therefore be excluded immediately. Case 2 $t_e = t_r$, $\Delta = t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds > 0 \Leftrightarrow \Delta = t_e = t_r = t_x < t_I$ From $t_x < t_I$ using (5.62), $\mu_3 = 0$ follows. Inserting this value into (5.59) gives

$$\mu_2 = \left. \frac{\partial NPV_c}{\partial t_r} \right|_{t_r = t_x = \Delta} = -\alpha e^{-\alpha \Delta} K_r < 0.$$

This contradicts $\mu_2 \ge 0$.

Case 3 $t_e = t_r$, $\Delta < t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds = 0 \Leftrightarrow \Delta < t_e = t_r = t_x = t_I$ Case 3 does not include an interval with a positive recoverables stock and has already been excluded from the set of solution candidates to investment project (b). See proof to Proposition 5.2.

Case 4 $t_e = t_r$, $\Delta < t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds > 0 \Leftrightarrow \Delta < t_e = t_r = t_x < t_I$ Both, first and third conditions are inactive. Then, $\mu_1 = 0$ and $\mu_3 = 0$ by (5.60) and (5.62), respectively. From (5.58) it follows

$$\mu_2 = -\left.\frac{\partial NPV_c}{\partial t_e}\right|_{t_e=t_r=t_x} = -e^{-\alpha t_r} \left[c_p^r + c_w - c_r\right] u(t_r).$$

Because $c_p^r + c_w - c_r > 0$ from (5.3) and $u(t) > 0 \ \forall t \ge \Delta$, μ_2 is negative and thus contradicts $\mu_2 \ge 0$.

Since all cases (1-4) with $t_e = t_r$ are excluded, for the optimal solution (t_e^*, t_r^*) it holds $t_e^* < t_r^*$. The remaining cases constitute the different types of solution candidates as stated in Proposition 5.4. This completes the proof to Proposition 5.4.

For all remaining cases (i)-(iv) holds $t_e^* < t_r^*$ which requires from (5.61) $\mu_2 = 0$. Inserting this value into (5.58) (reconsidering non-negativity of μ_1 , μ_3 , and $u(t_e^*)$) necessitates $\frac{\partial NPV_e}{\partial t_e} \geq 0$, yielding

$$\frac{\partial NPV_c}{\partial t_e} = \left(e^{-\alpha t_e^*}\left[c_w - \frac{h_u}{\alpha}\right] + e^{-\alpha t_x^*}\left[c_p^r - c_r + \frac{h_u}{\alpha}\right]\right)u(t_e^*) \ge 0 \quad (5.63)$$

Because of $u(t) > 0 \ \forall t \ge \Delta$ this is equivalent to

$$e^{-\alpha t_e^*} \left[c_w - \frac{h_u}{\alpha} \right] + e^{-\alpha t_x^*} \left[c_p^r - c_r + \frac{h_u}{\alpha} \right] \ge 0$$

and solving for $t_x^* - t_e^*$ finally yields

$$t_x^* - t_e^* \le \frac{1}{\alpha} \ln\left(\frac{\alpha \left(c_p - c_r\right) + h_u}{-\alpha c_w + h_u}\right) =: \tau_u, \tag{5.64}$$

i.e. the maximum length condition known from Kleber et al. (2002) for the special case of a single product. This completes the proof to Proposition 5.5.

Case (i) $t_e < t_r$, $\Delta < t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds > 0 \Leftrightarrow \Delta < t_e < t_x < t_I$ None of the conditions is active. Thus, $\mu_1 = 0$, $\mu_2 = 0$ as well as $\mu_3 = 0$. Inserted into (5.58), this yields

$$\frac{\partial NPV_c}{\partial t_e} = \left(e^{-\alpha t_e}c_w - h_u \int_{t_e}^{t_x} e^{-\alpha t}dt + e^{-\alpha t_x} \left[c_p^r - c_r\right]\right) u(t_e) = 0$$

Proceeding as above (5.63)-(5.64) leads to

$$t_x - t_e = \tau_u. \tag{5.65}$$

Further, inserting Lagrange Multipliers into (5.59) requires

$$\frac{\partial NPV_c}{\partial t_r} = \left(\left(e^{-\alpha t_r} - e^{-\alpha t_x} \right) \left[c_p^r - c_r \right] + h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \right) d(t_r) - \alpha e^{-\alpha t_r} K_r = 0.$$

$$\Leftrightarrow -e^{-\alpha t_r} [c_p^r - c_r] d(t_r) = h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_x} [c_p^r - c_r] d(t_r) - \alpha e^{-\alpha t_r} K_r$$
(5.66)

This completes the proof to Proposition 5.6.

Case (ii) $t_e < t_r$, $\Delta = t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds = 0 \Leftrightarrow \Delta = t_e < t_r < t_x = t_I$ Inserting $\mu_2 = 0$ into (5.59) yields

$$\begin{aligned} \mu_3 &= -\left. \frac{\partial NPV_c}{\partial t_r} \right|_{t_x = t_I} \frac{1}{d(t_r)} \\ &= -\left(e^{-\alpha t_r} - e^{-\alpha t_I} \right) \left[c_p^r - c_r \right] - h_u \int_{t_r}^{t_I} e^{-\alpha t} dt + \alpha e^{-\alpha t_r} K_r \frac{1}{d(t_r)}. \end{aligned}$$

 $\mu_3 \ge 0$ requires

$$\left(\left(e^{-\alpha t_r} - e^{-\alpha t_I} \right) \left[c_p^r - c_r \right] + h_u \int_{t_r}^{t_I} e^{-\alpha t} dt \right) d(t_r) - \alpha e^{-\alpha t_r} K_r \le 0$$

$$\Leftrightarrow -e^{-\alpha t_r} \left[c_p^r - c_r \right] d(t_r) \ge h_u \int_{t_r}^{t_I} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_I} \left[c_p^r - c_r \right] d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.67)

Further, inserting μ_3 into (5.58) leads to

$$\mu_{1} = \left. \frac{\partial NPV_{c}}{\partial t_{e}} \right|_{t_{e}=\Delta, \ t_{x}=t_{I}} + \left. \frac{\partial NPV_{c}}{\partial t_{r}} \right|_{t_{x}=t_{I}} \frac{u(\Delta)}{d(t_{r})}$$

$$= \left(e^{-\alpha\Delta}c_{w} - h_{u} \int_{\Delta}^{t_{I}} e^{-\alpha t} dt + e^{-\alpha t_{I}} \left[c_{p}^{r} - c_{r} \right] \right) u(\Delta)$$

$$+ \left((e^{-\alpha t_{r}} - e^{-\alpha t_{I}}) \left[c_{p}^{r} - c_{r} \right] + h_{u} \int_{t_{r}}^{t_{I}} e^{-\alpha t} dt \right) u(\Delta) - \alpha e^{-\alpha t_{r}} K_{r} \frac{u(\Delta)}{d(t_{r})}$$

$$= \left(e^{-\alpha\Delta}c_{w} - h_{u} \int_{\Delta}^{t_{r}} e^{-\alpha t} dt + e^{-\alpha t_{r}} \left[c_{p}^{r} - c_{r} \right] \right) u(\Delta) - \alpha e^{-\alpha t_{r}} K_{r} \frac{u(\Delta)}{d(t_{r})}.$$

$$(5.68)$$

Since $\mu_1 \ge 0$, (5.68) implies

$$\left(e^{-\alpha\Delta}c_w - h_u \int_{\Delta}^{t_r} e^{-\alpha t} dt + e^{-\alpha t_r} [c_p^r - c_r] \right) d(t_r) - \alpha e^{-\alpha t_r} K_r \ge 0$$

$$\Leftrightarrow e^{-\alpha\Delta}c_w d(t_r) \ge h_u \int_{\Delta}^{t_r} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_r} [c_p^r - c_r] d(t_r) + \alpha e^{-\alpha t_r} K_r.$$

$$(5.69)$$

This completes the proof to Proposition 5.7.

Case (iii) $t_e < t_r$, $\Delta = t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds > 0 \Leftrightarrow \Delta = t_e < t_r < t_x < t_I$ Both second and third conditions are inactive. Then, $\mu_2 = 0$ and $\mu_3 = 0$ from (5.61) and (5.62), respectively. Inserting both values into (5.59) yields

$$\frac{\partial NPV_c}{\partial t_r} = \left(\left(e^{-\alpha t_r} - e^{-\alpha t_x} \right) \left[c_p^r - c_r \right] + h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \right) d(t_r) - \alpha e^{-\alpha t_r} K_r = 0.$$

$$\Leftrightarrow -e^{-\alpha t_r} \left[c_p^r - c_r \right] d(t_r) = h_u \int_{t_r}^{t_x} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_x} \left[c_p^r - c_r \right] d(t_r) - \alpha e^{-\alpha t_r} K_r.$$
(5.70)

This completes the proof to Proposition 5.8.

Case (iv) $t_e < t_r$, $\Delta < t_e$, $\int_{t_x}^{t_I} (d(s) - u(s)) ds = 0 \Leftrightarrow \Delta < t_e < t_x = t_I$ Both, first and second conditions are inactive. Then, $\mu_1 = 0$ and $\mu_2 = 0$ from (5.60) and (5.61), respectively. Both values inserted into (5.58) yields the value for μ_3

$$\mu_3 = \frac{1}{u(t_e)} \left. \frac{\partial NPV_e}{\partial t_e} \right|_{t_x = t_I} = e^{-\alpha t_e} c_w - h_u \int_{t_e}^{t_I} e^{-\alpha t} dt + e^{-\alpha t_I} \left[c_p^r - c_r \right]$$

Inserting μ_3 into (5.59) gives

$$\frac{\partial NPV_c}{\partial t_r}\Big|_{t_x=t_I} + \frac{d(t_r)}{u(t_e)} \left. \frac{\partial NPV_c}{\partial t_e} \right|_{t_x=t_I} = 0$$

$$\Leftrightarrow e^{-\alpha t_e} c_w d(t_r) = h_u \int_{t_e}^{t_r} e^{-\alpha t} dt \, d(t_r) - e^{-\alpha t_r} [c_p^r - c_r] d(t_r) + \alpha e^{-\alpha t_r} K_r.$$
(5.71)

This completes the proof to Proposition 5.9.

Statistical Tests

The objective of the statistical analysis was to test the average performance of the heuristics (A) one against each other, where the results originated from the same experiments (matched pairs), but also (B) the change of the heuristics performance due to different settings (independent group means), two different types of tests had to be performed. In the first case, a paired t-test was carried out which, because of the large sample sizes, was approximated by a Normal z Test. For comparing independent group means w.r.t. the same heuristic in different settings, a single-sided version of the approximately two groups Normal z Test was performed. Because of the large sample size, the significance level α was set to 1% resulting in a quantile value of about 2.33. The difference of two average values can in both tests be assumed statistically significant if the corresponding absolute test value is larger than 2.33.

(A) The results of the first test are shown in Tables 5.7 and 5.8. As can be seen, the equality of average performances of two different heuristics can be rejected under all considered scenarios.

(B) For use during the second test, the standard deviation of the relative error when applying heuristic solutions instead of the optimal solution was required, which is depicted in Figure 5.9. Using these values, test values of the two groups Normal z Test have been calculated as shown in Figure 5.10. Only in two cases were differences of the compared means not significantly large. These have been highlighted by bold letters.

]	H1 vs. H	2]	H1 vs. H	3]	H1 vs. H	4
	mean	std.dev.	z test	mean	std.dev.	z test	mean	std.dev.	z test
Overall	0.052	0.077	44.62	0.069	0.085	52.81	0.072	0.084	55.88
Scenario I	0.058	0.080	21.73	0.074	0.084	26.10	0.077	0.083	27.51
Scenario II	0.033	0.046	20.37	0.044	0.049	25.46	0.045	0.048	26.54
Scenario III	0.075	0.097	28.95	0.096	0.109	33.20	0.103	0.107	36.11
Scenario IV	0.034	0.049	23.40	0.048	0.055	29.48	0.050	0.055	30.86
c_w low	0.027	0.040	20.75	0.024	0.046	16.27	0.035	0.044	24.79
c_w high	0.060	0.083	41.44	0.082	0.089	52.70	0.083	0.090	52.97
α low	0.031	0.047	31.61	0.046	0.056	39.57	0.047	0.056	41.16
α high	0.079	0.095	36.37	0.096	0.103	40.84	0.102	0.102	44.18
h_u low	0.059	0.087	32.11	0.081	0.095	40.35	0.082	0.094	41.04
h_u high	0.045	0.063	32.30	0.055	0.070	35.48	0.062	0.071	39.37
K_r low	0.040	0.060	40.81	0.054	0.070	48.05	0.058	0.069	51.63
K_r high	0.153	0.112	29.66	0.181	0.107	36.96	0.183	0.107	37.27

Table 5.7. Average difference, sample standard deviation and z test values for comparison of the average performance of different heuristics.

Table 5.8. Average difference, sample standard deviation and z test values for comparison of the average performance of different heuristics (continued).

	I	H2 vs. H3	3]	H2 vs. H	4	I	H3 vs. H	4
	mean	$\operatorname{std.dev.}$	z test	mean	$\operatorname{std.dev.}$	z test	mean	std.dev.	z test
Overall	0.016	0.030	35.11	0.020	0.025	52.24	0.004	0.016	14.92
Scenario I	0.016	0.025	18.49	0.019	0.022	25.54	0.003	0.011	8.12
Scenario II	0.011	0.016	19.79	0.012	0.014	24.41	0.001	0.005	7.52
Scenario III	0.021	0.042	18.91	0.028	0.032	33.21	0.007	0.025	10.59
Scenario IV	0.014	0.022	22.23	0.016	0.020	27.40	0.002	0.008	7.36
c_w low	-0.003	0.032	-2.77	0.008	0.011	23.50	0.011	0.030	11.57
c_w high	0.022	0.027	46.33	0.023	0.027	50.10	0.001	0.007	12.10
α low	0.015	0.023	31.83	0.017	0.021	38.01	0.002	0.008	10.51
α high	0.017	0.037	20.71	0.023	0.028	36.87	0.006	0.022	12.08
h_u low	0.022	0.030	34.14	0.023	0.029	37.35	0.001	0.008	6.25
h_u high	0.010	0.028	15.56	0.016	0.019	39.94	0.007	0.022	13.85
K_r low	0.015	0.029	31.09	0.019	0.023	50.03	0.004	0.017	14.12
K_r high	0.028	0.035	17.49	0.030	0.035	18.60	0.002	0.007	6.73

	H1	H2	H3	H4
	std.dev.	std.dev.	std.dev.	std.dev.
Overall	0.089	0.029	0.021	0.016
Scenario I	0.090	0.026	0.018	0.016
Scenario II	0.052	0.017	0.010	0.010
Scenario III	0.111	0.035	0.028	0.019
Scenario IV	0.059	0.023	0.015	0.014
c_w low	0.046	0.013	0.029	0.007
c_w high	0.094	0.029	0.017	0.017
α low	0.061	0.027	0.016	0.015
α high	0.107	0.030	0.025	0.017
h_u low	0.100	0.033	0.020	0.019
h_u high	0.072	0.018	0.022	0.009
K_r low	0.072	0.026	0.020	0.014
K_r high	0.112	0.034	0.025	0.026

Table 5.9. Standard deviation of the relative error of heuristic solutions from optimal solution.

Table 5.10. Z test values for comparison of the average performance of the same heuristic between different scenarios/settings.

	H1	H2	H3	H4
Scenario I vs. II	10.09	9.67	8.13	6.26
Scenario I vs. III	-6.75	-9.03	-6.52	-3.02
Scenario I vs. IV	-8.18	-10.39	-8.43	-3.30
Scenario II vs. III	-18.40	-19.75	-14.36	-10.18
Scenario II vs. IV	-3.27	-7.70	-6.74	-6.48
Scenario III vs. IV	-16.26	-12.95	-9.37	-4.29
c_w low vs. high	-26.98	-40.15	-1.70	-30.15
α low vs. high	-19.61	-6.16	-5.06	2.09
h_u low vs. high	11.85	22.09	9.18	25.69
K_r low vs. high	-26.15	-15.14	-9.28	-10.53

Conclusions

We have addressed dynamic issues in product recovery management. A continuous time framework was used which enables to account for external dynamic aspects such as demand seasonality and product life cycle effects but it also enables us to deal with internal cost dynamics. Since we dealt with fairly simple models, optimal policies could be determined that lead to general insights into the optimal behavior of product recovery systems.

In a basic model, there is a direct cost advantage of remanufacturing over production, an anticipation stock is used in order to enhance product recovery opportunities if a period with excess returns is followed by another with excess demand. As a main result, a maximal holding time for returns was derived which balances the direct cost advantage of remanufacturing and holding costs. This period limits the time interval which can be influenced by a current decision and it therefore can be used in order to specify the minimal length of the planning horizon within a rolling planning scheme for an integrated production planning. This approach was also useful in valuing product returns which is of importance for accounting issues.

An extended version of the basic model was used to investigate the smoothing effect of anticipation stocks in capacitated product recovery systems. In the case of a manufacturing constraint the current capacity of the system can be insufficient to service demand and thus, stock-keeping is required for bottleneck situations. Exploiting the holding cost advantage of recoverables, such demands are primarily filled from remanufacturing of additionally collected returns. Since the length of a corresponding collection interval depends on the total 'bottleneck size', it might exceed a time period which is motivated through a recovery cost advantage. In such cases, a serviceables inventory is used at times where excess production capacity is available during the collection interval in order to reduce its length. For determining the respective maximal interval lengths, a trade-off between serviceables and recoverables holding costs is made.

Beside the availability of returns there might also exist other limitations for remanufacturing, e.g. a maximal possible rate. Although in contrast to the previously considered case, the current capacity of the system always suffices to immediately satisfy demand, cost reductions due to exploiting the recovery cost advantage led to a diversity of situations where inventory is used. In a situation where demand is higher than both return rate and remanufacturing constraint, for instance before the constraint leaves the binding state recoverables are collected for later use. Under other circumstances even preremanufacturing to a serviceables stock can be favorable.

We also considered two strategic applications of dynamic product recovery. The integration of recovery knowledge acquisition into the basic framework allowed for remanufacturing even under circumstances where there is no immediate advantage over producing new items. Besides the usual results of fully or partly anticipating learning effects when having zero or positive interest rates, another outcome affects the decision when to start the remanufacturing process. This is only possible because there exists another way to fill demand. This decision is accompanied by another motivation to keep stock, namely to postpone the start of remanufacturing. A similar effect results mainly from the combined problem of choosing upon the product's design, a corresponding recovery mode, and the investment time into a remanufacturing facility under product life cycle conditions.

There exist several other recent applications of using the optimal control approach for dynamic product recovery than those already presented in the introductory chapter. Under the assumption of different variants of the same basic product with differing profitability of remanufacturing (e.g. spare parts for the original and an upgraded version of the product), it must be determined for which demand class returns should be used. In extending the basic model presented in Chapter 2, Kleber et al. (2002) showed that in contrast to that case it can be optimal to satisfy a certain demand from production although there still are recoverables available on stock from which demand could be serviced. The underlying trade-off balances differences in the profitability between the available recovery options and holding costs.

Another option that could be considered is not to immediately satisfy all demand but to backlog parts of it. This is preferable at the end of time periods with excess demand which are satisfied when excess returns become available. Here a trade-off between the associated costs (like price cuts required to keep a customer) which also may depend on the time a customer is required to wait and the remanufacturing cost advantage is balanced. An optimal control model analyzing this issue has been presented by Kiesmüller et al. (2000).

Throughout this work it was assumed that demand and returns could not be influenced by the decision maker. An active return acquisition management, however, can considerably enhance profitability by increasing or reducing the availability of returns when needed. This can be implemented through incentives like advertising or buy back prices for returns. An application of optimal control for dynamically setting a buy back price is provided by Minner and Kiesmüller (2002).

List of Symbols

Basic Notation Introduced in Chapters 2 and 3

Indices and general

t	time index
(.)(t)	time dependence
.*	optimal values
x^{-}	left-side limit of x
x^+	right-side limit of x

Parameters of the dynamic environment

d(t)	demand rate at time t
u(t)	return rate at time t
T	planning horizon

Processes (states and variables)

$y_s(t)$	physical stock serviceables
$y_u(t)$	physical stock recoverables
y_s^0	initial stock serviceables
y_u^0	initial stock recoverables
p(t)	production rate
\bar{p}	production capacity
r(t)	remanufacturing rate
\bar{r}	remanufacturing capacity
w(t)	disposal rate
w_0	initial disposal quantity

 $Cost \ and \ cash \ flow \ parameters$

α	discount rate or continuous interest rate
c_p	variable unit production cost

170 6 Conclusions

c_r	variable unit remanufacturing cost
c_w	variable unit disposal cost or negative salvage revenue
h_s	holding cost rate serviceables
h_u	holding cost rate recoverables

$Important\ optimization\ variables$

H	Hamiltonian
L	Lagrangian
λ_{\cdot}	co-state / adjoint variable
$\mu_{.}$	Kuhn-Tucker-Multiplier for pure control conditions
k_{\cdot}	Kuhn-Tucker-Multiplier for pure state conditions
$\theta_{\cdot} \in \Theta_{\cdot}$	time points where co-state jumps
$\eta_{.}$	jump heights at time points where co-state is not continuous
$ heta_{e,i}$	entry time of a Case i interval
$\theta_{x,i}$	exit time of a Case i interval

Miscellaneous

Case A \rightarrow Case B Transition from Case A to Case B

Additional Notation in Chapter 4

Learning related parameters

b	learning rate parameter
$\begin{array}{c}c^0_{(r)}\\R(t)\end{array}$	unit costs of producing (remanufacturing) a first item
$\hat{R}(t)$	cumulative remanufacturing up to t as a proxy to remanufacturing
	knowledge
R_0	initial stock of remanufacturing knowledge
$egin{array}{c} R_0 \ ilde{R} \end{array}$	break-even total remanufacturing quantity under zero discount
	rate conditions

Important optimization variables

$\lambda_R(t)$	value of acquiring knowledge at time t
λ_u^{\max}	maximum level of λ_u if remanufacturing takes place
θ_l	time after which remanufacturing takes place

Additional Notation in Chapter 5

Parameters of the dynamic environment

M	number of potential adopters
P	coefficient of innovation

Cash flow parameters

K_p^s	initial investment expenditures for single use production
$ \begin{array}{c} K_p^r \\ K_r \end{array} $	initial investment expenditures for reuse production
K_r^r	investment expenditures for remanufacturing facility
c_p^s	variable unit production cost at single use production
$\begin{array}{c} c_p^s \\ c_p^r \end{array}$	variable unit production cost at reuse production

Policy parameters

t_r	time of remanufacturing investment
t_e	start time of storing collected returns
t_x	time where all stored returns are used up

List of Figures

2.1	Product recovery system	10
2.2	Demands and returns in the example	10
2.3	Optimal case transitions in the basic model	22
2.4	Example of a Case 2 interval which has maximal length	23
2.5	Example for applying the forward algorithm	26
2.6	Optimal Case 2 intervals and co-state λ_u development in	
	Example 2.1	30
2.7	Optimal inventory levels in Example 2.1	30
2.8	Optimal Case $1/2$ intervals and co-state λ_u development in	
	Example 2.2.	32
2.9	Optimal inventory levels in Example 2.2	32
2.10	Optimal policies and co-state λ_u development in Example 2.3	33
2.11	Optimal inventory levels in Example 2.3	33
3.1	Optimal case transitions at the beginning of the planning	
~ ~	horizon when considering a manufacturing constraint	56
3.2	Optimal case transitions when considering a manufacturing	
	constraint	56
3.3	Optimal case transitions when considering a manufacturing	
~ .	constraint assuming $u < d$	57
3.4	Demands, returns and capacity constraint in Example 3.1	63
3.5	Optimal solution of Example 3.1	64
3.6	Optimal co-state developments in Example 3.1	64
3.7	Demands, returns and capacity constraint in Example 3.2	65
3.8	Optimal solution of Example 3.2	65
3.9	Optimal co-state developments in Example 3.2	65
		66
	Optimal solution of the unrestricted problem for Example 3.3	66
	Optimal solution of Example 3.3	67
3.13	Optimal co-state developments in Example 3.3	67

3.14	Optimal case transitions when considering a remanufacturing
	constraint
3.15	Optimal case transitions when considering a remanufacturing
	constraint assuming high demand
3.16	An interval where returns are kept during a bottleneck
	collection interval for later use within a consumption period
	with maximal length
3.17	A consumption period showing a Case $2(1)$ interval
	A consumption period showing a Case 1(2) interval
	A consumption period terminating into a Case $3(2)$ interval 83
	Optimal case transitions when considering a remanufacturing
0.20	constraint assuming low demand
3.21	An interval where returns are remanufactured with maximal
0.21	rate during a serviceables accumulation interval for later use
	within a demand bottleneck with maximal length
	within a demand bottleneck with maximal length
4.1	Single-use cameras recycled by Kodak between 1990 and 2003 97
4.2	Optimal case transitions in a product recovery system with
4.2	learning
4.3	Demands and returns in Examples 4.1 and 4.2
4.3 4.4	Optimal solution of Example 4.1
4.4	Optimal solution of Example 4.1
4.6	Optimal development of co-state λ_R in Example 4.1
4.7	Optimal solution of Example 4.2
4.8	Optimal development of co-state λ_u in Example 4.2
4.9	Optimal development of co-state λ_R in Example 4.2
	Demands and returns in Examples 4.3 and 4.4
4.11	1 1 1
	Optimal solution of Example 4.4
	Optimal development of co-state λ_u in Example 4.4
4.14	Optimal development of co-state λ_R in Example 4.4
5.1	A generic environment complying with assumptions A.1-A.3130
$5.1 \\ 5.2$	Demands and returns in a specific model
5.3	Optimal decisions in investment project (a)
5.4	Optimal decisions in investment project (b)
5.5	Typical shape of the objective function and its first derivative
	in a demand/return scenario with (i) high and (ii) low
.	investment expenditures
5.6	Return rate surpassing u^{crit} at optimal final investment time t_r^* . 138
5.7	Optimal decisions in investment project (c)141
5.8	Cases representing candidates for the optimal solution 143
5.9	Optimal decisions in investment project (c) when having a
	limited remanufacturing rate

List of Tables

2.1	Main results of optimal cases in the basic model 20
2.2	Initial inventory levels in Examples 2.1-2.3
2.3	Cost parameters in examined scenarios
2.4	Maximum discount rates α_{max} for different error levels when
	using Approximation 1 38
2.5	Maximum discount rates α_{max} for different error levels when
	using Approximation 2 38
3.1	Main results of optimal cases when considering a
	manufacturing constraint
3.2	Demands, returns and capacity constraint functions as well as
	cash flow parameters in three considered scenarios
3.3	Main results of optimal cases when considering a
	remanufacturing constraint
4.1	Main results of optimal cases in a product recovery system
	with learning
4.2	Optimal Case 2(1) intervals in Examples 4.1 and 4.2
5.1	Four considered return scenarios
5.2	Maximal and average NPV deviations of heuristics from
	optimal solution (in percent) within the considered return
	scenarios
5.3	Maximal and average NPV deviations of heuristics from
	optimal solution (in percent) with relative high and low
	recovery cost advantages151
5.4	Maximal and average NPV deviations of heuristics from
	optimal solution (in percent) with relative high and low
	discount rates

5.5	Maximal and average NPV deviations of heuristics from
	optimal solution (in percent) with relative high and low
	out-of-pocket holding cost rate
5.6	Maximal and average NPV deviations of heuristics from
	optimal solution (in percent) with relative high and low
	investment expenditures. $\dots \dots \dots$
5.7	Average difference, sample standard deviation and z test
	values for comparison of the average performance of different
	heuristics
5.8	Average difference, sample standard deviation and z test
	values for comparison of the average performance of different
	heuristics (continued)
5.9	Standard deviation of the relative error of heuristic solutions
	from optimal solution. $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 166$
5.10	Z test values for comparison of the average performance of the
	same heuristic between different scenarios/settings

References

- Adler, P. S. and Clark, K. B. (1991). Behind the learning curve: A sketch of the learning process. *Management Science*, 37(3):267–281.
- Argote, L. and Epple, D. (1990). Learning curves in manufacturing. *Science*, 247:920–924.
- Arrow, K. J. and Karlin, S. (1958). Production over time with increasing marginal costs. In Arrow, K. J., Karlin, S., and Scarf, H., editors, *Studies in the Mathematical Theory of Inventory and Production*, pages 61–69. Stanford University Press, Stanford.
- Baranczyk, R. (2001). Analyse optimaler Kontrollen in Modellen zur integrierten Produktions- und Recyclingplanung mittels linearer Optimierung. Diploma thesis, Faculty of Economics and Management, Otto-von-Guericke University Magdeburg, Germany.
- Bass, F. M. (1969). A new product growth for model consumer durables. Management Science, 15(5):215–227.
- Bellman, R. E. (1957). Dynamic Programming. Princeton University Press, Princeton, N.J.
- Corbey, M., Inderfurth, K., van der Laan, E., and Minner, S. (1999). The use of accounting information in production and inventory control. Working paper No 24/99, Faculty of Economics and Management, Otto-von-Guericke University Magdeburg, Germany.
- de Brito, M. P. and Dekker, R. (2003). Modelling product returns in inventory control – exploring the validity of general assumptions. *International Journal of Production Economics*, 81–82:225–241.
- de Brito, M. P. and Dekker, R. (2004). A framework for reverse logistics. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 3–27. Springer, Berlin.
- de Brito, M. P., Dekker, R., and Flapper, S. D. P. (2005). Reverse logistics: A review of case studies. In Fleischmann, B. and Klose, A., editors, *Distribution Logistics: Advanced Solutions to Practical Problems*, pages 243–282. Springer, Berlin.
- de Brito, M. P. and van der Laan, E. A. (2002). Inventory control with product returns. Econometric Institute Report 279, Erasmus University Rotterdam, The Netherlands.

- Debo, L. G., Toktay, L. B., and Van Wassenhove, L. N. (2005). Market segmentation and product technology selection for remanufacturable products. *Management Science*, 51(8):1193–1205.
- Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N. (2004). Quantitative models for reverse logistics decision management. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 25–41. Springer, Berlin.
- Dekker, R. and van der Laan, E. (2003). Inventory control in reverse logistics. In Guide, Jr., V. D. R. and Van Wassenhove, L. N., editors, *Business Aspects of Closed-Loop Supply Chains*, pages 175–200. Carnegie Mellon University Press, Pittsburgh, PA.
- Dekker, R., van der Laan, E. A., and Inderfurth, K. (2000). A review on inventory control for joint manufacturing and remanufacturing. MCPL Proceedings. IFAC Grenoble.
- Dobos, I. (1998). Production-inventory control under environmental constraints. International Journal of Production Economics, 56-57:123–131.
- Dutton, J. M. and Thomas, A. (1984). Treating progress functions as managerial opportunity. Academy of Management Review, 9:235–247.
- Ebert, R. J. (1976). Aggregate planning with learning curve productivity. Management Science, 23(2):171–182.
- Feichtinger, G. and Hartl, R. F. (1986). Optimale Kontrolle ökonomischer Prozesse: Anwendungen des Maximumprinzips in den Wirtschaftswissenschaften. de Gruyter, Berlin.
- Fleischmann, B. (2001a). On the use and misuse of holding cost models. In Kischka, P., Leopold-Wildburger, U., Möhring, R. H., and Radermacher, F.-J., editors, Models, Methods and Decision Support for Management - Essays in Honor of Paul Stähly, pages 147–164. Physica, Heidelberg.
- Fleischmann, M. (2001b). Quantitative Models for Reverse Logistics. Lecture Notes in Economics and Mathematical Systems 501. Springer, Berlin.
- Fleischmann, M., Bloemhof-Ruwaard, J. M., Dekker, R., van der Laan, E. A., van Nunen, J. A. E. E., and Van Wassenhove, L. N. (1997). Quantitative models for reverse logistics: A review. *European Journal of Operational Research*, 103:1–17.
- Fleischmann, M. and Minner, S. (2004). Inventory management in closed-loop supply chains. In Dyckhoff, H., Lackes, R., and Reese, J., editors, *Supply Chain Management and Reverse Logistics*, pages 115–138. Springer, Berlin.
- Gaimon, C. (1988). Simultaneous and dynamic price, production, inventory and capacity decisions. *European Journal of Operational Research*, 35:426–441.
- Geyer, R. and Van Wassenhove, L. N. (2005). The impact of constraints in closedloop supply chains: The case of reusing components in product remanufacturing. In Fleischmann, B. and Klose, A., editors, *Distribution Logistics: Advanced Solutions to Practical Problems, Lecture Notes in Economics and Mathematical Systems 544*, pages 203–219. Springer, Berlin.
- Goldstein, L. (1994). The strategic management of environmental issues: A case study of Kodak's single-use cameras. M.S. Thesis, Sloan School of Management, MIT, Cambridge, M.A.
- Grubbström, R. W. (1980). A principle for determining the correct capital costs of work-in-progress and inventory. *International Journal of Production Research*, 18(2):259–271.

- Guide, Jr., V. D. R., Harrison, T. P., and Van Wassenhove, L. N. (2003a). The challenge of closed-loop supply chains. *Interfaces*, 33(6):3–6.
- Guide, Jr., V. D. R., Jayaraman, V., and Linton, J. D. (2003b). Building contingency planning for closed-loop supply chains with product recovery. *Journal of Operations Management*, 21:259–279.
- Guide, Jr., V. D. R., Jayaraman, V., Srivastava, R., and Benton, W. C. (2000). Supply-chain management for recoverable manufacturing systems. *Interfaces*, 30(3):125–142.
- Guide, Jr., V. D. R. and Van Wassenhove, L. N. (2001). Managing product returns for remanufacturing. *Production and Operations Management*, 10(2):142–155.
- Guide, Jr., V. D. R. and Van Wassenhove, L. N. (2002). The reverse supply chains. *Harvard Business Review*, February 2002:2–3.
- Gulledge, Jr., T. R. and Khoshnevis, B. (1987). Production rate, learning, and program costs: Survey and bibliography. *Engineering Costs and Production Eco*nomics, 11:223–236.
- Hartl, R. F. (1995). Production smoothing under environmental constraints. Production and Operations Management, 4(1):46–56.
- Holt, C. C., Modigliani, F., Muth, J. F., and Simon, H. A. (1960). Planning Production, Inventories, and Work Force. Prentice-Hall, Englewood Cliffs, NJ.
- Inderfurth, K., Flapper, S. D. P., Lambert, A. J. D., Pappis, C. P., and Voutsinas, T. G. (2004). Production planning for product recovery management. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 249–274. Springer, Berlin.
- Jaber, M. Y. and Bonney, M. (2001). Economic lot sizing with learning and continuous time discounting: Is it significant? *International Journal of Production Economics*, 71:135–143.
- Kamien, M. I. and Schwartz, N. L. (1991). Dynamic Optimization: The Calculus of Variations and Optimal Control in Economics and Management. North-Holland, Amsterdam, 2nd edition.
- Kantor, P. B. and Zangwill, W. I. (1991). Theoretical foundation for a learning rate budget. *Management Science*, 37(3):315–330.
- Kaplan, R. S. and Atkinson, A. A. (1998). Advanced Management Accounting. Prentice Hall, Upper Saddle River, NJ, 3rd edition.
- Kiesmüller, G. P., Minner, S., and Kleber, R. (2000). Optimal control of a one product recovery system with backlogging. *IMA Journal of Mathematics Applied* in Business and Industry, 11:189–207.
- Kiesmüller, G. P., Minner, S., and Kleber, R. (2004). Managing dynamic product recovery: An optimal control perspective. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 221–247. Springer, Berlin.
- Kistner, K.-P. and Dobos, I. (2000). Optimal production-inventory strategies for a reverse logistics system. In Dockner, E. J., Hartl, R. F., Luptačik, M., and Sorger, G., editors, *Optimization, Dynamics, and Economic Analysis – Essays* in Honour of Gustav Feichtinger, pages 246–258. Physica, Heidelberg.
- Klausner, M. and Hendrickson, C. T. (2000). Reverse-logistics strategy for product take-back. *Interfaces*, 30(3):156–165.

- Kleber, R., Minner, S., and Kiesmüller, G. P. (2002). A continuous time inventory model for a product recovery system with multiple options. *International Journal* of Production Economics, 79:121–141.
- Kodak (1999). 1999 Annual Report. The Kodak Corporation, Rochester, NY.
- Kodak (2000). 2000 Annual Report. The Kodak Corporation, Rochester, NY.
- Kodak (2001). 2001 Annual Report. The Kodak Corporation, Rochester, NY.
- Kodak (2002). 2002 Annual Report. The Kodak Corporation, Rochester, NY.
- Kodak (2003). 2003 Annual Report. The Kodak Corporation, Rochester, NY.
- Kurawarwala, A. A. and Matsuo, H. (1996). Forecasting and inventory management of short life-cycle products. Operations Research, 44(1):131–150.
- Lapré, M. A., Mukherjee, A. S., and Van Wassenhove, L. N. (2000). Behind the learning curve: Linking learning activities to waste reduction. *Management Sci*ence, 46(5):597–611.
- Li, G. and Rajagopalan, S. (1998). A learning curve model with knowledge depreciation. European Journal of Operational Research, 105:143–154.
- Liao, W. M. (1979). Effects of learning on resource allocation decisions. Decision Sciences, 10:116–125.
- Mahajan, V., Muller, E., and Bass, F. M. (1993). New-product diffusion models. In Eliashberg, J. and Lilien, G. L., editors, *Marketing*, Handbooks in Operations Research and Management Science 5, chapter 8, pages 349–408. North-Holland, Amsterdam.
- Majumder, P. and Groenevelt, H. (2001). Competition in remanufacturing. Production and Operations Management, 10:125–141.
- Minner, S. and Kiesmüller, G. P. (2002). Dynamic product acquisition in closed loop supply chains. FEMM Working Paper No 9/2002, Otto-von-Guericke University Magdeburg, Germany.
- Minner, S. and Kleber, R. (2001). Optimal control of production and remanufacturing in a simple recovery model with linear cost functions. OR Spektrum, 23:3–24.
- Minner, S. and Lindner, G. (2004). Lot sizing decisions in product recovery management. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 157–179. Springer, Berlin.
- Pegels, C. C. (1976). Start up or learning curves some new approaches. Decision Sciences, 7:705–713.
- Reeves, G. R. and Sweigart, J. R. (1981). Product-mix models when learning effects are present. *Management Science*, 27(2):204–212.
- Reinhardt, F. L. (1999). Bringing the environment down to earth. Harvard Business Review, July/August:149–157.
- Seierstad, A. and Sydsæter, K. (1987). Optimal Control Theory with Economic Applications. Advanced Textbooks in Economics 24. North-Holland, Amsterdam.
- Seitz, M. A. and Peattie, K. (2004). Meeting the closed-loop challenge: The case of remanufacturing. *California Management Review*, 46(2):74–89.
- Sethi, S. P. and Thompson, G. L. (2000). Optimal Control Theory: Applications to Management Science and Economics. Kluwer Academic Publishers, Boston, MA, 2nd edition.
- Silver, E. A., Pyke, D. F., and Peterson, R. (1998). *Inventory Management and Production Planning and Scheduling*. John Wiley & Sons, New York, 3rd edition.

- Slack, N. and Lewis, M. (2003). Operations Strategy. Prentice Hall, Upper Saddle River, NJ.
- Smunt, T. L. and Morton, T. E. (1985). The effects of learning on optimal lot sizes: Further developments on the single product case. *IIE Transactions*, 17(1):33–37.
- Stock, J., Speh, T., and Shear, H. (2002). Many happy (product) returns. Harvard Business Review, July 2002:2–3.
- Sydsæter, K. and Hammond, P. J. (1995). Mathematics for Economic Analysis. Prentice Hall, Englewood Cliffs, NJ.
- Teunter, R. H., Inderfurth, K., Minner, S., and Kleber, R. (2005). Reverse logistics in a pharmaceutical company: A case study. In Flapper, S. D. P., van Nunen, J. A. E. E., and Van Wassenhove, L. N., editors, *Managing Closed-Loop Supply Chains*, pages 21–31. Springer, Berlin.
- Teunter, R. H., van der Laan, E., and Inderfurth, K. (2000). How to set the holding cost rates in average cost inventory models with reverse logistics? *Omega*, 28:409– 415.
- Teunter, R. H. and van der Laan, E. A. (2004). Valuation of inventories in systems with product recovery. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 275–291. Springer, Berlin.
- Thierry, M., Salomon, M., Van Nunen, J., and Van Wassenhove, L. N. (1995). Strategic issues in product recovery management. *California Management Review*, 37(2):114–135.
- Thierry, M. C. (1997). An Analysis of the Impact of Product Recovery Management on Manufacturing Companies. PhD thesis, Erasmus University Rotterdam, The Netherlands.
- Thompson, G. L. and Sethi, S. P. (1980). Turnpike horizons for production planning. Management Science, 26(3):229–241.
- Toffel, M. W. (2003). The growing strategic importance of end-of-life product management. *California Management Review*, 45(3):102–129.
- Toffel, M. W. (2004). Strategic management of product recovery. *California Management Review*, 46(2):120–141.
- van der Laan, E. A., Kiesmüller, G., Kuik, R., Vlachos, D., and Dekker, R. (2004). Stochastic inventory control for product recovery management. In Dekker, R., Fleischmann, M., Inderfurth, K., and Van Wassenhove, L. N., editors, *Reverse Logistics: Quantitative Models for Closed-Loop Supply Chains*, pages 181–220. Springer, Berlin.
- Wirl, F. (1991). Evaluation of management strategies under environmental constraints. European Journal of Operational Research, 55:191–200.
- Wright, T. P. (1936). Factors affecting the cost of airplanes. Journal of Aeronautical Sciences, 3:122–128.
- Yelle, L. E. (1979). The learning curve: Historical review and comprehensive survey. Decision Sciences, 10(2):302–328.
- Zangwill, W. I. and Kantor, P. B. (1998). Toward a theory of continuous improvement and the learning curve. *Management Science*, 44(7):910–920.